

Jeffrey, Susanne K E (2010) *Epidemiology, cost and prevention of road traffic crash injuries in Strathclyde, Scotland.* PhD thesis.

http://theses.gla.ac.uk/1448/

Copyright and moral rights for this thesis are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This thesis cannot be reproduced or quoted extensively from without first obtaining permission in writing from the Author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the Author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Glasgow Theses Service http://theses.gla.ac.uk/ theses@gla.ac.uk

Epidemiology, cost and prevention of road traffic crash injuries in Strathclyde, Scotland

Susanne K E Jeffrey BA(HONS) MSc

Thesis submitted for the degree of Doctor of Philosophy

Division of Developmental Medicine Faculty of Medicine University of Glasgow

March 2009

© [SUSANNE JEFFREY] [MARCH 2009]

EXECUTIVE SUMMARY

Background

Road traffic crash (RTC) injuries affect 20 to 50 million people worldwide every year, causing premature death or disability as well as incurring large costs to individuals and society. In the UK, the number of RTC casualties is underestimated if based solely on police records, as many casualties are unreported to the police. "Safety" (speed and red light) cameras have shown to be an effective way of combating RTCs and in 2000 a national scheme was rolled out in the UK.

Aim and objectives

The overall aim of the study was:

To investigate the epidemiology, cost and prevention of RTC injuries in the Strathclyde police region of Scotland.

The specific objectives of the study were:

- To establish the overall epidemiology and accuracy of reporting of RTC injuries in Strathclyde.
- 2. To determine the epidemiology of RTC injuries and the effectiveness of safety cameras at the camera sites in Strathclyde with special reference to different road users, RTC types and severity, before and after camera installation.
- 3. To estimate the economic burden of hospital admissions due to RTC injuries in Strathclyde and at the camera sites before and after installation.

Methods

Nine years (1997 to 2005) of police road casualty records (STATS19) and National Health Service hospitalisation records (SMR01) from the Strathclyde region were linked.

The linkage resulted in nearly 11,000 police casualty records relating to approximately 30,000 hospital and death records. Unlinked RTC hospital and police casualties (nearly 9,000 and 70,000 respectively) were also utilised in the analysis.

The study employed a range of epidemiological and economic methods. These included descriptive epidemiology (evaluating distributions of linked and unlinked records, length of stay and cost analysis), analytical epidemiology (examining associations using chi square and logistic regression models) and interventional epidemiology (before and after study). The economic evaluation utilised weighted mean costs.

The focus of analysis was threefold: 1. Epidemiology of RTC, injuries and accuracy of police recording, 2. Epidemiological impact of safety cameras, 3. Cost of road traffic crashes a) in Strathclyde and b) at safety camera sites.

Results

<u>Epidemiology of RTC injuries in Strathclyde</u>: Older age and less protected road users (i.e. pedestrians and two-wheeled vehicle users) had a higher risk of a more severe outcome in RTCs. Head injuries were more common among pedestrians and pedal cyclists, while car occupants more often suffered injuries to the thorax and abdomen/lower back/lumbar spine.

<u>Accuracy of police reporting</u>: 45% of RTC hospital admissions were not recorded by police. Casualty characteristics significantly associated with underreporting were: no third party involvement, older age, casualties from early in the study period, type of road user (especially pedal cyclist), hospitalisation as a day case and female gender.

Seriously injured casualties recorded by police (STATS19) declined in frequency more than the RTC hospitalised injuries (SMR01) (38% and 21% respectively). Linked SMR01 casualties that were coded "slight" by the police increased by 5% over time, while linked SMR01 casualties coded "serious" declined by 27%.

<u>Safety camera impact</u>: Compared to the rest of Strathclyde, there was a significantly greater downward linear time trend of RTC incidence at the camera sites. The impact of cameras on RTCs over time appeared stable. Cameras seemed to be effective in reducing the incidence of serious or fatal RTC injuries, as well as injuries associated with multiple-vehicle and non-junction RTCs.

<u>Cost of RTC casualties in Strathclyde</u>: Total inpatient costs were conservatively estimated at £7.3 million yearly (linked records). Head and lower extremity injuries incurred the highest total costs (28% and 34% respectively). Pedestrian injuries, constituting 36% of the total, incurred 44% of total costs. Casualties from deprived areas, and pedestrians in particular, incurred higher hospital costs than other road user groups.

<u>Cost of RTCs at safety camera sites</u>: 17% of all injured before safety camera installation were hospitalised, while 13% of casualties after installation were hospitalised. The mean costs of (surviving) casualties admitted to hospital declined by 24% after installation and the mean daily cost declined by 55%.

Conclusions

RTC injury incidence in Strathclyde declined over the study period, which is in line with expectations of developed countries. Young and elderly people as well as unprotected road users carry a disproportionately great RTC injury burden.

Many hospitalised RTC casualties were not recorded by police and there appears to have been an increasing tendency over time for police officers to report injuries as slight rather than serious.

National (UK) statistics of RTCs should be interpreted with caution in the light of these findings and routinely linking police and hospital data would enhance the quality of RTC casualty statistics. Linking police and hospital RTC records provide a more comprehensive source for road traffic analysis than any of the sources separately. Routine data linkage would also facilitate the evaluation of time trends in relation to national road casualty reduction targets.

The study indicates that the most costly RTCs occur in areas with high levels of deprivation, a history of pedestrian RTCs, elderly and child casualties, roads with many non-junction RTCs and 30 mph speed limits.

The evaluation of safety cameras strongly suggests that they are effective in reducing both road casualty incidence and severity and that the reduction in incidence is sustained over time. Additionally, safety cameras in Strathclyde may have contributed to a saving of over \pounds 5 million. Cameras thus fulfil an important public health, as well as law enforcement, function and should continue to play a central role in traffic calming.

This study has demonstrated the value of utilising multiple data sources in the road traffic injury field.

ACKNOWLEDGEMENTS

I owe thanks to many people who have supported me, in various ways, on the road to completing this piece of work. Most grateful thanks go to my supervisor, Dr David Stone, Professor of Paediatric Epidemiology, University of Glasgow, who has provided invaluable guidance and support coupled with a great amount of patience throughout my time as a PhD student. We have, together, endured both thesis fatigue and published papers whilst keeping a supervisor-student relationship alive at (literally) very long distance.

Many thanks are also due to my "extended" supervisory team who have aided me with their expertise in different fields: Dr Avril Blamey, Senior Public Health Adviser, NHS Health Scotland, Kirstin Dickson, Health Economist, NHS Ayrshire and Arran, Dr Mhairi Mackenzie, Senior Lecturer in Public Policy, University of Glasgow and Kirsten Major, Director of Strategic Planning and Performance, NHS Ayrshire and Arran.

I would also like to thank a number of very important people in the PEACH Unit at University of Glasgow: Rita Dobbs for aiding me in all administrative matters and being encouraging, Mhairi Campbell and Janne Pearson for taking the time to proof-read and providing excellent comments and Dr Andrea Sherriff for valuable statistical advice.

I would like to acknowledge the people who have contributed in retrieving data and other necessary information: Chris Cooper, Senior Analyst, Strathclyde Police Information Resources, Gladys Cadden, Project Manager, Strathclyde Safety Camera Partnership, Steven Whitelocke, Analyst, Strathclyde Safety Camera Partnership and David Clark, Principal Information Analyst, Information Services Division.

Thanks are also due to my family: my mother who has helped in looking after my children, Irma and Alisdair, who unknowingly of what reason, have sacrificed time with me –I promise to make it up to you. My final acknowledgement goes to my husband, Cameron, who has provided the time and space for me to work in my spare time, as well as love and support.

TABLE OF CONTENTS

EXECUTIVE SUMMARY	2
ACKNOWLEDGEMENTS	6
TABLE OF CONTENTS	7
LIST OF FIGURES	11
	10
	12
DEFINITIONS, GLOSSARY AND ABBREVIATIONS	14
CHAPTER 1 . OVERVIEW OF THE EPIDEMIOLOGY, COST AND PREVENTION OF ROAD T	
CRASHES	I7
	18
1.1 I IME TRENDS IN INCIDENCE OF ROAD TRAFFIC CRASHES	19
1.1.1 Giobal time trends in Sectland	19
	20
1.2 NISK FACTORS AND CAUSES OF ROAD TRAFFIC CRASHES	20 21
1.2.1 Driving behaviour related to age and gender	21
1.2.1.2 Exposure to road traffic crash risk related to gender and age	22
1.2.1.3 Risk related to children and elderly	22
1.2.2 Substance use and road traffic crash risk	23
1.2.2.1 Alcohol intake and risk of road traffic crash	24
1.2.2.2 Police detection of drink driving	24
1.2.2.3 DITIK UTVITUTI TVOIVETTETT and age	25 25
1.2.2.5 Smoking and risk of road traffic crash	26
1.2.3 Vehicle speed and road traffic crash risk	26
1.2.3.1 Definition of speed-related road traffic crashes	26
1.2.3.2 Speed related to road traffic crash incidence	27
1.2.3.3 Speeding drivers	27
1.2.4 Deprivation and road traffic crash risk	
1.2.5 Seasonally and road traffic crash risk	
1.2.0 Provinus road traffic crash or injury and road traffic crash risk	
1.3 CONSECUENCES OF POAD TRAFFIC CRASHES	29
1.3.1 Road traffic crash injury and sequelae	30
1.3.1.1 Life-threatening and disabling injuries	31
1.3.1.2 Head and neck injuries resulting from road traffic crashes	31
1.3.1.3 Long-term effects of road traffic crash injury	32
1.3.2 Risk factors associated with injury severity	32
1.3.2.1 Age and gender as risk of road traffic crash consequences	33
1.3.2.2 Alconol and risk of road traffic crash consequences.	33
1.3.2.4 Vehicle speed and risk of road traffic crash consequences	34
1.3.2.5 Other factors relating to risk of road traffic crash consequences	35
1.3.3 Psychosocial sequelae	35
1.3.3.1 Prevalence of psychosocial disorders following road traffic crashes	36
1.3.3.2 Psychosocial long-term effects following road traffic crashes	36
1.3.4 Risk factors associated with psychosocial disorders following a road traffic crash.	37
1.4 VALIDITY OF ROAD TRAFFIC CASUALTY REPORTING	3/
1.4.1 Characteristics of casualities missed by the police	30
1.4.2 Recommendations on road traffic casuality in have a solution worldwide	30 30
1.4.4 Linking hospital and police records of road traffic casualties in LIK	30
1.5 Costs of Road TRAFFIC CRASHES	40
1.5.1 Types of costs incurred by road traffic crashes	40
1.5.2 Costs of road traffic crash iniuries	41
1.6 NATIONAL ROAD SAFETY POLICY	42
1.7 PREVENTION OF ROAD TRAFFIC CRASHES	43
1.7.1 The three E's in prevention: Engineering, Education and Enforcement	43
1.7.2 Types of road traffic crash prevention	44

1.7.2.1	Red light cameras	44
1.7.2.2 I	Noving radar	44
1.7.2.3	Changing speed limits	45
1.7.2.4	Road engineering	45
1.7.2.5	Adaptive speed limiters	46
172 5000	d comorco	40
1.7.3 Speed	J Callelas Spood comorae in Australia	
1.7.3.1	Speed cameras in New Zealand	47
1733	Speed cameras in New Zealand	
1734	Speed cameras in LIK	49
1.7.3.5	New speed camera strategy in UK: Safety cameras	
1.7.3.6	Safety cameras in Strathclyde region	
1.7.3.7	Summary of five speed camera evaluations	
1.8 SUMMAR	Y OF CHAPTER 1	53
CHAPTER 2 . AIM	AND OBJECTIVES OF THE STUDY	56
2.1 CONTEXT	OF THE STUDY	57
2.2 RATIONA	E	
2.3 AIM AND	OB JECTIVES	57
2.0 RESEARC		58
		58
		50
2.0 SOIVIIVIAR	TOF CHAFTER Z	39
CHAPTER 3. MET	HODS	60
		<i>c</i> 1
3.1 OVERVIE	W OF METHODOLOGICAL APPROACH	01
3.1.1 Epide	miological methods	
3.1.1.1	Descriptive epidemiology	62
3113	Interventional epidemiology	
312 Econ	nic evaluation	63
313 Study	shire evaluation	64
3.2 DATA SOL	UPCES	65
321 Police	a road casualty records: STATS19	05
3.2.1 TUILE	hospitalisation records: Scottish marhidity records 01	
222 500#	sh index of multiple deprivation	07
3.2.3 Stolli 2.2.4 Strath	sn index of multiple deprivation	07
3.2.4 Suau	Camora site selection methods	
3242	Sites and crashes selected	08
33 LINKAGE	OF POLICE AND HOSPITAL POAD TRAFFIC CASUALTY RECORDS	70
331 How t	this linkage differs from others	71
332 Paran	neters for linkage	71
3321	Police road casualty records: STATS19	71
3.3.2.2	NHS hospitalisation records: SMR01	
3.3.3 Data	linkage methods and results	
3.3.3.1	Matching standards	
3.3.3.2	Linkage results	
3.3.3.3	Estimated accuracy of linkage	
3.4 CHARACT	FERISTICS OF THE LINKED POLICE-HOSPITAL DATA BASE	75
3.4.1 Police	eroad casualty data: STATS19	75
3.4.1.1	Casualty demographics	75
3.4.1.2	Casualty severity	75
3.4.1.3	Crash characteristics	76
3.4.2 NHS I	hospital discharge data: SMR01	77
3.4.2.1	njury definitions	77
3.4.2.2	Length of hospital stay	77
3.4.3 Unlink	(ea SMRU1 and STATS19	
3.4.4 Statis	tical power	85
3.5 ANALYSE	S	85
3.5.1 Epide	miology of road traffic crashes, injuries and accuracy of police recording	85
3.5.1.1		86
3.5.1.2	Stuay period	86
3.5.1.3	Sevency categories III STATSTY and SMRUT	80
3.3.1.4 I ステイテ	Linkeu anu unimikeu valegunes Assassina tima trands	07
3516	Tests of association with epidemiological characteristics	87
0.0.1.0		

3.5.1.7	Length of hospital stay analysis	. 88
3.5.2 Epide	miological impact of safety cameras	89
3.5.2.1	Records used in safety camera (STATS19) evaluation	. 89
3.5.2.2	Study period in safety camera (STATS19) evaluation	. 89
3.5.2.3	Safety cameras included in study	. 89
3.5.2.4	Assessing time trends in safety camera (STATS19) evaluation	. 90
3.5.2.5	Test of epidemiological characteristics in safety camera (STATS19) evaluation	. 90
3.5.2.6	Pre and post camera installation	. 91
3.5.3 Cost	of road traffic crashes in Strathclyde region	92
3.5.3.1	Records used in RTC cost in Strathclyde region	. 92
3.5.3.2	Study period for RTC cost in Strathclyde region	. 92
3.5.3.3	Follow-up admissions in RTC cost in Strathclyde region	. 93
3.5.3.4	Weighted mean costs	. 93
3.5.3.5	Costs by epidemiological variables	. 94
3.5.3.6	Cost of fatalities	. 95
3.5.3.7	Distribution of cost	. 95
3.5.4 Cost	of road traffic crashes at safety camera sites	97
3.5.4.1	Records used in RTC cost at safety camera sites	. 97
3.5.4.2	Study period in RTC cost at safety camera sites	. 97
3.5.4.3	Epidemiological variables included in RTC cost at safety camera sites	. 98
3.5.4.4	Injury definitions in RTC cost at safety camera sites	. 98
3.5.4.5	Costs per road traffic crash at safety camera sites	. 98
3.5.4.6	Maximum and mean maximum costs	. 99
3.6 SUMMAR	Y OF CHAPTER 3	99
HAPTER 4. RES	/UL I 5	•••••
4.1 EPIDEMIC	DLOGY OF ROAD TRAFFIC CRASHES, INJURIES AND ACCURACY OF POLICE RECORDING	6102
4.1.1 Time	trends in STATS19 and SMR01	. 102
4.1.1.1	Distribution of STATS19 data over time and severity	103
4.1.1.2	Distribution of the linked vs. unlinked STATS19 and SMR01 records	104
4.1.1.3	Time trends of linked and unlinked data	106
4.1.2 Deter	minants of road traffic crashes and severity	. 109
4.1.2.1	Age: incidence and severity	109
4122	Gender: incidence and severity	111
4123	Deprivation: incidence and severity	112
4124	Road user: incidence and severity	112
4125	Third party involvement: incidence and severity	115
41.3 Risk (of hospitalised casualties not being recorded by police: univariable and multivation	niahle
regulte 117		
1131	Linivariable results	117
4.1.3.1	Multivariable results	117
4.1.J.Z	of initial bit read user	120
4.1.4 Type	of injury by road user	. 120
4.1.5 Repo	rting levels by injury and length of hospital stay	. 122
4.2 EPIDEMIC	DLOGICAL IMPACT OF SAFETY CAMERAS 1	23
4.2.1 Epide	miology of road traffic crashes in Strathclyde region vs. at safety camera sites?	: 124
4.2.1.1	Time trends in incidence of road traffic crashes	124
4.2.2 Epide	miology of road traffic crashes at safety camera sites	. 126
4.2.2.1	Impact of cameras on road traffic crash epidemiology	126
4222	Camera effect over time	128
4.3 COST IMI	PACT OF ROAD TRAFFIC CRASHES	29
431 Coot	impact of road traffic crashes in Strathelyde region	120
7.J.I UUSLI 1211	Poad traffic casualty cost by injury type	. 129 130
4.3.1.1	Noau traffic excualty cost by mjuly type	121
4.3.1.Z	Noau traffic cocupity cost by road context: Stratholyde region	131
4.3.1.3	Road traffic equality cost by Poat Context. Stratholy de legion. Creation Orabolistic region.	134
4.3.1.4	Noau traine casually cost by Scottish index of infutiple Deprivation: Strathciyde region.	134
4.3.1.5	Road traffic cosualty cost by age. Strathclyde region	133
4.3.1.6	Road traine casualty cost over time: Stratnelyde region	130
4.3.2 Cost	impact of road traffic crashes at safety camera sites	. 137
4.3.2.1	Road trattic casualty cost by road user: safety camera sites	138
4.3.2.2	Road traffic casualty cost by injury type: safety camera sites	139
4.3.2.3	Cost by road traffic crashes: safety camera sites	139
4.3.2.4	Cost by type of road traffic crash at junction / non-junction: safety camera sites	140
4.3.2.5	Cost by type of road traffic crash and number of vehicles involved: safety camera sties	140
4.4 SUMMAR	Y OF CHAPTER 4 1	41
	CUSSION	
HAFTER J. DIS		•••••
5.1 MAINEIN	IDINGS	45

5.1.1	Epidemiology of road traffic injuries in Strathclyde region	145
5.1.2	Accuracy of police reported road traffic casualties	146
5.1.3	Epidemiological impact of safety cameras	146
5.1.4	Cost of road traffic crashes in Strathclvde region	147
515	Cost of road traffic crashes at safety camera sites	148
52 IN	TERPRETATION OF RESULTS	149
521	Endeminlogy of road traffic casualties in Strathclyde region	1/0
522	Accuracy of police reported read traffic casualties	149 150
5.2.2	Enidemiological import of potent comprop	
5.2.3	Cost of model (method on Salety Carrieras	
5.2.4	Cost of road traffic graphes of sefety servers sites	
5.2.5	Cost of road traffic crashes at safety camera sites	
5.3 51	RENGTHS AND WEAKNESSES OF STUDY	158
5.3.1	Strengths	158
5.3.1.	1 Validity of data sources and linkage	
5.3.1.	2 Size of sample and follow-up admissions	
5.3.1.	3 Regression towards the mean	
5.3.2	Weaknesses	160
5.3.2.	1 Before and after study	
5.3.2.	2 Retrieving the data	
5.3.2.	3 Validity of data sources and linkage	
5.3.2.	4 Errors related to analyses and results	
5.4 IM	PLICATIONS	162
5.5 UN	IANSWERED QUESTIONS	166
5.6 SU	GGESTIONS FOR FURTHER RESEARCH	167
5.6.1	Suggestions for continuing research on the linked data	168
5.6.1.	1 Study setting	168
5.6.1.	2 Data sources and linkage	168
5.6.1.	3 Analyses	169
5.6.2	Suggestions for other research	169
5.6.2.	1 Epidemiology of road traffic crashes in Strathclyde	169
5.6.2.	2 Accuracy of police reported casualties	170
5.6.2.	3 Impact of safety cameras	170
5.6.2.	4 Cost of road traffic crashes	171
5.7 Su	IMMARY OF CHAPTER 5	171
5.7.1	Epidemiology of road traffic casualties in Strathclyde	171
5.7.2	Accuracy of police reported road traffic casualties	
5.7.3	Epidemiological impact of safety cameras	173
5.7.4	Cost impact of road traffic crashes in Strathclyde	174
5.7.5	Cost impact of road traffic crashes at safety camera sites	
CHAPTER 6	. CONCLUSIONS AND RECOMMENDATIONS	
6.1 Co	NCLUSIONS	178
6.2 RE	COMMENDATIONS	180
6.2.1	Recommendations for scientists	180
622	Recommendations for practitioners	180 180
VI.C.C	Recommendations for policy makers	100 101
623		101 107
6.2.3		
6.2.3 6.3 SU	MMARY OF CHAPTER 6	102
6.2.3 6.3 SU REFERENCE	IMMARY OF CHAPTER 6	
6.2.3 6.3 Su REFERENCE APPENDIX	IMMARY OF CHAPTER 6	404
6.2.3 6.3 Su REFERENCE APPENDIX	Published paper	194
6.2.3 6.3 Su REFERENCE APPENDIX I. II.	Published paper	194 201
6.2.3 6.3 Su REFERENCE APPENDIX I. II. III.	Published paper Stats19 data guide Scottish Morbidity Records 01 and General records office Scotland mortality codes	194 201 204
6.2.3 6.3 Su REFERENCE APPENDIX I. II. II. III. V. V	Published paper Stats19 data guide Scottish Morbidity Records 01 and General records office Scotland mortality codes ICD 10 codes The Scottish record linkage system	194 201 204 207 209
6.2.3 6.3 SU REFERENCE APPENDIX I. II. II. II. V. V. VI	Published paper Stats19 data guide Scottish Morbidity Records 01 and General records office Scotland mortality codes ICD 10 codes The Scottish record linkage system	194 201 204 207 208 208

LIST OF FIGURES

Figure 1 Number of publications on PubMed with search terms: epidemiology, traffic and injuries by year
Figure 2 Map of UK to the left with Scotland in a darker shade. Map of Scotland to the right with Strathclyde in a darker shade
Figure 3 Map from ArcView of safety camera sites70
Figure 4 Flow chart of the selection process of follow-up admissions stemming from road traffic crashes between 1997 and 2004 (next page)80
Figure 5 Histogram of aggregated length of hospital stay for the selected admissions limited to 1,5 years (the y-axis should have been 8,000 and x-axis 1,200 in order to display all cases, but this would not have illustrated the skewness so well)
Figure 6 This is LoS (as in figure 5 above) transformed using the natural logarithm
Figure 7 Hospital admission costs up to 1,5 years post RTC96
Figure 8 Log transformed hospital admission costs up to 1,5 years post RTC96
Figure 9 Casualties by SMR01 road traffic crashes admissions and STATS19 killed and seriously injured records in Strathclyde region 1997-2005 per 100,000 population103
Figure 10 Numbers of SMR01 road traffic crash records over time by total, linked to STATS19 and unlinked records
Figure 11 Proportions of linked and unlinked STATS19 and SMR01 records 1997-2005108
Figure 12 Percentage of injured that had a fatal outcome within 1.5 years, by age and gender112
Figure 13 STATS19 recorded road traffic crashes rates in Strathclyde per 100,000 population by year and severity
Figure 14 Frequency distribution of road traffic crashes at camera sites and the remainder of Strathclyde by year
Figure 15 Rate ratios and 95% confidence intervals of road traffic crash incidence reduction after camera installation by road traffic crash type
Figure 16 Mean hospital costs (£) by SIMD quintile (1=, most deprived, 5= most affluent) and road user

LIST OF TABLES

Table 1 Methods, results and limitations of five speed camera studies	2
Table 2 Matching standards and linkage results –table provided by Information Statistics Division74	4
Table 3 Distribution of records by selection criteria and steps	0
Table 4 Tests of Normality of log transformed LoS values	4
Table 5 Number of casualties in each data set8	4
Table 6 Tests of Normality of log transformed cost9	7
Table 7 Casualty severity STATS19 data, Strathclyde, 1997-2005	4
Table 8 Road traffic crash casualty rates in Strathclyde per 100,000 population and trend gradient	4
Table 9 Distribution of linked and unlinked STATS 19 casualties over time (1997-2005) in Strathclyde10	5
Table 10 SMR01 data on road casualties unlinked and linked (to police data, STATS19), Strathclyde1997-2005	5
Table 11 Road traffic crash casualty rates in Strathclyde per 100,000 population by linked and unlinked data10	7
Table 12 Linked and unlinked SMR01 and STATS19 by year and police severity grading10 10°	7
Table 13 Road traffic crash casualty rates in Strathclyde per 100,000 population by linked and unlinked data and police severity grading10	8
Table 14 Linked and unlinked STATS19 and SMR01 by age group, data for 1997-200511	0
Table 15 Injury severity by age excluding unlinked SMR01 and year 2005	1
Table 16 Linked and unlinked STATS19 and SMR01 by gender, data for 1997-200511	1
Table 17 Injury severity by gender excluding unlinked SMR01 and year 2005	2
Table 18 Linked and unlinked STATS19 and SMR01 by SIMD, data for 1997-200511	3
Table 19 Injury severity by SIMD excluding unlinked SMR01 and year 200511	3
Table 20 Linked and unlinked STATS19 and SMR01 by road user, data for 1997-200411	5
Table 21 Injury severity by SIMD excluding unlinked SMR01 and year 200511	5
Table 22 Linked and unlinked STATS19 and SMR01 by collision/non-collision, data for 1997-2005	6
Table 23 Injury severity by collision/non-collision, excluding unlinked SMR01 and year 200511	6
Table 24 Results from univariable logistic regression11	8
Table 25 Results from multivariable logistic regression model 11	9
Table 26 Injury by road user: results from univariable logistic regression. Reference category: car occupant	1
Table 27 STATS19 severity proportions and mean length of stay of hospital admissions by (single)injury, comparing 1 st and 3 rd three year time periods12	3
Table 28 STATS19 recorded road traffic crash rates and trend gradient in Strathclyde per 100,000 population 12	5
Table 29 Rate ratios and 95% confidence intervals of road traffic crash incidence reduction after camera installation	6

Table 30 Rate ratios and 95% confidence intervals of road traffic crash incidence reduction after camera installation by road traffic crash type
Table 31 Rate ratios and 95% confidence intervals of "before" and "after" road traffic crashes, grouped by year
Table 32 Mean and total costs (\pounds) of hospital admissions by inpatient and day surgery130
Table 33 Mean and total costs (\pounds) of (surviving) hospital admissions by single injury130
Table 34 Mean and total costs (£) of slight injuries, hospital admissions and fatalities by road user 1997-2004
Table 35 Mean and total costs (£) of (surviving) hospital admissions by road user and vehicle type
Table 36 Mean and total costs (£) of slight injuries, hospital admissions and fatalities by road user and vicinity of junction
Table 37 Mean and total costs (£) of slight injuries, hospital admissions and fatalities by road user and posted speed limit on road
Table 38 Mean and total costs (£) of slight injuries, hospital admissions and fatalities by road user and SIMD
Table 39 Mean and total costs (£) of slight injuries, hospital admissions and fatalities by age136
Table 40 Mean and total costs (£) of, hospital admissions by time of road traffic crash
Table 41 Proportions of road traffic crashes and days Pre and Post SCI
Table 42 Total hospital admission cost up to 6 months post road traffic crash by injury severity 138
Table 43 Total hospital admission (survivors only) costs up to 6 months post road traffic crash, by road user 139
Table 44 Injury costs (surviving) hospitalised casualties Pre SCI and Post SCI
Table 45 Costs by type of road traffic crash140
Table 46 Costs by junction / non-junction type of road traffic crash, hospitalised survivors only140
Table 47 Cost by number of vehicles involved in road traffic crash, hospitalised survivors only141

DEFINITIONS, GLOSSARY AND ABBREVIATIONS

AIS	Abbreviated Injury Scale: is an anatomical scoring of injury severity. The scores range from 1=minor injury to 6= unsurvivable although the difference between each step of the scale are not equal.
BAC	<i>Blood Alcohol Concentration</i> : intoxication level of alcohol used in medical or legal situations. It is calculated by mass per volume and affected by weight, gender and body fat. BAC range from 0.02% (one standard drink) to 0.40%, which is the lethal level for 50% of the population.
Casualty	<i>Casualty</i> : is in this context referred to as a person who are injured or killed in a road traffic crash.
СТ	<i>Computed Tomography</i> : medical imaging method used to produce three dimensional and other types of type scans.
Crashes	<i>Crashes</i> : are here referred to as the road traffic incidence commonly referred to as "accidents". "Accidents" is not an acceptable term in traffic injury research as it implies that the incident is related to fate, chance or the act of god.
CVC	<i>Cervical Vertebral Column</i> : is the medical term for what is more commonly referred to as whiplash injuries, which are often associated with rear end road traffic crashes.
DALYs	<i>Disability Adjusted Life Years</i> : a combination of morbidity and mortality in number of healthy years lost.
DfT	Department for Transport: is the UK transport department and 'provides leadership across the transport sector to achieve its objectives, working with regional, local and private sector partners to deliver many of the services'.
DSM	Diagnostic and Statistical Manual of Mental Disorders: provides diagnostic criteria for mental disorders, including PTSD.
ETSC	<i>European Transport Safety Council</i> : 'is a Brussels-based independent non-profit making organisation dedicated to the reduction of the number and severity of transport crash injuries in Europe. ETSC provides an impartial source of expert advice on transport safety matters to the European Commission, the European Parliament, and Member States.'
FUA	Follow-up Admission: any hospital admissions after the first admission.

GIS	Geographical Information Systems: 'captures, stores, analyzes, manages, and presents data that refers to or is linked to location.'
GNP	<i>Gross National Product</i> : measure of national income and output (GNP = 'Value of all goods and services produced in a country in one year, plus income earned by its citizens abroad, minus income earned by foreigners in the country.'
GROS	<i>General Register Office for Scotland</i> : 'are responsible for the registration of births, marriages, civil partnerships, deaths, divorces, and adoptions' in Scotland.
ICD-9/10	International Classification of Diseases, 9th and 10th revisions: 'the international standard diagnostic classification for all general epidemiological, many health management purposes and clinical use.'
Injury	<i>Injury</i> : is in this context synonymous with trauma i.e. physical or emotional harm. Injury in this study is referred to as trauma inflicted by road traffic crashes.
ISD	Information Services Division, Scotland: 'national organisation for health information, statistics and IT services.'
KSI	<i>Killed or Seriously Injured</i> : abbreviation used in UK road casualty statistics based on police judgement of severity.
LoS	Length of hospital stay
MAIS	Maximum Abbreviated Injury Scale: if there are multiple injuries involved, as often is the case of RTCs the highest AIS (see above) is used as MAIS.
MRI	<i>Magnetic</i> Resonance Imaging: medical imaging technique to visualise structure and function of body. MRI is more detailed than CT.
MUARC	Monash University Accident Research Centre: 'is Australia's largest injury prevention specialist.' Conducting 'research, consultancy and training include safety in all modes of transport, in the workplace, in the community and in the home.
NHS	National Health Service: publicly funded health care in UK
OR	Odds Ratio: describes the strength of association and is often derived using logistic regression.
PCS	<i>Post-Concussion Syndrome</i> : a milder form of head injury with a range of symptoms.

PIC	Personal injury collision: term used to describe injury RTCs in safety camera project when analysing RTC blackspots.
PTSD	<i>Posttraumatic Stress Disorder</i> : an anxiety disorder that can develop after traumatic event with or without injury. It is a severe and ongoing emotional response to extreme psychological trauma.
RTCs	Road Traffic Crashes: this term has been selected as the most appropriate for this study and is used throughout the thesis. Other abbreviations commonly used elsewhere are MVA (Motor Vehicle Accidents), PIC (Personal Injury Collisions) and RTA (Road Traffic Accidents).
SIMD	Scottish Index of Multiple Deprivation: deprivation measure developed for Scotland comprising seven different domains (current income, employment, health, education, skills and training, housing, geographic access and crime).
SMR01	Scottish Morbidity Records 01: hospital discharge records in Scotland.
STATS19	Road traffic crash records collected in UK by the police, including information on crash circumstances, vehicle and casualties.
TRL	Transport Research Laboratory: UK base institution that provides independent research, consultancy, advice and testing for all aspects of transport.
WHO	<i>World Health Organisation</i> : a specialized agency of the United Nations (UN) that acts as a coordinating authority on international public health (copied from Wikipedia).

CHAPTER 1. OVERVIEW OF THE EPIDEMIOLOGY, COST AND PREVENTION OF ROAD TRAFFIC CRASHES

Contents of the Chapter 1

GLOBAL BURDEN OF INJURIES	
1.1 TIME TRENDS IN INCIDENCE OF ROAD TRAFFIC CRASHES	
1.1.1 Global time trends	
1.1.2 Time trends in Scotland	
1.2 RISK FACTORS AND CAUSES OF ROAD TRAFFIC CRASHES	
1.2.1 Age and gender patterns by road traffic crash risk	
1.2.1.1 Driving behaviour related to age and gender	
1.2.1.2 Exposure to road traffic crash risk related to gender and age	
1.2.1.3 Risk related to children and elderly	
1.2.2 Substance use and road traffic crash risk	
1.2.2.1 Alcohol intake and risk of road traffic crash	
1.2.2.2 Police detection of drink driving	
1.2.2.3 Drink driving involvement and age	
1.2.2.5 Smoking and risk of road traffic crash	
1.2.3 Vehicle speed and road traffic crash risk	
1.2.3.1 Definition of speed-related road traffic crashes	
1.2.3.2 Speed related to road traffic crash incidence	
1.2.3.3 Speeding drivers	
1.2.4 Deprivation and road traffic crash risk	
1.2.5 Seasonality and road traffic crash risk	
1.2.6 Psychopathology and road traffic crash risk	
1.2.7 Previous road traffic crash or injury and road traffic crash risk	
1.3 CONSEQUENCES OF ROAD TRAFFIC CRASHES	
1.3.1 Road traffic crash injury and sequelae	
1.3.1.1 Life-threatening and disabling injuries	
1.3.1.2 Head and neck injuries resulting from road traffic crashes	
1.3.1.3 Long-term effects of road traffic crash injury	
1.3.2 Risk factors associated with injury severity	
1.3.2.1 Age and gender as risk of road traffic crash consequences	
1.3.2.2 Alconol and risk of road traffic crash consequences.	
1.3.2.3 Type of todu user and risk of road traffic crash consequences	
1.3.2.5 Other factors relating to risk of road traffic crash consequences.	
1.3.3 Psychosocial sequelae	
1.3.3.1 Prevalence of psychosocial disorders following road traffic crashes	
1.3.3.2 Psychosocial long-term effects following road traffic crashes	
1.3.4 Risk factors associated with psychosocial disorders following a road traffic cra	sh
1.4 VALIDITY OF ROAD TRAFFIC CASUALTY REPORTING	
1.4.1 Characteristics of casualties missed by the police	
1.4.2 Recommendations on road traffic casualty linkage	
1.4.3 Linking hospital and police records of road traffic casualties worldwide	
1.4.4 Linking hospital and police records of road traffic casualties in UK	
1.5 COSTS OF ROAD TRAFFIC CRASHES	
1.5.1 Types of costs incurred by road traffic crashes	
1.5.2 Costs of road traffic crash injuries	
1.6 NATIONAL ROAD SAFETY POLICY	
1.7 PREVENTION OF ROAD TRAFFIC CRASHES	
1.7.1 The three E's in prevention: Engineering, Education and Enforcement	
1.7.2 Types of road traffic crash prevention	
1.7.2.1 Red light cameras	44
1.7.2.2 Moving radar	
1.7.2.3 Changing speed limits	
1.7.2.4 Koad engineering	
1.7.2.5 Addptive speed infiners	
17.3 Speed cameras	
1.7.3.1 Speed cameras in Australia	
1.7.3.2 Speed cameras in New Zealand	
1.7.3.3 Speed cameras in Europe	
1.7.3.4 Speed cameras in UK	
1.7.3.5 New speed camera strategy in UK: Safety cameras	
1.7.3.6 Safety cameras in Strathclyde region	
1.7.3.7 Summary of five speed camera evaluations	

Global burden of injuries

Injury is a worldwide problem [1]. It accounts for a large proportion (9%) of overall mortality and global burden of disease (12%), estimated using disability-adjusted life years (DALYs) or healthy years lost due to premature death or disability [2]. Road traffic crashes (RTCs) account for about 25% of all fatal injuries in the world and 22% of injury-related DALYs. RTCs especially affect young people - over 50% of RTC fatalities occur in young adults (aged 15-44). In 2000, an estimated 1.26 million people died in RTCs and this is only the tip of the iceberg, as RTC injuries affect between 20 and 50 millions every year (the wide range of this estimate is because of the known under-reporting of casualties) [3]. In total, it is estimated (conservatively) that between the first fatal road traffic crash in 1896 (a pedestrian in London hit at 4 mph) [4], and the end of 1997, 25 million people have lost their lives in RTCs [1]. According to the World Health Organization (WHO), globally, an average of 3,242 people lose their lives through traffic injuries every day [1] (a yearly toll of 1,183,330 fatalities). In the European Union more than 50,000 people are killed and 150,000 disabled for life, in RTCs every year. This leads to 200,000 families who are either bereaved or have a family member who is disabled for life [5].

RTC injury rates vary immensely between different parts of the world and 90% of fatalities occur in the low and middle-income countries [2]. Africa and Latin America have high RTC mortality rates (e.g. El Salvador: 41.7 per 100,000 population) [1], which stands in stark contrast to the Netherlands, Sweden and the United Kingdom (UK), which have the lowest rates (6.8, 6.7 and 5.9 per 100,000 population respectively) [6].

Pedestrians and two-wheeler users (bicyclists, occupants of mopeds and motorbikes) are especially vulnerable in traffic and carry a heavy burden of the injury total, particularly in low- and middle-income countries (due to more mixing with other road users and a greater diversity and intensity of traffic) [1]. A review of 24 studies of developing countries showed that in 16 of these studies pedestrians accounted for between 41% and 75% of the total RTC fatalities [7]. In high-income countries the proportions are different and in France, Germany and Sweden car occupants comprise

more than 60% of RTC fatalities, while countries such as Japan, Denmark and UK have less than 47% car occupant fatalities of the total (these nations have a higher rate of pedestrian fatalities (over 20%)) [8]. In India 26% of all RTC fatalities occurred in bicyclists (1993) [7] compared to 22% in Netherlands of (1990)[9], and only 2% in United States of America (USA, 1995) [9].

RTC is one of the major causes of trauma admissions. In a review of 15 studies of developing countries, between 30 and 86% of trauma admissions were due to RTCs [7]. A US study showed that out of 80,000 children who were involved in an RTC 12,800 (16%), received some form of health care [10]. There are major consequences to RTCs including both physical and psychological disabilities. Children appear to recover physically faster than adults [11], but they suffer a high rate of psychological problems [12,13]. A study of emergency department attendees, following RTC, showed that 55% of people still suffered from the consequences a year after the RTC [14]. The people who were seriously injured in this study were twice as likely to experience travel anxiety, post traumatic stress disorder (PTSD), work and financial consequences, a year after the RTC.

1.1 Time trends in incidence of road traffic crashes

RTC injuries have been a part of society since the introduction of the moving vehicles in the end of the 19th century. The injury rates increased until the 1960s to 1970s when the trend turned in developed countries. The turnaround in trends was probably mainly due to a wide range of road safety measures (e.g. seatbelt use traffic calming interventions etc.) [1]. This section includes a review of the current global time trends comprising future estimates and time trends in Scotland.

1.1.1 Global time trends

The total count of RTCs and its consequences in terms of injuries and fatalities is continuously rising in the world, although there is a clear difference in time trends in high and low-income countries. High income countries (including UK) have experienced a decline in fatal RTC incidence since the 1960s, while there has been an increase in low and middle income countries [1,15].

A research study by the World Bank [15] estimated that the future count of the world's road fatalities will increase by about 66 percent between 2000 and 2020, but this will be divided very unequally across the countries where a 28% decline is expected in the high-income countries while countries like China and India can expect major increases (92 and 147% respectively). By 2020 high-income countries are predicted to have less than 8 fatalities per 100,000 pop while low-income countries have nearly 20 per 100,000 pop.

1.1.2 Time trends in Scotland

In Scotland, RTCs (in which someone was killed or injured) increased up until the mid 1960s where it reached a plateau (at about 23,000 RTC per year) until the end of the 1970s [16]. There was only a temporary dip in the RTC count during fuel crises in the early 1970s. From the early 1980s onwards there has been a declining trend of RTCs and in 2006 there were 13,000 injury RTCs. In terms of casualties, comparing casualty counts in 1996 with 2006, this shows a reduction in all casualty severity categories: fatalities (357 vs. 314), serious (4,041 vs. 2,625) and total (21,716 vs. 17,267).

1.2 Risk factors and causes of road traffic crashes

There are a variety of factors associated with the occurrence of RTCs. The risk factors can, broadly, be categorised in four groups: factors influencing exposure to risk (e.g. deprivation), crash involvement (young male), crash severity (speed or seatbelt use) and severity of injuries after RTC (lack of early care) [1].

In this section, risk factors associated with exposure to risk and crash involvement are discussed. Risk factors associated with crash and injury severity are discussed further in section 1.3.4 (including type of road user).

1.2.1 Age and gender patterns by road traffic crash risk

The disproportionate male to female ratio, in terms of RTCs, is recognised world-wide [17]. According to Australian figures males were involved in 63% of total crashes and 86% of alcohol related crashes although males comprise only half of the driving population [18].

In Scotland, males in the age group 17-22 and 30-59 had double the driver RTC rate compared with females from the same age group (11 per thousand population, versus 5.1 and 6.2 per thousand population in 1999). During 1981-85 the rate of RTCs involving female drivers increased while RTCs involving male drivers decreased and the ratio fell from 4:1 (1981-85) to 2:1 (1999) [19].

Females are more involved in non-fatal than in fatal crashes which may reflect the fact that females are more often injured in urban crashes at lower speed. There is also a higher rate of females involved in crashes in high-income countries than in low-income countries [3].

1.2.1.1 Driving behaviour related to age and gender

The variation in male/female death rates is likely to be caused by difference in speed, alcohol use and other factors that influence the outcome in serious and fatal crashes [17,20]. Males are generally considered to be more inclined to take risks than females including alcohol and speed [20-23] and younger drivers are more likely to engage in more risky behaviour (such as speeding and drink driving, see further section 1.2.2.3) [20-25]. Young drivers are also more confident in their ability and perceive some driving situations less hazardous than older drivers [22].

Behavioural risks in terms of psychology are discussed further in section 1.2.6.

1.2.1.2 Exposure to road traffic crash risk related to gender and age

Some of the male/female differences could possibly be explained through different rates of exposure to risk [26,27]. A US study of crashes and gender, involving over 11 million crashes, found that the reason for greater death rates in male drivers compared to female drivers was the difference in miles driven (male/female ratio 1.74). Males have more fatalities per crash (male/female ratio 1.97[27]). Females, however, have more crashes per mile driven (male/female ratio 0.90 [27]) including both injury crashes and all reported crashes [26]. Other research has found that male drivers have a higher frequency of crashes involving single vehicles [25], which includes 'loss of control' of the vehicle, and these types of crashes are more likely to have a fatal outcome [28]. In terms of driver hospitalisations, over double as many males as females were involved in 'loss of control' type of crashes compared to crashes with other vehicles (where the rate war nearly equal between males and females) [28].

Males are more likely to have a crash at night-time than females (1.2 time the female rate) while females more often have crashes during daytime hours (1.4 times the male rate) [26]

Younger drivers have a high crash rate per mile driven, but a lower rate of death per crash. This could be due to the fact that younger people are better protected because of their overall better health [27]. The fatal and serious injury rates in Scotland were 1.12 per thousand populations for 16-22 year old car users while only 0.42 per thousand populations for car users 23-59 year old [29].

1.2.1.3 Risk related to children and elderly

Children in the developing world are at higher risk of RTC injury than in the developed world, but they account for a relatively small proportion of the reported casualties [3]. In contrast, children in USA and UK account for a higher proportion of RTC casualties than other countries. UK has three times as many casualties under the age of nine compared to Italy. Reasons behind this include social patterns (walking to school etc) and population distributions. In Scotland, 18% of RTC emergency admissions were children (2004). Children had, on average, a shorter length of stay than adults (3 days v 4.1 days) and the total inpatient days due to RTC for the year was 18,166 [30]. Boys (0-14) had over double the fatal injury risk of girls in RTCs in Scotland (OR 2.38 in 2002-06), while equal risk of a fatal pedestrian RTC in the same time period; the fatal child pedestrian m:f ratio declined from 2.03 in 1982-86 to 0.95 in 2002-06 [31].

Older drivers have a high crash rate per mile driven, but research indicates that older drivers primarily are a risk to themselves rather than other age-groups [32]. There is on the other hand, research indicating that the claim that older people have a higher crash rate per mile driven may be explained by the "low mileage bias", which means that drivers, independent of age, have a higher risk of RTC when driving a lower annual mileage [33].

Older people have a higher risk of severe or fatal injury outcome from RTC due to their vulnerability [33] and the risk of dying in a crash may be more related to this factor than to a decline of motor skills associated with aging [32].

In terms of pedestrian RTCs, children under nine and people older than 50 are at greatest risk (18.8% and 49% respectively) [34]. In England and Wales there was a decline in pedestrian and cycle child-fatalities in 1985-1992. One of the main reasons for this decline, however, was that children travelled more by car than walked or cycled [35]. Child pedestrians are more often hit by male drivers under the age of 40 [36] and it has been suggested that the pedestrian is at fault in a third of the collisions [34].

Elderly and RTC risk is discussed further in section 1.2.6 on psychopathology.

1.2.2 Substance use and road traffic crash risk

The effects of alcohol are well researched but there are fewer data on other substance use (including medical and recreational drugs) in relation to RTC [1]. It is more difficult to assess the impact of

these, as measuring drugs is not as straightforward as with alcohol [37,38]. According to WHO there is currently no strong evidence that drugs increase crash risk significantly and they call for studies to assess this urgently [1]. This subject will therefore not be explored further here.

1.2.2.1 Alcohol intake and risk of road traffic crash

The risk of involvement in a crash increases significantly at blood alcohol concentration (BAC) above 0.04 g/dl. [39]. The crash risk increases by at least 9 times at 0.05-0.09 BAC and at very high BAC (\geq =0.15) the crash risk increases 3-600 fold compared to zero BAC [40]. The risk of a fatal crash is 4.8 times more common for young drivers (aged 16-20) with alcohol levels of BAC equal to or over 0.08, compared to no alcohol [41].

In the 'SUN' countries (Sweden, UK and the Netherlands) the proportions of fatally injured drivers with a BAC over 0.1 is 14%, 20% and 17% respectively, which may reflect each country's legal limits (0.02, 0.08 and 0.05 g/dl respectively), penalties and enforcement [42]. In the UK (1999) 48% of all fatally injured adult pedestrians and 33% of all fatally injured drivers, tested positive for alcohol [43].

1.2.2.2 Police detection of drink driving

An Australian study showed that police detected 80% of drink driving incidents between 6 pm and 6am. During daytime, however, there was a higher risk of alcohol detection through RTC investigation rather than routine enforcement [18]. This both reflects the fact that police enforcement is more likely to occur during evenings and weekends when alcohol consumption is higher and that more RTCs occur during the daytime when more cars are on the roads.

There has been a reduction in RTC related drink driving arrests in Scotland from 3.8% in 1986 to 2.1% in 1999. The number of drink drive crashes fell 35% between 1988 and 1999 and the numbers

of casualties fell by 39% over the same period. There are no estimates of RTC involvement of drivers influenced by alcohol lower than the legal limit of 0.08 BAC [29].

1.2.2.3 Drink driving involvement and age

Driving under the influence of alcohol is more common among younger people [18,23]. In an Australian study, drivers aged 18 to 25 years were involved in half of all the drink drive arrests that were detected through a crash. Additionally, alcohol detection following a road crash was more likely for old and young drink drivers than middle-aged drink drivers. In this study, drink-driving RTCs comprised 2% of all reported crashes, but according to the study this should be seen as an under-estimation of the true level of alcohol related crashes [18].

1.2.2.4 Alcohol and risk of pedestrian road traffic crash

An Australian study found that 46% of pedestrian fatalities aged 16 or over had consumed alcohol and, in total, 59% of males and 17% of female fatalities had a detectable BAC.[44]

A study from USA of pedestrian casualties showed that they were more likely to be binge drinkers, alcohol dependent and drug dependent than compared to the remaining population of unintentional trauma [45].

In a Swedish study, alcohol was detected in 19% of pedestrian fatalities and males tested positive more often than females (24% v 11%) [46] and the mean age of the fatalities with detected alcohol was lower than the remaining group (49 years v 59 years). Additionally a greater proportion of victims testing positive occurred during nights and weekends [46], which is similar to other findings [47].

In Scotland, 31% of all pedestrian casualties had consumed alcohol and of these, 87% were male. Pedestrian casualties were twice as likely to be admitted to hospital if they have been drinking and pedestrians aged 40-49 were much more likely to be in a RTC if they had been drinking than other age groups [47].

1.2.2.5 Smoking and risk of road traffic crash

The relationship between smoking and RTCs was analysed in an Israeli study. Cigarette smokers made up 28% of study population, while 40% of drivers in RTC were smokers (p=0.005). The frequency of being involved in RTC as drivers was 0.69% for smokers and 0.4% for non-smokers. The reason for this, suggest the authors, could be that a driver who smokes is distracted from paying attention to driving. Smokers may also have a greater risk of an underlying disease that could affect their performance. Additionally, a "person who risks his health by smoking might be a less cautious driver as well". [48]. This statement is supported by a more recent study showing that smokers are more likely to drink drive, binge drink, travelling with a drunk driver, having a previous RTC injury and be less likely to use seatbelts [49].

1.2.3 Vehicle speed and road traffic crash risk

Speed influences the risk of RTC in several ways. With higher speed, a driver's time to react is shorter, it is easier to lose control of a vehicle and it is harder for surrounding road users to judge the speed of an oncoming vehicle (with high speed) and have time to avoid it. The stopping distance of a vehicle also increases with higher speed [50].

1.2.3.1 Definition of speed-related road traffic crashes

A major factor in both frequency of occurrence of RTC and severity of resultant injuries includes inappropriate and excessive speed (driving too fast for prevailing conditions and surpassing the speed limit exceedingly).

A UK report states that approximately 15% of crashes are reported as having "excessive speed" as a contributory factor [51]. Several other factors are also regarded speed-related, including 'following too close' or 'losing control in a bend' and it is therefore probable, that speed influences more RTCs than the 'excessive speed' factor alone indicates. Combining 'excessive speed' with relevant additional factors, suggests that speeding causes one third of fatal RTCs.

1.2.3.2 Speed related to road traffic crash incidence

It is estimated that the greatest reduction in road casualties could be achieved through reducing the speed of the fastest drivers. Furthermore, RTCs would increase by 10% if the proportion of speeders doubled and if the average speed increased by 1 mph, RTCs would go up by 19% [52]. RTC frequencies are also expected to drop by 5% if the average speed is reduced by 1 mph, but this general rule varies according to type of road.

In UK urban areas 4% of RTCs were due to speeding while an additional 21 % had other speed related factors. Findings also suggest that if everyone kept within the speed limit, the RTC rate would drop by 20% [53]. In a 1994 TRL investigated the impact of traffic speed on the number RTCs. The findings suggested a 5% increase in injury RTCs per 1 mph increase in average speed [54].

1.2.3.3 Speeding drivers

Speeders (drivers flashed by a speed camera or stopped by police) have been found to be 'RTC magnets' i.e. more prone to be in a RTC [55]. In terms of speed cameras, drivers are more likely to get caught speeding further away from their home: only 15% of people caught speeding by a fixed camera live within 2 km of the camera site and 63% live further away than 15 km [56].

1.2.4 Deprivation and road traffic crash risk

It is well known that deprivation is associated with injury risk. A study of pedestrian casualties in USA showed that these casualties were more likely to be black, unemployed, unmarried, alcohol or drug dependent, low income and educational achievement, not have a driving licence and of younger age [45].

A Swedish study of childhood injuries found that socio-economic differences in traffic injuries increase with the child's age and peak at ages 10-14 and 15-19 [57]. A study of child injury fatalities in England and Wales showed that child pedestrian from social class V are five times more at risk of a fatal injury than children in social class I [58], which is similar to a Swedish study showing a 20-30% higher risk of child pedestrian and bicyclists from manual worker families compared to children from high and intermediate level salaried families [59].

A study of young drivers found that there was 80% higher risk for young drivers from manual worker families of injury in traffic compared to young drivers with parents of higher socioeconomic status [60].

A Scottish study of pedestrian road traffic casualties found that residents from the 15% most deprived areas were nearly four times as likely to be in an RTC than residents from the 15% most affluent areas (19.9 versus 5.1 per 10.000 people per year) [61].

Lower socioeconomic levels are also related to worse injury severity [62].

1.2.5 Seasonality and road traffic crash risk

There is a seasonal trend to RTCs and in Scotland adult car user casualties vary by month and the annual averages over the years 1995-99 showed a peak in October-November. This peak had 31% more 'adult car user casualties' than April, which had the lowest rate. The peak time of day for adult car user casualties was from 4 to 6 pm [29].

1.2.6 Psychopathology and road traffic crash risk

Social problems at a young age can be associated with risky driver behaviour. A study showed that a history of school suspension carried an increased risk of repeated RTCs injury [63]. The risky behaviour included not using a seatbelt, drink driving, riding with a drunk driver, binge drinking, and speeding for excitement. Having a history of school suspension was also associated with drink driving convictions and suspension of license.

Personality disorders (borderline and/or antisocial) were associated with higher risk of a fatal RTC outcome for males aged 26 or over (in a study of males only) [64]. Young drivers with high levels of aggression or alienation from society may carry an increased risk of RTC [65].

Anxiety was associated with higher RTC incidence in a study of professional drivers [66].

There are indicators that cognitive impairment related to Alzheimer's disease is associated with pedestrian fatalities among older people [67] and that this group is often partially responsible for the RTC, injured in traffic situations of low complexity; involved in RTCs with reversing vehicles and injured in off road RTCs [68].

1.2.7 Previous road traffic crash or injury and road traffic crash risk

Having a previous injury can increase the risk of an RTC in different circumstances. People with a previous traumatic brain injury have double the risk of RTC compared to no injury, while those with brain injury after stroke incidents do not [69].

1.3 Consequences of road traffic crashes

The consequences of RTCs may be serious and long-lasting. They include the more tangible effects of injuries such as stay in hospital, disability and loss of income, but also less obvious psychosocial

effects. Children appear to physically recover faster than adults [11], but they suffer a high rate of psychological problems. One study showed that over half of RTC casualties still suffered from the consequences a year after the RTC [14] and casualties who were seriously injured in this study were twice as likely to experience travel anxiety, PTSD,, work and financial consequences a long time after the RTC. Age, alcohol, type of road user and vehicle speed all play a major part in how serious a RTC will be for an individual.

This section comprises three main parts. The first holds a discussion on injuries resulting from RTC including both characteristically differences between life threatening and disabling injuries as well as a review of head injuries and so called "whiplash" injuries. This also includes a discussion on the long-term effects of injuries.

The second part includes a review of risk factors that are associated with how severe a RTC may be for an individual including specific discussion on age and gender, alcohol, type of road user and speed of vehicle. Vehicle speed is of extra interest in this study as part of the assessment includes speed cameras.

The final part discusses the psychosocial effects of RTCs and the risk factors for RTC injury associated with psychosocial disorders.

1.3.1 Road traffic crash injury and sequelae

RTCs results in a range of physical injuries including everything from trivial to fatal. Fatalities constitute about 1% of casualties (results from this study, see further results section 4.1.1.1) and are only the tip of the injury ice-berg. As RTC injuries can be quite different in terms of threat to life and disability, many sophisticated scoring methods have been developed to aid in making fast and accurate decisions on how to prioritise injured casualties and how to best provide care.

1.3.1.1 Life-threatening and disabling injuries

Life threatening injuries usually differ from disabling injuries in various respects [70,71]; injuries to legs or neck are most disabling while injuries to the abdomen/pelvis, chest and head are more likely to be fatal [70]. For example, strain of the cervical spine (often referred to as "whiplash") constitutes the majority of car occupant injuries and, although these injuries score low on the abbreviated injury score (AIS 1, see glossary for explanation), they constitute a high proportion of insurance costs and after-effects including sick leave [11,34,72]. In a study of over 20,000 car occupants, nearly 10% of casualties with AIS1 sustained permanent medical impairment [73]. By contrast, if chest injuries with a severity of AIS 2-4 are survived, there are no or few disability consequences [71,73]. Furthermore, although hospitalised casualties result in longer periods of disability, most of the disability burden is carried by non-hospitalised casualties [74].

1.3.1.2 Head and neck injuries resulting from road traffic crashes

RTCs are the most common cause of severe head injury [75,76] and can cause substantial disability as well as less obvious effects such as residual cognitive impairment or emotional problems [77]. Milder forms of head injury consequences include post-concussion syndrome (PCS) consisting of headaches, fatigue, depression and anxiety, insomnia, concentration difficulties and emotional changes, such as irritability and mood swings [78]. There is, however, limited information the prevalence of PCS in RTC head injured people for several reasons. Firstly, these symptoms are common in the general population; secondly, it is difficult to diagnose with Computed Tomography (CT) or Magnetic Resonance Imaging (MRI) scans; lastly, findings using neuropsychological tests may be distorted by litigation i.e. the study-subject may be suing for compensation and could be motivated to perform under his/her ability in tests [79].

Cervical vertebral column (CVC) injuries (whiplash injuries) affect at least one person in 8 out of 10 RTCs involving two cars [34] and it more often affects women than men [34,72]. Women are also on sick-leave for longer from the consequences of CVC and about one in every nine people who suffer

CVC are still on sick-leave 4-6 years after the time of injury [72]. CVC victims may suffer from headache, thoracic and low back pain, fatigue, sleep disturbance and other ill health [80].

1.3.1.3 Long-term effects of road traffic crash injury

Children recover quicker from RTC injuries than adults and long-term effects increase significantly after the ages 40-50 (a study showed that 16% of children and 47% of adults suffered long term effects[11]). The physical impairments include incapacity to perform simple movements, needing to rest more, physically slower and tires more often than previously. The after-effects decline over time for adult casualties from about 60% at 6-9 months to 40% at 4-4.5 years post-RTC and more severe injuries cause a higher rate of long-term effects.

Long-term effects also include unemployment, lower incomes, learning performance, reduced scope for social contacts and difficulties with personal and domestic tasks [11].

A UK study showed that at three years post-RTC 5% of injured RTC casualties were having further surgery and 4% were still outpatients [81].

1.3.2 Risk factors associated with injury severity

There are several risk factors that contribute to the severity of injury and, according to WHO, the main ones are: lack of crash protection in-vehicle, insufficient roadside protection, not using protective devices in vehicles, not using protective crash helmets, speeding and the consumption of alcohol [1]. Injury type and severity is also very dependent on what type of road user is involved and the most common injury among car occupants is CVC injuries [34] while the majority of head injuries are suffered by pedestrians.

1.3.2.1 Age and gender as risk of road traffic crash consequences

Males are overall approximately three times more likely to die in RTCs compared to females [2] and males have consistently higher mortality rates than females in all nations, regardless of income level or age group (even in terms of children aged 5-14) [1]. In high-income countries the age group 15-29 years have the highest RTC injury rate, while in low- and middle-income countries the over 60s have the highest rate [1]. According to WHO, in 2002 180,500 children were killed in RTCs and 97% of these were in low and middle income countries.

In contrast to incidence, females are at higher risk of more severe outcomes from RTC injuries than males [82]. They have an increased risk of a fatal traffic injury compared with males and studies have reported varying increased risk of fatality of up to 50% [83,84]. These findings, however, appear to be limited to younger females (age 20-35 [84]). The reason for this is not clear; it is possible that it could be due to the fact that females live, on average, six years longer than males. Males may therefore be less healthy than females at an older age i.e. an injury is more likely to be fatal [83]. Another theory holds that as the increased risk in relation to age, for women, is similar to the childbearing years (preteen to late 50's), it could be related to biological reproductive factors [84]. Furthermore, there have been to date no female crash dummies made, but a Swedish group of researchers are currently in the process of developing a uniquely female dummy [85].

Older drivers are more vulnerable to the "traumatic effect of crashes". Drivers aged 80 or over have a five times greater risk of a fatal injury than drivers aged 40-49 years. Younger drivers (<30 years), however, had less than 80% risk of a fatal crash compared to drivers aged 40-49 years [83].

1.3.2.2 Alcohol and risk of road traffic crash consequences

Alcohol consumption can both increase [86] and decrease the risk of a fatal outcome of an RTC depending on the alcohol concentration. The risk of a fatal RTC among drivers with a blood alcohol concentration (BAC) of <0.10 was 50% less than for sober drivers, while a BAC of >0.19 tripled the

risk of a fatal injury compared to sober drivers. However, BAC >0 and <0.05 is associated with 45% greater risk of driver errors than sober drivers and at BAC > 0.20 the risk increases by 700% [83].

1.3.2.3 Type of road user and risk of road traffic crash consequences

The injury consequences of RTC vary widely on several factors. In terms of type of road user, pedestrians and motorcyclist are more likely than other road users to suffer severe injuries [87], especially head injuries [88]. In a study in the European Union, motorized two-wheelers had a 20 times higher risk of being in a fatal RTC compared to car occupants. Cycling and walking carries about 7-9 times the fatality risk of car occupants [89]. Public transport is the safest option and travelling by bus carries a 10 times lower risk of a fatal RTC than travelling by car.

The majority of killed pedestrians die of an injury to the head and pedestrian injuries, in general, are dominated by knee/lower leg and head injuries [34,90,91]. Pedestrians are also more likely to sustain pelvic injuries than other road users. Car occupants on the other hand are more likely to sustain abdominal and spinal cord injuries [88].

Different injury characteristics, according to type of road user, also affect the time of death in relation to the crash; car occupants are more likely to die at the site of the crash due to ruptured thoracic aorta while pedestrian deaths more often occur in hospital through a head injury [91].

Pedestrians are more likely to die in RTCs than other casualties [88]. In 1999 pedestrians made up 18% of all casualties in Scotland and 30% of those were fatal or seriously injured compared to 19% of the remaining casualties [29].

1.3.2.4 Vehicle speed and risk of road traffic crash consequences

Severity of injury is highly related to speed of the vehicle. The probability of a belted car occupant (in a front seat) becoming severely injured triples as speed increases from 20 to 30 mph, and at 40 mph it

is five times greater [92]. For pedestrians there is a significant change between 30 and 40 mph where mainly non-fatal injuries become mostly fatal [93].

Two thirds of all crashes resulting in death or serious injury (KSI) occur on roads with a speed limit of 30 mph or less [94]. If everyone adhered to the speed limit, it is estimated that the RTC rates would drop by 20% [53].

A study of fatal RTC data found that a very high proportion of the vehicles involved in fatal RTCs had exceeded the speed limit excessively, although those vehicles made up a very small proportion of the traffic in total. For example, drivers speeding at 50 mph, on a 30 mph urban road, make up about 1% of total traffic but 7% of all fatal-RTC-involved vehicles [51].

Driving at 70 mph or more increases the odds of fatality by 164% compared with speeds less than 35 mph [83] and a US study has estimated that if speed limits were set to <=65 mph nationwide (USA) 3,000 lives would be saved every year [95].

1.3.2.5 Other factors relating to risk of road traffic crash consequences

The weight of a casualty is related to risk of serious injury and fatality. The odds ratio for fatality was 1.013 for each kilo increase in body weight and 1.08 for serious injury. This could, partly, be due to other 'co-morbid' factors in overweight people [96].

1.3.3 Psychosocial sequelae

Road traffic crashes can have major psychological impact, even if the physical injury was only minor or if there were no injury at all [97] and often cause posttraumatic stress disorder (PTSD), phobic travel anxiety, general anxiety or depression [14]. These problems are similar to what people suffer from through a large disaster e.g. the Lockerbie airplane crash. However, the support resources available for a 'run-of-the-mill' RTC-victim is not the same and after the physical injuries are treated
they are left to deal with psychological effects themselves [98]. On the other hand subjects accepting the responsibility of the RTC and even other peoples' death appear to suffer little distress or guilt during interviews [99].

1.3.3.1 Prevalence of psychosocial disorders following road traffic crashes

Common psychological after-effects includes frequent depression, sleeplessness and restlessness/nervousness and a slow-down of intellectual processes [11]. Children are often affected by posttraumatic stress disorder (PTSD) and a prevalence of 25% PTSD in RTC child-casualties [12] and 43% of PTSD and/or an indication of clinically significant depression and anxiety [13] have been recorded. Parents of child-casualties are also affected by PTSD (15%) and this is more prevalent in parents who witness or are involved in the RTC, but PTSD in children or parents is not related to injury severity [12]. The injury score AIS is not necessarily a good indicator of PTSD but patients' own ratings of the severity of their injury is highly correlated with PTSD [100]

1.3.3.2 Psychosocial long-term effects following road traffic crashes

At one year after the RTC 55% of respondents in a UK study said they still had medical, psychiatric, legal or social problems. The people who were seriously injured were twice as likely to experience travel anxiety, PTSD, work and financial consequences, a year after the RTC [14]. Similarly, it has been reported that two fifths of casualties suffered from anxiety or depression three months post-RTC (especially those suffering from multiple injuries) [99]. Nearly a quarter of casualties had psychiatric problems after one year, and during the first year after the RTC 11% of the subjects satisfied DSM-II-R criteria (Diagnostic and Statistical Manual of Mental Disorders) for post-traumatic stress disorder [99]. A fifth of the casualties suffered from travel anxiety that was disabling which is similar to the findings of a Canadian study that 18% of people had a fear of driving 3-4 years post-RTC [101].

1.3.4 Risk factors associated with psychosocial disorders following a road traffic crash

Female passengers suffers a higher level of phobic travel anxiety than male passengers after a RTC [14].

Motorcyclists are more likely to suffer psychiatric disorders than other casualties, which could be due to more severe injuries and social vulnerability [99].

A previous history of major depression is a risk factor for developing PTSD after an RTC [102]. Seriously injured RTC casualties are twice as likely to experience travel anxiety, PTSD, work and financial consequences, a year after the accident [14]. Casualties with severity score >4 have more financial and physical problems than others and they are more likely to suffer depression [81].

1.4 Validity of road traffic casualty reporting

There is growing evidence that the sources of information on RTC casualties are incomplete and especially police recorded data [103-105]. Police records are the main source of information on RTCs in most countries although it may be misleading to rely only on police data [106-110] especially in terms of non-fatal injuries [111]. A special cause for concern is where police data are used to evaluate severity level of RTCs, as the judgement of severity by a police officer is not necessarily accurate [111-113].

This section begins with a summary of what type of casualties are missed by the police, followed by a review of official recommendations on linking police and hospital RTC records. The section ends with a review of exercises in linking hospital and police records worldwide and in UK.

1.4.1 Characteristics of casualties missed by the police

Severe RTCs in UK are estimated to be underreported by as much as 50% and it has been hypothesised that this has risen over later years [103] as the decrease in STATS19 serious RTC casualties is much steeper than in RTC hospital admissions [110,114].

RTCs in motorcyclists and pedal cyclists are most likely to be unreported by the police, especially where no other vehicles were involved [109,113,115-118]. Underreporting has also been linked to pedestrians [113] and children [113,116], as well as RTC casualties claiming financial compensation and casualties that had been in control of a motor vehicle [109].

Information on severe RTCs is more common than for RTC causing minor injuries [117] although these data are necessary in order to do research on "whiplash" injuries [119], which is a major burden both from an economic and human suffering perspective [72,80].

1.4.2 Recommendations on road traffic casualty linkage

To overcome the problem of incomplete RTC registers, the UK Department for Transport (DfT) recommends that periodic comparisons of STATS19 and hospital inpatient data on road casualties should be carried out. DfT propose that one-to-one matching of STATS19 and hospital inpatient records be performed to try to validate the time trends of more seriously injured casualties reported by STATS19 [120]. Such a linkage is currently in process for England [121].

In the light of evidence that linked databases are less biased than those based on police records alone [108] the European Transport Safety Council (ETSC) recommends linked databases as the basis for injury reporting in order to meet the needs of EU road safety policy support [122]. The ETSC asserts that "information about injuries has to come from hospital records but additional, linked data are needed" and that the objective of these databases is "monitoring of reporting completeness and injury patterns and identifying road safety priorities".

One of the main problems with linking data is sharing different data sources. There are usually legal limitations and other restrictions that need to be overcome [1] apart from data sources often not being synchronised.

Linking data aims to achieve a more complete and comprehensive view to the complex nature of RTCs, but there is also a cost-beneficial aspect of utilising several data sets in road traffic injury prevention; the costs incurred by linking or using several data sets, is far lower than the costs associated with misjudging RTC statistics [110].

1.4.3 Linking hospital and police records of road traffic casualties worldwide

Linkage exercises have been attempted for some time in many countries [117,123-127] including UK [118,121,128,129]. In Australia an iterative procedure including name identifiers was implemented for ten years .[117,127]. In a European project linking hospital and police data (PENDANT), three countries were included using separate methodologies and different information [125]: France utilised a manual method resulting in a small number of records; the Netherlands linked police and hospital data for the whole nation over seven consecutive years (and reported substantial problems with levels of recording of police records [123]); Spain linked police and emergency department records from Barcelona for one year. Sweden has come some way in completing a full national linked police-hospital database, which to date includes two-thirds of the country [124].

1.4.4 Linking hospital and police records of road traffic casualties in UK

In UK, various linkages have been performed including one London based study linking police and emergency department data in three hospitals to estimate the extent of RTC injuries across the city [128]. A further two studies carried out by the Transport Research Laboratory (TRL, [118,130]) used a national sample of 16 and 18 hospitals collecting emergency department and inpatient data that were linked to STATS19 over one and three years respectively. The TRL data linkage did not, however, include the full medical database [123]. In 1994-94, the TRL linked over 8,000 English records from the Trauma Audit and Research Network (TARN) [123,129]. Lastly, the TRL also linked police and hospital data from three hospitals in Manchester 1993 for six months [131].

In Scotland, the TRL performed a substantive linkage of a sample of casualties over 16 years (1980-1995) providing an overview of trends in clinical data including International Classification of Disease (ICD) codes translated to Maximum Abbreviated Injury Scale code (MAIS) for each of the six body regions and length of stay in hospital [132].

The different linkages produced varying linkage rates (% of hospital records that linked to a police record): Australia 64% [117], England 70-87% [128] and 46% [118].

1.5 Costs of road traffic crashes

Although the sequelae of RTCs, in terms of immediate injury impact, have been investigated in great detail [34,70,91], little is known about the long-term health and economic consequences of RTCs, and the financial costs are probably greatly underestimated [133].

Estimating the economic impact of road traffic crashes is important. It aids in comparing the RTC impact with other causes of mortality and morbidity, as well as providing a basis for cost benefit analysis of interventions and promoting understanding of the scale of the problem [1].

The remainder of this section includes a review of where different costs are derived from followed by what cost estimates include and cost by injury and road user.

1.5.1 Types of costs incurred by road traffic crashes

Valuating RTCs economically is usually based on a combination of tangible costs, such as hospital costs (direct cost of injury), long-term care costs, loss of productivity or cost of vehicle damage, and

more indirect costs such as an evaluation of human suffering and loss of life. Many of these methods are difficult and controversial. One method is called "willingness-to-pay", which estimates what people would like to pay for avoiding injury or death and another compares lost life with lost earnings and is referred to a the "human capital" method [1].

The UK cost estimates of RTC comprise several elements. The value of preventing a casualty has been estimated by Department for Transport [134] to include the following:

- loss of output due to injury (earnings and employers costs etc).
- ambulance and of hospital treatment costs.
- human costs, based on "willingness to pay" values (representing "pain, grief and suffering to the casualty, relatives and friends, and, for fatal casualties, the intrinsic loss of enjoyment of life over and above the consumption of goods and services").

Additionally, costs for crashes also include:

- costs of damage to vehicles and property.
- costs of police and insurance.

1.5.2 Costs of road traffic crash injuries

Costs of road traffic crashes vary between countries –from as low as 0.3% of Gross National Product (GNP) to over 4%. In high-income countries RTC costs make up a larger proportion of GNP (about 2%) than in low-income countries (about 1%) [3].

Costs associated with RTC injuries are often higher than other types of injury [135] and the highest RTC injury hospital costs are incurred by injuries to the lower extremities (hip/thigh and knee/lower leg) and head injuries. Estimates from Australia suggests these make up 33% and 27% respectively

[136], similar to USA findings [137]. Both of these studies also found that severe spinal cord and brain injuries cost more per case.

Hospitalised pedestrians RTC casualties incur high costs; in a study from New Zeeland they comprised 10% of casualties but incurred 18% of total costs [138].

A linkage exercise completed in Italy (linking hospital and police RTC records) estimates that the costs of RTC casualties make up 1.3% of total hospital costs [126].

1.6 National road safety policy

Several countries set national targets in order to help place road safety higher on the national agendas of priorities. Targets help secure resources and to evaluate progress. The targets should preferably be long-term and easy to measure. Some examples of fatality reduction targets are: Australia -10% (8 years), USA -20% (12 years), European Union -50% (10 years), Finland -37% (10 years), France 50% (5 years) and Sweden 50% (11 years) [1].

In 2000, the UK government set national targets for reducing casualties by 2010. The reduction targets and the achievement in 2006 (in brackets) were as follows (quoted directly from Road Accidents Scotland 2006 [16]):

By 2010 it is hoped that there will be, compared with the average for 1994-98:

- a 40% reduction in the number of people killed or seriously injured in road accidents (30,6%).
- a 50% reduction in the number of children killed or seriously injured (39,0%).
- a 10% reduction in the slight casualty rate, expressed as the number of people slightly injured per 100 million vehicle kilometres (7,2%).

The report, however, acknowledge that there may be problems with the reporting of casualties to the police and therefore STATS19 (police RTC recording) may not be a reliable source to use when estimating RTCs.

1.7 Prevention of road traffic crashes

Preventing RTCs has been an ongoing challenge since the start of the moving vehicles, beginning through legislation in UK: a ban on riding on footpaths in 1835 and a speed limit on self-propelled vehicles of 4mph in the country and 2mph in towns in 1865 [139]. Standardised road signs have been in place since early 1900s and the first automatic traffic lights were installed in the mid 1920s.

This section covers a short review of the three E's of injury prevention (engineering, education and enforcement). This is followed by a section on specific RTC prevention measures including highly successful measures such as road humps and daytime running lights. The section ends with a detailed account of speed cameras, both worldwide and in UK.

1.7.1 The three E's in prevention: Engineering, Education and Enforcement

Today there is an immense collection of preventive measures to choose from, all relating to one or all of the three E's: *Engineering, Education* and *Enforcement*, as well as the more time related terms of *primary, secondary* and *tertiary* prevention. Primary prevention aims to avoid RTCs (e.g. speed cameras), secondary prevention aims to make RTCs less severe (e.g. seat belts) and tertiary prevention aims to reduce the negative impact from the RTC (e.g. hospital treatments). For example, engineering on primary level includes safer roads (e.g. footpaths) or good visibility in cars, while on secondary level engineering includes seat belts or side impact bars.

1.7.2 Types of road traffic crash prevention

The more successful interventions include seat belt laws (secondary engineering and enforcement prevention) and graduated driving licensing systems (primary educational and enforcement prevention). The seat belt laws enforced in the UK produced a 15% reduction in hospital admissions [140] and graduated driving licensing systems generates a crash reduction of 4-60%, the wide range may, according to WHO, be due to varying methodologies used [1].

The focus of the following sections is on primary enforcement and engineering prevention in traffic calming, as part of the study involves speed cameras. Speed cameras are discussed in section 1.7.3.

1.7.2.1 Red light cameras

In terms of camera enforcement there are fixed speed cameras and red light cameras, as well as moving police patrols with radars. Red light cameras were installed in Glasgow 1991-93 and the RTCs caused by red light running fell by a third including both injury and non-injury RTCs. The study also revealed that other factors must have contributed to the overall decline, since injury RTCs in the 'crossing carelessly' category declined by 54% (44% of the total reduction in injuries) over the same period [141].

1.7.2.2 Moving radar

The effectiveness of 'moving mode radar' speed enforcement over two years was assessed in Victoria, Australia. The project included up to 73 units and also comprised television advertising relevant to the mobile radar enforcement. The study found that there was a four-day residual enforcement effect on casualty crashes in rural areas, but the effect diminished five to seven days afterwards. In the region where the enforcement had been present, there was a 28% reduction of crashes during the first four days. Outer Metropolitan regions, however, showed no reduction in crashes when mobile radar was used [141,142].

Covert (unmarked car) mobile radar speed enforcement has been shown to be more effective in reducing casualty crashes than overt (marked car) enforcement. This was especially true in rural region of Victoria, during the 1 to 4 days after covert mobile radar enforcement had been in place. These circumstances gave a net 22% reduction in casualty crashes. The combination of overt and covert enforcement also gave a reduction, while using overt enforcement only was not as effective [143].

1.7.2.3 Changing speed limits

Victoria, Australia, also made changes to speed zones in order to be in line with the rest of the nation. The rationalisation meant that some speed zones were increased while others were decreased. Certain speed zone alterations resulted in highly significant casualty crash reductions i.e. the change from 100 to 80 km/h reduced casualty crashes by 46% and the decreased speed limit of 75 to 60 km/h resulted in a 43% reduction. Increasing the permitted speed from 75 to 80 km/h resulted in an increase in casualty crash frequency of 10.7% (highly significant) [144]. According to a policy review of speed management by Dft [145] speed limits by themselves have a very small impact on vehicle speeds. Research suggests that reducing speed limits, but not using any other intervention reduce the average speed by only a quarter of the reduction required e.g. only an average of 2.5 mph slower for a change of 40 to 30 mph limit [54].

1.7.2.4 Road engineering

Road humps, chicanes and other road engineering measures reduces speed in urban areas by 10 mph on average and are very effective at reducing child pedestrian casualties. Road engineering is far better in achieving lower speeds and RTC reduction than static signs alone [146].

Vehicle activated signs also appear to reduce speeds and the incidence of collisions [147]. These signs give a message about appropriate speed or an upcoming hazard and activate when an individual car is e.g. driving too fast towards a hazard such as a bend or junction.

Applying an edge-line to a road without a centre line appears to increase speed, while replacing a centreline with an edge-line decrease speed. The effects of adding an edge-line to a road with a centreline were unclear in this study [148].

1.7.2.5 Adaptive speed limiters

Another way of reducing the speed is, according to Dft [147], to employ adaptive speed limiters in vehicles. These are already fitted to heavy goods vehicles, which maximises their speed to 56 or 60 mph depending on how heavy they are. The technique of extending this, to adapt to any given road, already exists, and involves using a digital map in the vehicle and a global positioning system that, by satellites, can identify the location of a vehicle. These systems can either give drivers information about the speed limit, or be used together with an "adaptive speed control system". The benefits of this system could be great, potentially resulting in a 35% reduction in RTCs. This in turn would free police and courts to deal with other crimes [147].

1.7.2.6 Daytime running light

There is evidence that the use of daytime running lights (DRL) in cars and motorcycles might reduce RTCs. According to a European review [149], findings from different studies show that a reduction of 'multi-party' road traffic accident of 5-15% is achieved (car crashes) and that possibly 32% of motorcycle crashes are avoided. This review also included a cost-benefit calculation of how many lives could be saved, and injuries avoided, if DRL was introduced in the 12 EU nations that has not yet implemented mandatory DRL. This calculation is based on the assumptions that DRL reduces 15% of fatalities, 10% of serious injuries and 5% of slight injuries of 'multi-party daytime accidents'

(which is estimated to be 40% of the total). This could result in 2,359 fatalities, 17,507 serious injuries and 51,113 slight injuries avoided annually in Europe.

1.7.3 Speed cameras

Several studies evaluating speed cameras have concluded that they contribute to reducing the incidence of RTCs [150-157], fatalities and injuries [158-160] although the evidence so far is relatively weak [161]. No studies, to this author's knowledge, have reported in detail either the epidemiological impact or the type of crashes that are influenced by cameras.

1.7.3.1 Speed cameras in Australia

Australia has had speed cameras for a long time and back in 1989, the Victorian Government announced a new Road Safety Strategy to break the escalating RTC trend [162]. This included the introduction of red light cameras and slant radar speed cameras. The evaluation of this initiative, undertaken by Monash University Accident Research Centre (MUARC), included several factors believed to have contributed to the road trauma reduction i.e. increased random breath testing and speed cameras (both supported by publicity), reduced economic activity, reduced alcohol sales and improved the road system through treatment of RTC blackspots. Results showed that a combination of anti-speeding and drink-driving programs contributed to reductions in serious casualty crashes by an estimated 22-25%. Adding the RTC blackspot treatments to these reductions resulted in an overall decrease, through road safety initiatives, by 23% in 1990, rising to 30% in 1993-1996 [150,151].

These findings were challenged by White et al. Examining the RTC trends and the implementation of speed camera visually suggested that the major increase in speed camera traffic infringement notices did "not occur until at least two years after crash numbers had started to decline" [163]. The criticism was met with MUARC highlighting certain details in their reports. According to Cameron et al, the 'serious casualty crashes' (referred to by White et al) peaked in 1988 and represented only a third of all

crashes while the peak in 'casualty crashes' happened in 1989. Additionally Cameron et al state that "It is emphasised that MUARC's research has been confined to assessing factors which contributed to the reductions in road trauma in Victoria during the 1990s. MUARC has not evaluated the factors that may have been responsible for the turnaround in crashes of each level of severity prior to 1990" [164].

1.7.3.2 Speed cameras in New Zealand

New Zealand has used visible mobile speed cameras since 1993. They have been located in specific areas that have had many speed related crashes. In 1997, a trial began using hidden speed cameras in some 100 km/h roads. For the initial year of operation, the covert cameras (together with publicity) were associated with reduced speed, RTCs and casualties both at the camera sites and on other 100 km/h roads, while the overt cameras had a more localised effect [153].

1.7.3.3 Speed cameras in Europe

A Swedish study of speed cameras reported both a reduction in speed and the incidence of RTCs. There were a significant reduction at the black-spot site (to the date of the report), but also on the road as a whole (the total stretch between two cities). There was, however, a suspicion that this intervention would follow a similar pattern as a signpost intervention on the same road stretch, in 1990 (a light-sign stating that this is an 'Accident Blackspot'). The RTC rates dropped dramatically for two years (until 1992), but thereafter returned to the previous rates. As a conclusion, Kronberg et al, believed that there is a tendency for drivers to reduce speed initially while they learn how the speed cameras work/where they are and that the drivers have to be faced with new 'surprises' to keep them unsure about the extent of the surveillance [152].

A recent study in Barcelona found a reduced risk of RTC with rate ratio (RR) 0.73 comparing before and after camera installation [157]. It was estimated that 364 collisions were prevented and 507 fewer people injured during the two year following the camera installation.

A study in Norway of 64 camera sites showed a reduction in injury RTC of 20% and a reduction in 26% on roads which complied with both having a high density and rate of RTC [165] [165].

1.7.3.4 Speed cameras in UK

The first speed cameras in Britain were set up in West London 1992. These cameras were estimated to have reduced the incidence of road fatalities by 70%, seriously injured by 27% and slight injuries by 8% at the camera sites, during the initial three years of operation [158]. In early 1990s, the costs and benefits of traffic light and speed cameras were investigated involving 10 police forces in England and Wales and evaluating 78 red light camera sites and 174 speed camera sites [166]. All but one of the selected areas showed a decrease in RTCs post camera installation. RTCs were on average reduced by 18% for red-light cameras (0.48 RTCs per site per year) and by 28% for speed cameras (1.25 RTCs per site per year). In total, the red-light cameras were estimated to produce a potential reduction of 116 injury RTCs per year and the speed cameras possibly to have prevented 525 injury RTCs per year. The outcome of this project appears to have prepared the ground for the Safety Camera pilot projects (see later).

A study in Wales of mobile speed cameras at 101 sites resulted in a lower than expected frequency of injury crashes at the sites (51% reduction) and surrounding areas (up to 500 meters) and these reductions were sustained for two years after intervention [156].

In Cambridgeshire, a study was set up to develop a method to deal with seasonality and trend in evaluating speed cameras at the same time as it aided in distinguishing between real effects from the cameras and regression to the mean. The study observed a 31% decrease in injury RTC [154]. In a

continuation of the same study, it appeared that in the immediate vicinity of the cameras there were RTC reductions of nearly 46% but a lower reduction was found in the surrounding areas [155].

1.7.3.5 New speed camera strategy in UK: Safety cameras

Included in the Government's road safety strategy published in 2000 [167] was a cost-recovery component in safety camera and red light camera enforcement, which allows using money from fines for operational costs, road safety education and research. The UK safety camera pilot scheme began in April 2000 and utilises safety cameras in an attempt to reduce RTC rates in identified RTC black-spots. Camera location is determined by rigorous analysis of both personal injury collisions (PIC) and speeding history on RTC-prone roads. Positioning cameras based on these analyses serves both to avoid accusations that the cameras are used to make revenue and also 'to ensure that the maximum benefits are achieved' [168]. The pilot scheme was regarded as highly successful and was rolled out nationally in 2001-2002.

Safety cameras appear, according to recent research, very effective in reducing PICs in black-spot areas. Results, from the thee year evaluation [160], showed that there were 33% fewer injury collisions and 40% fewer people killed or seriously injured at the camera sites, when controlled for long-term trends. This includes a total of 870 fewer people killed or seriously injured and 4,030 fewer injury collisions.

In Strathclyde, it was reported that vehicle speed declined greatly once cameras were installed. Before the speed camera installation, 64% of vehicles exceeded the speed limit. Three months later, when the camera houses were set up, the proportion of speeding vehicles declined to 37%. This then declined further, to 23%, when the cameras were flashing [169].

1.7.3.6 Safety cameras in Strathclyde region

The national safety camera programme is operated locally by regional Safety Camera Partnerships between police and local authorities. Strathclyde safety camera partnership was one of eight pilot camera schemes established in UK in 2000. It initially only covered Glasgow City but has since expanded to include the local authorities of the whole of West of Scotland i.e. North Lanarkshire, South Lanarkshire, Argyll & Bute, East Dunbartonshire, West Dunbartonshire, Renfrewshire, East Renfrewshire, Inverclyde, East Ayrshire, North Ayrshire, South Ayrshire as well as Glasgow City. The region comprises over two fifths of Scotland's population - about 2.3 million - and suffers from a relatively high rate of social deprivation. Strathclyde region has a diverse combination of rural and urban roads. The area should formally be called Strathclyde Police Force Area, but is here referred to Strathclyde region for simplicity.

The criteria for safety camera installation have altered slightly over the years, but have always involved assessing evidence of the contribution of speeding to RTC rate and severity. The evaluation of the partnerships has involved assessing how they perform in terms of RTC reduction in a number of pilot areas in the UK. Included in the remit of the partnerships was an assessment of the impact of the cameras through research, which is why the collaboration with Glasgow University and National Health Service (NHS) Ayrshire & Arran was initiated in 2002 (leading to the 'context study' of this PhD, see below).

The study 'Health Impact of Safety Cameras' commenced in 2002 and initially aimed to consider the health impact (in its widest sense) of the introduction of camera sites of the pilot project in Glasgow and throughout the (then) Strathclyde partnership area. The study initially included 12 cameras subsequently expanded to include 48 camera sites. The Partnership currently operates 136 cameras (see later).

1.7.3.7 Summary of five speed camera evaluations

Summary of methods used in five speed camera studies are outlined in Table 1.

Author	Study design	Results	Limitations / solution
Hess S & Polak JW, UK, 2003	Before and after study including calculating severity weights.	Cameras showed the largest effect (46% decrease of weighted injury RTC) in the immediate vicinity (250 m), but also had significant effect on larger areas.	Long-term means were assessed in order to control for regression to the mean. Seasonality and time trends were also controlled for.
Christien S M et al. UK, 2003	Controlled before and after study of 101 mobile speed camera sites where two methods of assessing effectiveness were evaluated (circular zone around the camera sites and distance from sites).	Injury crashes were reduced with a significant rate ratio of 0.49 and cameras had a sustained effect two years after intervention.	Camera sites were matched with control sites on posted speed limit, road class and RTC. As a controlled before and after study, with well matched intervention and control sites, the sites will (if at all) be equally affected by regression to the mean.
Chen G et al. Canada, 2000	Before and after study of 30 mobile speed cameras over two years.	Daytime RTC reductions: 25% of RTC, 11% injuries (casualties collected by ambulance) and 17% reduction in fatalities.	Controlling for trend, seasonality, and amount of driving, but no assessment of regression to the mean.
Elvik R, Norway, 1997	Controlled before and after study of fixed speed cameras at 64 sites.	20% reduction in injury RTC (26% reduction on roads with both high RTC density and rates).	The empirical Bayesian method was used in controlling for general trends in the number of RTCs and regression to the mean.
Hooke A et al. UK, 1996	Before and after study including 174 camera sites with (usually) 3 years before installation from 9 different police forces.	Overall reduction in RTCs at sites at 28% (ranging 21% to 48% in police areas).	Different collection methods in police forces could affect number of RTCs included. No control for trends or other confounders.

Table 1 Methods, results and limitations of five speed camera studies

1.8 Summary of Chapter 1

- The time trends in road traffic crash (RTC) incidences in the UK are downward. Globally, however, it is estimated that by 2020 RTC fatalities will increase by 66% (compared to 2000).
- Several factors increase the risk of RTCs and these include being male, young, deprived, under the influence of alcohol and/or smoker. Underlying causes include exposure and risk taking. Aggression, anxiety disorder, a previous traumatic brain injury and Alzheimer's disease are also associated with a higher risk of RTCs.
- Speed influences the risk of RTC in many ways: shorter time available to react, easier to lose control, difficulty for other road users to react in time and longer stopping distance. Speeding may be defined as inappropriate speed (driving too fast for prevailing conditions) or excessive speed (surpassing the speed limit exceedingly). If vehicle speed were reduced, the incidence RTCs would drop (by an estimated 20% if everyone kept within the speed limit).
- Disabling injuries are typically different from life threatening injuries; injuries to legs or neck are most disabling while injuries to the abdomen/pelvis, chest and head are more likely to be fatal. Although hospitalised casualties have longer disability, the majority of disability burden is carried by non-hospitalised casualties. In general, children recover quicker than adults.
- RTCs are the most common cause of severe head injury and may result in a range of problems from severe disability to less obvious effects such as residual cognitive impairment or emotional problems. One of the most common injuries following an RTC is cervical vertebral column (CVC) injury (whiplash). CVC victims may suffer from headache, thoracic and low back pain, fatigue and sleep disturbance and other ill health. CVC may require several years sick-leave and more often affect females than males.

- Long-term effects from RTC injuries include physical impairments as well as unemployment, lower incomes, reduced learning performance, reduced scope for social contacts and difficulties with personal and domestic tasks.
- RTCs may have major psychological impact, even if the physical injury was only minor or if there was no injury at all. Psychological sequelae include posttraumatic stress disorder (PTSD), phobic travel anxiety, general anxiety and depression.
- Severity of injury is correlated with several variables including old age, being female, being an unprotected road user (e.g. pedestrian), personality disorders, non-use of seatbelt and speed. The combination of speed and being relatively unprotected makes motorized two-wheelers especially vulnerable in traffic and carries a heavy burden of the injury total.
- There is growing evidence that the routine statistics (especially police recorded data) on RTC casualties are incomplete. Minor injuries, RTCs with no third party involvement, and two-wheel road users are most likely to be missed by police. A special cause for concern is where police data are used to evaluate the severity level of RTCs, as the judgement of severity by a police officer is not necessarily accurate. A way of overcoming this problem is to link police records with hospital data, which has been done in trials in the UK. Linking records is recommended by a variety of authorities.
- Relatively little is known about the costs of RTCs and there are various ways of estimating these. Valuations are usually based on a combination of tangible costs, such as hospital costs (direct cost of injury), long-term care costs, loss of productivity or cost of vehicle damage, and more indirect costs such as an evaluation of human suffering and loss of life (e.g. the willingness-to-pay method).
- Costs associated with RTC injuries appear more costly than other types of injury, especially those incurred by injuries to the lower extremities and head injuries.

- Setting national targets is a common method in prioritising road safety on national agendas. In 2000, UK government set national targets for reducing casualties by 2010. These included a 40% reduction in all killed or seriously injured (50% for children) and a 10% reduction in the slight casualty rate.
- There are many countermeasures in combating RTC injury. These can be categorised under one or all of the so-called three Es: *Engineering, Education* and *Enforcement*, as well as the time related terms of *primary, secondary* and *tertiary* prevention. Initiatives include seatbelt use, changing speed limits, road engineering (e.g. road humps), daytime running lights and speed cameras.
- Several studies evaluating speed cameras have concluded that they contribute to reducing the incidence of RTCs, fatalities and injuries although the evidence so far is relatively weak. Australia, New Zealand and UK began installing speed cameras in early 1990's.
- A new speed camera (called safety cameras) initiative was rolled out in UK in 2000 and Strathclyde region was one of the pilot camera schemes. This scheme was aimed at reducing fatal and serious injuries and includes several criteria for installing safety cameras based on the history of serious/fatal injuries and speed on the road. The scheme has shown to be effective with 33% fewer injury collisions and 40% fewer people killed or seriously injured at the camera sites.

CHAPTER 2. AIM AND OBJECTIVES OF THE STUDY

CONTENTS OF CHAPTER 2

2.1	CONTEXT OF THE STUDY	
2.2	RATIONALE	
2.3	AIM AND OBJECTIVES	
2.4	RESEARCH QUESTIONS	
2.5	NULL HYPOTHESES	
2.6	SUMMARY OF CHAPTER 2	

This study was a result of a collaboration between Strathclyde Police, Glasgow University and NHS Ayrshire & Arran. The study included retrospective and prospective analysis of cost and health implications following RTCs that occurred prior to, and after, the establishment of safety cameras. This involved determining economic, individual and social implications of RTCs that occurred at selected camera sites.

2.2 <u>Rationale</u>

Road casualties remain a major public health challenge in all countries. The implementation of evidence based countermeasures is hindered by a lack of high quality research and, in particular, the near impossibility of conducting randomised controlled trials in a highly pollicised sector of public policy. To investigate the epidemiology of RTCs in Strathclyde, and to try to determine the effectiveness of safety cameras in reducing injury incidence and severity, this study exploited the availability of routinely collected road casualty data in both the police force and the NHS in Scotland's most populous region.

2.3 Aim and objectives

The overall aim of the study was:

To investigate the epidemiology, cost and prevention of RTC injuries in Strathclyde.

The specific objectives of the study were:

 To establish the overall epidemiology and accuracy of reporting of RTC injuries in the Strathclyde region of Scotland.

- 2. To determine the epidemiology of RTC injuries and the effectiveness of safety cameras at the camera sites in Strathclyde with special reference to different road users, RTC types and severity, before and after camera installation.
- To estimate the economic burden of hospital admissions due to RTC injuries in Strathclyde and at the camera sites before and after installation.

2.4 Research questions

- 1. What is the epidemiology of RTC casualties in Strathclyde region and how reliable are the routine data?
- 2. What are the epidemiological characteristics of RTCs that are prevented through safety camera enforcement?
- 3. What hospital admission costs are incurred by different types of injuries and casualties in Strathclyde and at safety camera sites?

2.5 Null hypotheses

- 1. Police RTC records include all hospitalised road traffic casualties.
- 2. All types of RTCs are equally preventable through safety camera enforcement.
- 3. The economic burden of RTC hospital admissions is equal for all injury and casualty characteristics in Strathclyde and before and after camera installation on safety camera sites.

2.6 Summary of chapter 2

- This study was a result of a collaboration between Strathclyde Police, Glasgow University and NHS Ayrshire & Arran.
- Aim: To investigate the epidemiology, cost and prevention of RTC injuries in Strathclyde.
- The study direction was threefold:
 - 1. Epidemiology and accuracy of reporting of RTC injuries.
 - 2. Epidemiology of RTC injuries and the effectiveness of safety cameras at the camera sites
 - 3. Economic burden of hospital admissions in Strathclyde and at safety camera sites.

CHAPTER 3. METHODS

Contents of Chapter 3

3.1 (OVERVIEW OF METHODOLOGICAL APPROACH	61
3.1.1	Epidemiological methods	61
3.1.1	1.1 Descriptive epidemiology	62
3.1.1	1.2 Analytical epidemiology	
3.1.1	1.3 Interventional epidemiology	
3.1.2		
3.1.3	Study setting and period	
3.2	DATA SOURCES	
3.2.1	Police road casualty records: STATS19	
3.2.2	NHS hospitalisation records: Scottish morbidity records 01	
3.2.3	Scottish index of multiple deprivation	
3.2.4	Strathclyde Safety Camera Partnership	
3.2.4	4.1 Camera site selection methods	
3.2.4	4.2 Sites and crashes selected	
3.3 1	LINKAGE OF POLICE AND HOSPITAL ROAD TRAFFIC CASUALTY RECORDS	
3.3.1	now trils initiage differs from others	
3.3.2	Parameters for linkage	
3.3.4	2.1 Police toad casually records: STATS19	
333	Data linkage methods and results	
3.3.3	3.1 Matching standards	
3.3.3	3.2 Linkade results	
3.3.3	3.3 Estimated accuracy of linkage	74
3.4	CHARACTERISTICS OF THE LINKED POLICE-HOSPITAL DATA BASE	75
3.4.1	Police road casualty data: STATS19	
3.4.1	1.1 Casualty demographics	75
3.4.1	1.2 Casualty severity	75
3.4.1	1.3 Crash characteristics	76
3.4.2	NHS hospital discharge data: SMR01	
3.4.2	2.1 Injury definitions	
3.4.2	2.2 Length of hospital stay	
3.4.3	Unlinked SMR01 and STATS19	84
~		
3.4.4	Statistical power	
3.4.4 3.5	Statistical power	
3.4.4 3.5 3.5.1	Statistical power ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording	
3.4.4 3.5 3.5.1 3.5.2	Statistical power ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording 1.1 Records used	
3.4.4 3.5 3.5.1 3.5.1 3.5.2	Statistical power ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording 1.1 Records used 1.2 Study period	
3.4.4 3.5 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1	Statistical power ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording 1.1 Records used 1.2 Study period	85
3.4.4 3.5 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1	Statistical power ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording	85
3.4.4 3.5 3.5.1 3.5.2 3.5.2 3.5.2 3.5.2 3.5.2 3.5.2 3.5.2	Statistical power ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording	85
3.4.4 3.5 3.5.1 3.5.2 3.5.2 3.5.2 3.5.2 3.5.2 3.5.2 3.5.2 3.5.2	Statistical power ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording	85
3.4.4 3.5 3.5.1 3.5.2 3.5.2 3.5.2 3.5.2 3.5.2	Statistical power ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording	85
3.4.4 3.5 3.5.1 3.5.2 3.5.2 3.5.2 3.5.2 3.5.2 3.5.2 3.5.2	Statistical power ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording	85
3.4.4 3.5 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.2 3.5.2 3.5.2 3.5.2	Statistical power ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording	85
3.4.4 3.5 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.2 3.5.2 3.5.2 3.5.2 3.5.2 3.5.2 3.5.2 3.5.2	Statistical power ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording	85
3.4.4 3.5 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.2 3.	Statistical power ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording	85 85 85 85 86 86 86 87 87 87 88 88 88 88 88 89 89 89 89 90 90
3.4.4 3.5 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.2 3.	Statistical power ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording	85 85 85 86 86 86 87 87 87 88 88 88 88 89 89 89 90 90 91
3.4.4 3.5 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.2 3.5.	Statistical power ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording	85 85 85 86 86 86 87 87 87 88 88 88 88 89 89 89 90 90 91 92
3.4.4 3.5 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.2 3.5.3 3.5.2 3.5.	Statistical power ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording	85 85 85 86 86 86 87 87 87 88 88 88 89 89 89 89 90 90 90 91 92 92
3.4.4 3.5 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.2 3.5.	Statistical power ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording	85 85 85 86 86 86 87 87 87 88 88 88 89 89 89 90 90 90 90 91 92 92 92
3.4.4 3.5 3.5.1 3.5.1 3.5.1 3.5.1 3.5.1 3.5.2 3.5.	Statistical power ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording	85 85 85 86 86 86 87 87 87 88 88 88 89 89 89 90 90 90 90 91 92 92 92 93
3.4.4 3.5 3.5.1 3.5.1 3.5.1 3.5.2 3.5.	Statistical power. ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording. 1.1 Records used. 1.2 Study period 1.3 Severity categories in STATS19 and SMR01 1.4 Linked and unlinked categories. 1.5 Assessing time trends. 1.6 Tests of association with epidemiological characteristics 1.7 Length of hospital stay analysis Epidemiological impact of safety cameras. 2.1 Records used in safety camera (STATS19) evaluation 2.2 Study period in safety camera (STATS19) evaluation 2.3 Safety cameras included in study. 2.4 Assessing time trends in safety camera (STATS19) evaluation 2.5 Test of epidemiological characteristics in safety camera (STATS19) evaluation 2.6 Pre and post camera installation Cost of road traffic crashes in Strathclyde region 3.1 Records used in RTC cost in Strathclyde region 3.2 Study period for RTC cost in Strathclyde region 3.3 Follow-up admissions in RTC cost in Strathclyde region 3.4 Weighted mean costs	85 85 85 86 86 86 87 87 87 88 88 89 89 89 90 90 90 90 90 90 91 92 92 92 93 93
3.4.4 3.5 3.5.1 3.5.1 3.5.1 3.5.2 3.5.	Statistical power. ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording. 1.1 Records used. 1.2 Study period. 1.3 Severity categories in STATS19 and SMR01 1.4 Linked and unlinked categories. 1.5 Assessing time trends. 1.6 Tests of association with epidemiological characteristics 1.7 Length of hospital stay analysis Epidemiological impact of safety cameras. 2.1 Records used in safety camera (STATS19) evaluation 2.2 Study period in safety camera (STATS19) evaluation 2.3 Safety cameras included in study. 2.4 Assessing time trends in safety camera (STATS19) evaluation 2.5 Test of epidemiological characteristics in safety camera (STATS19) evaluation 2.5 Test of epidemiological characteristics in safety camera (STATS19) evaluation 2.6 Pre and post camera installation Cost of road traffic crashes in Strathclyde region 3.3 Follow-up admissions in RTC cost in Strathclyde region 3.3 Follow-up admissions in RTC cost in Strathclyde region 3.4 Weighted mean costs 3.5 Costs by epidemiological variables.	85 85 85 86 86 86 87 87 87 88 88 89 89 89 89 90 90 90 90 90 90 90 90 90 9
3.4.4 3.5 3.5.1 3.5.1 3.5.1 3.5.2 3.5.	Statistical power. ANALYSES. Epidemiology of road traffic crashes, injuries and accuracy of police recording. 11 Records used. 12 Study period. 13 Severity categories in STATS19 and SMR01 14 Linked and unlinked categories 15 Assessing time trends. 16 Tests of association with epidemiological characteristics 17 Length of hospital stay analysis Epidemiological impact of safety cameras. 21 Records used in safety camera (STATS19) evaluation 22 Study period in safety camera (STATS19) evaluation 23 Safety cameras included in study. 24 Assessing time trends in safety camera (STATS19) evaluation 25 Test of epidemiological characteristics in safety camera (STATS19) evaluation 26 Pre and post camera installation Cost of road traffic crashes in Strathclyde region 33 Follow-up admissions in RTC cost in Strathclyde region 33 Follow-up admissions in RTC cost in Strathclyde region 34 Weighted mean costs 35 Costs by epidemiological variables 36 Cost of fatalities	85 85 85 86 86 86 87 87 87 88 88 89 89 89 89 90 90 90 90 90 90 90 90 90 9
3.4.4 3.5 3.5.1 3.5.1 3.5.1 3.5.2 3.5.	Statistical power. ANALYSES. Epidemiology of road traffic crashes, injuries and accuracy of police recording. 11 Records used. 12 Study period. 13 Severity categories in STATS19 and SMR01 14 Linked and unlinked categories 15 Assessing time trends. 16 Tests of association with epidemiological characteristics 17 Length of hospital stay analysis Epidemiological impact of safety cameras. 21 Records used in safety camera (STATS19) evaluation 22 Study period in safety camera (STATS19) evaluation 23 Safety cameras included in study. 24 Assessing time trends in safety camera (STATS19) evaluation 25 Test of epidemiological characteristics in safety camera (STATS19) evaluation 26 Pre and post camera installation Cost of road traffic crashes in Strathclyde region 33 Follow-up admissions in RTC cost in Strathclyde region 33 Follow-up admissions in RTC cost in Strathclyde region 34 Weighted mean costs 35 Costs by epidemiological variables 36 Cost of fatalities 37 Distribution of cost	85 85 85 86 86 86 87 87 87 88 88 89 89 89 89 90 90 90 90 90 90 90 90 90 9
3.4.4 3.5 3.5.1 3.5.1 3.5.2 3.5.	Statistical power. ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording. 1.1 Records used 1.2 Study period 1.3 Severity categories in STATS19 and SMR01 1.4 Linked and unlinked categories 1.5 Assessing time trends. 1.6 Tests of association with epidemiological characteristics 1.7 Length of hospital stay analysis Epidemiological impact of safety cameras. 2.1 Records used in safety camera (STATS19) evaluation 2.2 Study period in safety camera (STATS19) evaluation 2.3 Safety cameras included in study. 2.4 Assessing time trends in safety camera (STATS19) evaluation. 2.5 Test of epidemiological characteristics in safety camera (STATS19) evaluation 2.6 Pre and post camera installation Cost of road traffic crashes in Strathclyde region 3.1 Records used in RTC cost in Strathclyde region 3.2 Study period for RTC cost in Strathclyde region 3.3 Follow-up admissions in RTC cost in Strathclyde region 3.4 Weighted mean costs 3.5 Costs by epidemiological variables 3.6 Cost of fatalities 3.7 Distribution of cost Cost of road traffic crashes at safety camera sites 3.4 Records used in RTC cost in	85 85 85 86 86 86 87 87 87 88 89 89 89 89 90 93 93 94 95 95 95 97 97
3.4.4 3.5 3.5.1 3.5.1 3.5.2 3.5.	Statistical power. ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording	85 85 85 86 86 86 87 87 87 88 89 89 89 89 90 93 93 94 95 95 97 97 97 97 97
3.4.4 3.5 3.5.1 3.5.1 3.5.2 3.5.	Statistical power. ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording	85 85 85 86 86 86 87 87 87 88 89 89 89 89 90 97 97 97 98
3.4.4 3.5 3.5.1 3.5.1 3.5.2 3.5.	Statistical power. ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording. 1.1 Records used 1.2 Study period 1.3 Severity categories in STATS19 and SMR01 1.4 Linked and unlinked categories 1.5 Assessing time trends. 1.6 Tests of association with epidemiological characteristics 1.7 Length of hospital stay analysis <i>Epidemiological impact of safety cameras</i> . 2.1 Records used in safety camera (STATS19) evaluation 2.2 Study period in safety camera (STATS19) evaluation 2.3 Safety cameras included in study 2.4 Assessing time trends in safety camera (STATS19) evaluation 2.5 Test of epidemiological characteristics in safety camera (STATS19) evaluation 2.6 Pre and post camera installation <i>Cost of road traffic crashes in Strathclyde region</i> 3.3 Follow-up admissions in RTC cost in Strathclyde region 3.3 Follow-up admissions in RTC cost in Strathclyde region 3.4 Weighted mean costs 3.5 Costs by epidemiological variables. 3.6 Cost of road traffic crashes at safety camera sites 3.7 Distribution of cost. <i>Cost of road traffic crashes at safety camera sites</i> 3.4 Records used in RTC cost at safety camera sites	85 85 85 86 86 86 87 87 87 88 89 89 89 90 95 97 97 97 98 98 98 98
3.4.4 3.5 3.5.1 3.5.1 3.5.2 3.5.	Statistical power. ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording. 1.1 Records used 1.2 Study period 1.3 Severity categories in STATS19 and SMR01 1.4 Linked and unlinked categories 1.5 Assessing time trends. 1.6 Tests of association with epidemiological characteristics 1.7 Length of hospital stay analysis <i>Epidemiological impact of safety cameras</i> . 2.1 Records used in safety camera (STATS19) evaluation 2.2 Study period in safety camera (STATS19) evaluation 2.3 Safety cameras included in study. 2.4 Assessing time trends in safety camera (STATS19) evaluation 2.5 Test of epidemiological characteristics in safety camera (STATS19) evaluation 2.6 Pre and post camera installation <i>Cost of road traffic crashes in Strathclyde region</i> 3.1 Records used in RTC cost in Strathclyde region 3.2 Study period for RTC cost in Strathclyde region 3.3 Follow-up admissions in RTC cost in Strathclyde region 3.4 Ocst of road traffic crashes at safety camera sites 3.5 Olow-up admissions in RTC cost at safety camera sites 3.6 Cost of fatalities 3.7 Distribution of cost. <i>Cost of road traffic crashes at safety camera sites</i>	85 85 85 86 86 86 87 87 87 88 89 89 90 93 93 95 97 97 97 98 98 98 98
3.4.4 3.5 3.5.1 3.5.1 3.5.2 3.5.	Statistical power ANALYSES Epidemiology of road traffic crashes, injuries and accuracy of police recording. 1.1 Records used 1.2 Study period 1.3 Severity categories in STATS19 and SMR01 1.4 Linked and unlinked categories 1.5 Assessing time trends. 1.6 Tests of association with epidemiological characteristics 1.7 Length of hospital stay analysis Epidemiological impact of safety cameras. 2.1 Records used in safety camera (STATS19) evaluation 2.2 Study period in safety camera (STATS19) evaluation 2.3 Safety cameras included in study. 2.4 Assessing time trends in safety camera (STATS19) evaluation 2.5 Test of epidemiological characteristics in safety camera (STATS19) evaluation 2.6 Pre and post camera installation Cost of road traffic crashes in Strathclyde region 3.1 Records used in RTC cost in Strathclyde region 3.2 Study period for RTC cost in Strathclyde region 3.3 Follow-up admissions in RTC cost in Strathclyde region 3.4 Weighted mean costs 3.5 Cost by epidemiological variables 3.6 Cost of fatalities 3.7 Distribution of cost Cost of road traffic crashes at safety camera sites 4.1 Records used in RTC cost at saf	85 85 85 86 86 86 87 87 88 89 89 89 90 93 93 95 97 97 97 98 98 98 99 99

3.1 Overview of methodological approach

The study comprised a variety of epidemiological approaches that play a major part in modern traffic injury research. Additionally, the epidemiological findings were supplemented by economic information to enhance the utility of the results. This section includes an overview of the methodology used starting with a brief history on how epidemiological methods have been used in public health and more specifically in road traffic injury research. It also includes an account of the three main strands of epidemiological methods: descriptive, analytical and interventional.

The section ends with a description of the economic evaluation utilised in the study and what study setting and period were incorporated.

3.1.1 Epidemiological methods

Epidemiological methods are often used in public health research. One of the more famous early (even founding) applications of epidemiological methods was in London during the outbreak of cholera in 1854, where Dr. John Snow, through plotting the disease on a map, identified a water pump which (by removing the handle) ended the disease. Epidemiology has traditionally been utilised for infectious and chronic disease and it has only more recently been used for road traffic injury research. By searching PubMed (an electronic search engine for medically published research) using the words "traffic", "injuries" and "epidemiology" five-yearly a clear pattern emerges (see Figure 1): in 1965 only 7% of articles on "traffic" "injuries" contained "epidemiology", while in 1985 31% and 2005 39% did.



Figure 1 Number of publications on PubMed with search terms: epidemiology, traffic and injuries by year

Epidemiological methods are commonly divided in to three categories, which all, to some extent, has been utilised in this study: descriptive, analytical and interventional.

3.1.1.1 Descriptive epidemiology

Descriptive epidemiology has been used to illustrate the scale of the problem, in this case road traffic injuries and its consequences. The results are most often displayed in tables without statistical tests and include counts and percentages. In this study descriptive analyses have been used to show the distribution of linked and unlinked records, length of stay and in all cost analysis.

3.1.1.2 Analytical epidemiology

Analytical epidemiology in this study included assessing time trends using linear regression and examining associations using chi square tests and logistic regression models. Regr Regression analysis was used several times in assessing time trends including time trends of linked and unlinked records.

Chi square tests were used when comparing the epidemiological characteristics of RTCs in Strathclyde region with the RTC characteristics at the safety camera sites and logistic regression was used when assessing the impact of various factors in whether or not an RTC was recorded by the police.

3.1.1.3 Interventional epidemiology

The gold standard of interventional epidemiology is the randomised controlled trial, but this was impossible in the context of this study for several reasons. Firstly, most of the safety cameras were already installed when the study commenced and, secondly, the safety camera initiative is a nationwide scheme where there would have been extremely difficult to intervene in the methodology. It was therefore decided to use a before and after intervention analysis, hereafter referred to as Pre and Post safety camera installation (Pre and Post SCI). This involved assessing rates and calculating rate ratios of Pre and Post SCI by various factors to determine the epidemiological effect (if any) of the safety cameras.

3.1.2 Economic evaluation

The epidemiological investigation was extended to an economic evaluation and costs were applied to the descriptive epidemiological analysis of injury, length of stay and characteristics of RTC casualties. This enabled the calculation of an economic estimate of savings to an intervention for future costbenefit analysis. Safety cameras, for example, are relatively costly to install and run though they generate funds through fines.

Additionally, cost estimates may be utilised to help understand where an intervention might have the greatest economic impact. The economic estimates produced in this study represent only a small proportion of the total costs, incurred by RTCs, as only acute hospital costs are included (see

information on other costs in section 1.5). These are, however, substantial and will aid in understanding some of the economical impact of RTC casualties.

3.1.3 Study setting and period

The Strathclyde police region of Scotland is a mixed urban-rural region that is home to more than 2.3 million residents (around two-fifths of the population of Scotland). It includes eight police force domains namely Argyll, Bute and West Dunbartonshire, Ayrshire, Glasgow Central and West, Glasgow North East and East Dunbartonshire, Glasgow South and East Renfrewshire, North Lanarkshire, Renfrewshire and Inverclyde, South Lanarkshire. Scotland's largest city, Glasgow, is situated in Strathclyde.



Figure 2 Map of UK to the left with Scotland in a darker shade. Map of Scotland to the right with Strathclyde in a darker shade.

(These images are licensed under the Creative Commons Attribution ShareAlike 3.0 License and are free to use)

Strathclyde contains concentrations of severe social deprivation, a factor that is correlated with the risk of RTCs (particularly pedestrian casualties) and perhaps also their likelihood of being reported.

The whole recorded RTC population of Strathclyde region was utilised i.e. not just a sample.

The study period included nine years (1997 to 2005) for RTC occurrence and an additional 6 months of hospital admissions. Safety cameras were installed from 2000 to 2004. The study began in autumn 2002 and over the years additional data were made available and consequently the study period was expanded (from 2003 to 2005/6).

3.2 Data sources

The data used for the study came from two main sources namely the police records for PICs (STATS19) and Scottish Morbidity Records (SMR01). These two data sets were linked by Information Statistics Division Scotland (ISD). Additionally, deprivation classification, population estimates, camera site information and cost estimates were utilised. More specifically, the six data sources and their origins utilised in this study were:

- Police records on road traffic crashes: STATS19 (from Strathclyde Police)
- Hospital admissions: Scottish Morbidity Records (SMR01, from ISD)
- Deprivation categories: Scottish Index of Multiple Deprivation (SIMD, from Scottish Executive through ISD)
- Population estimates (from General Registry of Scotland)
- Safety camera site information (from Strathclyde Safety Camera Partnership)
- Cost estimates (from Department of Transport)

All, apart from the cost estimates, were collated and/or retrieved by the author. The cost estimates were obtained and adapted to Scottish inpatient records with the help of a health economist.

STATS19 data, safety camera site information and population estimates were retrieved and updated as time passed (approximately yearly, beginning in 2003). SMR01 data were combined with the STATS19 data in the linkage process by ISD, as were the relevant SIMD data (see section 3.3).

The remainder of this section includes a more detailed account of police road casualty records (STATS19), NHS hospitalisation records (SMR01), Scottish index of multiple deprivation (SIMD) and data supplied by the Strathclyde Safety Camera Partnership.

3.2.1 Police road casualty records: STATS19

The STATS19 recording system is used UK-wide and is used to inform all national statistics on RTCs involving personal injuries. STATS19 was first introduced in 1949 and appeared in its current form in 1979. It is reviewed every 5 years and updated from January 2005.

STATS19 should include all RTCs involving death or personal injury caused by one or more vehicles (if reported to the police within 30 days of occurrence [170]). STATS19 hold information on casualties, RTC circumstances and vehicles.

The RTC circumstances and outcomes are well described in STATS19, which comprises 29 variables on general RTC circumstances, 22 on vehicle (including driver) information and 13 on casualty involvement and outcome. There is one 'general' record for each PIC, while there is one record for each injured person and vehicle involved.

3.2.2 NHS hospitalisation records: Scottish morbidity records 01

SMR data have been collected since 1961 and is one of the worlds first and most complete national health data sets. SMR data are used for both epidemiological monitoring and NHS management purposes. The inpatient record (SMR01, non-obstetric and non-psychiatric) hold one record for every episode of care, including three types of information i.e. patient identification and demographic details, episode management and clinical information[171]. This incorporates the international classification of diseases (ICD, which comprise the external code for identifying RTC victims and injury classification) and length of stay [172].

The study involved all hospitalised traffic related casualty-records from SMR01, for Strathclyde region 1997-2005.

3.2.3 Scottish index of multiple deprivation

Deprivation is virtually synonymous with poverty and comprises a number of dimensions apart from economic. It is a concept that is widely used in public health research.

Road traffic crashes and their sequelae are highly correlated with deprivation and as Strathclyde region is one of the most deprived areas of UK, it is of great importance to include a measure of deprivation in any study of public health in this location.

The Scottish Index for Multiple Deprivation (SIMD) is a relatively new development in Scotland and provides deprivation measure for each electoral ward, which is smaller than for example post code sectors (as used in Carstairs deprivation scores [173]), and it is recommended that SIMD is used for analysis of data back to 1997 [174]. SIMD is based on seven different domains: income, employment, housing, health, education, skills and training and geographic access to services and telecommunications. It provides a rank on data zone level (6,505 in Scotland) for each of the different domains and one overall rank (most deprived = 1, least deprived 6,505) [175]. Using the relative ranks allows for analysing SIMD in deciles or vigintiles i.e. decile 1 includes the 10% most deprived data

zones (vigintile 1 = 5% most deprived) and decile 10 the 10% least deprived data zones (vigintile 20 = 5% least deprived) [176].

SIMD (2004) was applied to the linked and unlinked data through transforming full postcodes of casualties and drivers and thereafter applied by ISD in the linkage process.

Population estimates in Strathclyde region were obtained from the General Register Office for Scotland (GROS) for 1997-2005. Population denominators were used in calculating casualty rates per 100,000 population.

3.2.4 Strathclyde Safety Camera Partnership

The Strathclyde Safety Camera Partnership was the main collaborator in the study and provided help on retrieving police data and identification of RTCs on camera sites. For general information on Strathclyde Safety Camera Partnership see section 1.7.3.6.

3.2.4.1 Camera site selection methods

Strathelyde Police have performed various analyses in order to determine the most appropriate roads on which to place safety cameras. There were two different organisations identifying RTC clusters in Strathelyde region i.e. all council road departments and Atkins Consultancy. The councils identify RTC clusters on a biannual basis for their whole region (a requirement under the 1988 Road Traffic Act), using the criteria of 8 injury RTCs per 500 metres in a three-year period to be considered a cluster. This information is then used to assess sites for possible remedial treatment. The councils method concentrates on counts of RTCs and does not distinguish clusters in terms of severity or junction collisions. In cities and towns these clusters tend to be mainly at junctions. Atkins Consultancy performs cluster analysis specifically for the Safety Camera Partnership. This is done using the regulations as set out by the Scottish National Handbook [177] which includes regulations for fixed camera sites. A summary of this follows:

- Site length: Between 400-1500 metres
- Number of fatal and serious collisions (KSI): At least 4 KSI per km in last three calendar years (not per annum)
- Number of personal injury collisions (PIC): At least 8 PIC per km in last three calendar years
- *Causation factors*: Causation factors indicate that speeding was a contributory factor in some or all of the collisions collision sites that are clearly not speed-related have been de-selected
- *85th percentile speed at (or approach to) collision hot spots*: 85th percentile speed at or above guidelines (10% above speed limit plus 2mph i.e. 35mph in a 30 zone) for free-flowing traffic (excluding any rush-hour periods)
- Percentage over the speed limit: At least 20% of drivers are exceeding the speed limit
- Site conditions are suitable for the type of enforcement proposed: Loading and unloading the camera can take place safely
- *Distribution of collisions*: Collisions are clustered close together around a single stretch of road or junction

3.2.4.2 Sites and crashes selected

The Safety Camera Partnership provided data on 47 safety cameras installed between 2000 and 2004, including map coordinates of the cameras and installation dates. Camera sites were defined by Strathclyde Police as a stretch of road surrounding the camera, radiating 500 metres (usually) in each direction from the camera.

Data on the RTCs occurring at the camera sites before and after installation of the safety cameras (Pre and Post SCI) were retrieved in collaboration with the Partnership, Strathclyde Police Information Resources Department and the author using geographical information system software and police RTC casualty reports. An illustration of a safety camera site is shown below.



Figure 3 Map from ArcView of safety camera sites [Pentagon illustrates the camera, Xs the limits of the site and the dots the RTCs.]

3.3 Linkage of police and hospital road traffic casualty records

As discussed previously in section 1.4, the idea of police-NHS record linkage is not new. Linking police RTC records with hospital records is helpful to achieve a more accurate estimate of RTC frequencies and its impact. In this study, linkage enabled estimates of the accuracy of police recording of RTC casualties, assessing the health impact of safety cameras and producing cost estimates.

This section includes an account of how this linkage differed from others. This is followed by information on the perimeters set for linkage i.e. what was included from police and hospital records in the linkage. The section ends with information on how the linkage was performed by ISD. This takes account of the matching standards utilised in linking the data, the results of the linkage and the estimated accuracy of the linked data base.

3.3.1 How this linkage differs from others

The linked data in this study differ from other studies in one, or several, of the following ways:

- All available follow-up hospital admissions (up to 8 years for 1997) were included
- Fatalities arising from RTCs were included (also fatalities occurring long after the RTC).
- The study period in this study was relatively long (9 years).
- Address postcodes were utilised for most of the linking, which should enhance data accuracy.

This aided in assessing the full impact on the NHS of more serious injuries and any temporal changes.

3.3.2 Parameters for linkage

In order to link police and hospital databases, approval was sought both at Strathclyde Police (informal procedure including justification of linkage etc) and ISD (formal application procedure that went through their Privacy Advisory Committee).

Linkage was performed by ISD Scotland in 2006.

3.3.2.1 Police road casualty records: STATS19

STATS19 records from Strathclyde region, including all PICs occurring in the years 1997 to 2005, were retrieved from Strathclyde Police Information Resources. Postcodes, age, gender and date of RTC were supplied to ISD for linkage. Postcodes were more frequent in the latter years of STATS19 records.
The first study year was determined by the need to have at least three years of data available prior the installation of safety cameras. The last year was determined by when the linkage materialised i.e. as much as possible. The end year altered over time as the linkage procedure became delayed.

3.3.2.2 NHS hospitalisation records: SMR01

The selected admissions for this study were those ICD-9 and ICD-10 (international classification of diseases, 9th and 10th revisions respectively) diagnostic codes for RTC injuries (ICD-9: E810-E819; ICD-10: V01 - V82, V87, V89.2). All non-traffic RTC injuries were excluded (as STATS19 does not record non-traffic RTCs, defined by ICD-10 as "any vehicle accident that occurs entirely in any place other than a public highway" [178]).

The SMR01 data were internally linked with all hospital episodes belonging to an individual over time. This allowed episodes constituting a single continuous inpatient stay (including intra- and interhospital transfers) or re-admissions to be identified.

Death records with mention of RTC were obtained from General Register Office for Scotland (GROS) and attached to the SMR01 data or directly to a STATS19 record (if death was instant and no hospitalisation was required).

For initial admissions, the hospitalisation data were restricted to hospitals located within or bordering Strathclyde region (to allow for RTC casualties admitted to neighbouring hospital care) namely: Greater Glasgow, Argyll & Clyde, Ayrshire & Arran, Forth Valley, Lanarkshire, Lothian, and Dumfries & Galloway board areas. Subsequent admissions included the whole of Scotland, but not the remainder of UK (as these records are not available to ISD). All casualties were included independent of residence, as it was the location of the RTC that was relevant (i.e. all RTCs within the Strathclyde Police Force area were included). STATS19 and SMR01, for Strathclyde region 1997-2005, were linked by ISD Scotland. ISD link databases routinely and were considered the safest option for this procedure.

3.3.3.1 Matching standards

The linkage methods involved matching postcodes, age, gender and date (of RTC and admission) and a default partition (link due to high probability weight utilised by ISD [179,180]). The concept is initially based on a capture-recapture study by Razzak and Luby [181], who used it to estimate TRI (death and injuries) in Pakistan. Anita Morrison, when working in the Paediatric Epidemiology and Community Health (PEACH) unit, has thereafter adapted the method [182] and it was modified slightly further considering the requirements of this study (see standards set in Table 2Table 2. For further information on the (probabilistic) linkage method performed by ISD see appendix V).

3.3.3.2 Linkage results

57% of the linked records utilised postcodes for matching, 36% had postcodes missing but an exact match on age, gender and date, and 7% had a high probability (direct) match (see Table 2). The linkage resulted in over 10,000 police road casualty records that related to approximately 30,000 hospital and death records.

	Frequency	Percent	Valid Percent	Cumulative Percent
A : Exact match on age, sex, date & pcode	4057	37.8	37.8	37.8
B : Exact match on age, sex & pcode - date 1 day before	1084	10.1	10.1	47.8
C : Exact match on sex, date & pcode - age within 2 yrs	438	4.1	4.1	51.9
D : Exact match on sex & pcode - age<=2 yrs, date=1 day out	134	1.2	1.2	53.2
E : Exact match on age, sex & date - pcode near exact	405	3.8	3.8	56.9
F : Exact match on age, sex, & date - pcode missing	3855	35.9	35.9	92.8
G : Default partition - link due to high probability weight	772	7.2	7.2	100.0
Total	10745	100.0	100.0	

Table 2 Matching standards and linkage results -table provided by Information Statistics Division

3.3.3.3 Estimated accuracy of linkage

Historically, the Scottish Medical Record Linkage system has been shown to have 3% false positive and 3% false negative rates [183], but due to the limited identifying information available, this RTC linkage may not have achieved as high accuracy. As names or unique common person identifiers were not available, it was, according to ISD, impossible to estimate the true positive and false negative linkage rates with any degree of accuracy. However, ISD performed rigorous clerical checking of a large sample of best matching pairs at a wide range of probability weights. Depending on whether a postcode was available or not different probability thresholds were set.

A crude estimate of the false negative rate was 2.6% - i.e. 2.6% (1,802) of the unlinked STATS19 records were missed links (i.e. should have been linked). A crude estimate of the false positive rate (see appendix V for an explanation of how this was calculated) was 15.6% - i.e. 15.6% (1,607) of the linked STATS19 records were false positives (i.e. should not have been linked). As a result, the number of missed linked records was nearly the same as the number of incorrectly linked records i.e. resulting in a correct number of records. This does not, however, avoid possible misclassification bias when characteristics of linked data are considered.

3.4 Characteristics of the linked police-hospital data base

STATS19 and SMR01 contribute different information and this section describes which variables from STATS19 and SMR01 were utilised in the linked and unlinked databases. Additionally, a description of an analysis that was required in selecting what follow-up admissions were related to the RTC is included here, together with the results of this analysis. The section ends with information on the statistical power of the study.

3.4.1 Police road casualty data: STATS19

Police RTC records comprise three different areas: casualty information, vehicle information and information about the RTC circumstances. In this study the main information used came from the casualty and context records. For a data guide of STATS19 see appendix ii.

3.4.1.1 Casualty demographics

Casualty demographics included age, gender and postcode. Postcodes were used to retrieve the Scottish Index of Multiple Deprivation (SIMD).

Age was divided in to subcategories of 0-14, 15-24, 25-34, 35-44, 45-54, 55-64 and 65+, which appears to be common and sensible categories often utilised in injury analysis. SIMD is described in detail in section 3.2.3.

3.4.1.2 Casualty severity

Severity of injury is defined by a police officer as being fatal, serious or slight [184]:

The definition of a fatal injury is where the death occurred within 30 days of RTC.

Serious injury includes fracture, internal injury, severe cuts, crushing, burns (excluding friction burns), concussion, severe general shock requiring hospital treatment, detention in hospital as an in-patient, either immediately or later and injuries to casualties who die 30 or more days after the RTC (from injuries sustained in that RTC).

Slight injury includes sprains, whiplash injury (not necessarily requiring medical treatment), bruises, slight cuts, slight shock requiring roadside attention, persons who are merely shaken and who have no other injury should not be included unless they receive or appear to need medical treatment. Although not all fatal or serious injuries recorded by the police will require hospital admissions, all hospital admissions should (by definition) be recorded as either fatal or serious casualties on STATS19.

3.4.1.3 Crash characteristics

Vehicle type in combination with casualty class (driver/rider, passenger and pedestrian) determined what type of road user the casualties were. Road users were grouped as pedestrians, pedal cyclists, moped or motorcyclists, car occupant or occupant of other vehicle. Drivers/passengers were also analysed in some instances.

RTC circumstances included information on whether or not it was a junction related RTC (within 20 metres of a junction) and if the RTC occurred in daylight or darkness. Numbers of casualties and number vehicles involved in the RTC were also retrieved (both grouped in to one, two and three or more). Time of RTC was defined by years (1997-2005) and in analysis of safety cameras this was further divided in to pre or Post SCI.

RTC map coordinates were utilised when retrieving collisions occurring on camera sites.

3.4.2 NHS hospital discharge data: SMR01

The SMR01 variables most important in this study were the injury definitions and length of hospital stay. These were analysed and aggregated to a form to suit the amount of data and analyses performed. For a data guide of SMR01 and GROS see appendix iii and for information on selected ICD codes see appendix iv.

3.4.2.1 Injury definitions

Injuries were identified from the top level of the ICD external causes of injury as: Injuries to the abdomen, lower back or lumbar spine; ankle and foot; elbow and forearm; head; hip and thigh; knee and lower leg; neck; shoulder and upper arm; thorax; wrist and hand. A single injury could therefore conceal several sub-diagnoses within the same heading e.g. head injury. Multiple injuries were excluded in analysis of specific injuries as it would be difficult to determine which injury was the main source of length of stay.

3.4.2.2 Length of hospital stay

Determining length of hospital stay (LoS) was a rather complex procedure (and was accomplished after receiving the linked data set as described in section 3.3.3), as all follow-up admissions available within the full time period were included in the SMR01 data supplied by ISD. The key processes involved in selecting the follow-up admissions (FUA) relevant to the RTC were as follows:

i. All admissions that were deemed transfers were aggregated (<2 days between visits) with the previous admissions and LoS were summed (15% of SMR01 records including FUA records). *Rationale and method*: These were probably not separate hospital episodes, but transfers between wards. These were included with previous admission (excluding the information on diagnosis, but adding on LoS). Death

records were included as separate records whether or not they were adjacent to a hospital admission in time.

- ii. All first admissions were included (first admissions were those SMR or GROS records that were directly linked to the STATS19 records through ISDs linkage process based on date, postcodes etc.) (36% of SMR01 records including FUA records). Rationale and method: These were the admissions that ISD had considered as appropriate to link to the casualty information and no further judgement was made on these.
- iii. All follow-up admissions (FUA) with a different external causal diagnosis were excluded (12% of FUA records i.e. excluding 1st admissions). Rationale and method: Other external causes, such as assault or poisoning, are unlikely to be related to a RTC
- iv. All FUAs with a diagnosis deemed related to the initial injury were included. (28% of FUA records). *Rationale and method*: This was determined by statistical analysis where P values were calculated using chi squared tests; criteria for inclusion for further analysis were that no more than 1 cell could have an expected count of <5 and a significant p value of <0.2. A total of 41 follow-up diagnoses (FUD) were considered significantly related to one or more injuries. The specific injury that was related to a FUD was identified by analysing the proportions of those in relation to all injury categories and the FUD that had more than 30% over the expected proportion of FUD were considered statistically related (e.g. the follow-up diagnosis "cerebrovascular diseases had a p value of 0.09 (i.e. statistically related to one or more of the 1st diagnoses) and casualties with injuries to the thorax, 13% of total casualties, had 21% of this diagnosis i.e. 62% over the expected rate). See table of chi square statistics in appendix VI.
- v. FUAs within 6 months of a previously included FUA (i.e. an admission linked to the first admission in some way) were included (19% of FUA records). Rationale and

method: Since hospital admissions are not everyday occurrences for the large majority of the population, together with the assumption that it is unlikely that two (or more) unrelated admissions would be closer in time than 6 months, the decision was made that admissions close in time (to an RTC related admission) were more likely, than not, to be as a consequence of the RTC. Admissions of patients that had been put on waiting list within 182 days of previous admission were also included, while admissions put on waiting list before RTC were excluded.

vi. The remaining FUAs were excluded (41% of SMR01 records). Rationale and method: This was a large number of FUAs and many of these may have been false negatives (as in missed links).

The initial number of all records was 30,080, but this were before aggregating transfers and excluding 2005 RTCs. Subsequently, there were a total of 23,594 records left of which 13,817 were FUAs. 7,314 of FUAs were excluded (53%). The final number of casualties was 9,777 (of 10,737) and the final number of records (SMR and GROS) was 16,280, of which 6,067 (37%) were FUAs. The table below shows the final distribution of records according to the selection methods utilised and Figure 4 shows a flow-chart of the process step by step.

				Selection											
Excluded			ded					Included							
				C	Other exter	nal factor				Othe	r external	factor			
		N	D	Y	es	No Yes									
				1st re	cord	1st re	ecord		1st re	ecord			1st record		
		N	D	N	lo	Ν	10	Y	es	No Yes					
				Other	Death	Other	Death	Other	Death	Other	Death	Death	Other	Death	Total
	No	Within 6 m of	No	3339	72	589	16			9060	665		50	2	13793
ictor		previous adm	Yes	2070	137	422	16	2293	282			3			5223
Related fa	s	Within 6 m of	No			393	5	1545	22						1965
	Υe	previous adm	Yes			242	13	2229	125			4			2613
Total				5409	209	1646	50	6067	429	9060	665	7	50	2	

Table 3 Distribution of records by selection criteria and steps

A further selection, limiting time of FUA inclusion to 1.5 years after RTC, was used in order to provide equal exposure time for FUAs for all casualties. All FUAs, as included in the above selection were used in some analysis (see further details in section 3.5 including the specific methods used for analyses).

Figure 4 Flow chart of the selection process of follow-up admissions stemming from road traffic crashes between 1997 and 2004 (next page), excluding step i, which was aggregating transfers.



Each casualty was analysed as one record where only the aggregated LoS of all selected FUAs were included. Different variables were added for different time limits i.e. FUAs within 6 months, 1.5 years and the full available period. For example, a casualty may have had 15 days LoS within the first 6 months of RTC, another 4 in the 1.5 years to come (19 days LoS at 1.5 year) and an additionally 2 in the remaining time period (= 21 days for the full period, which depends on when the RTC occurred).

In order to include day surgery in calculation of LoS these were counted as 0.5 days.

"Length of stay" analysis is very complex due to the highly skewed distribution of such data (not symmetrical and with a long tail to the right, see figure). This distribution is very unlike the *normal* distribution (bell-shaped symmetrical) which is required for most statistical tests. Attempts were made to transform the LoS data in this analysis in order to achieve a normal distribution using both the natural logarithm (LN) and square root. The distribution before and after (LN) transformation is shown in Figure 5 and Figure 6 and the conclusion is that the transformation did not work i.e. the Kolmogorov-Smirnov test showed a sign. value of less than 0.05 = not a normal distribution (see Table 4). Tests of means with 95% confidence intervals were therefore not made on this data.



Figure 5 Histogram of aggregated length of hospital stay for the selected admissions limited to 1,5 years (the y-axis should have been 8,000 and x-axis 1,200 in order to display all cases, but this would not have illustrated the skewness so well).



Figure 6 This is LoS (as in figure 5 above) transformed using the natural logarithm

	Kolmogorov-Smirnov(a)						
	Statistic df Sig						
COMPUTE LN_LoS = LN(LoS_NEW)	.176	8822	.000				

Table 4 Tests of Normality of log transformed LoS values

a Lilliefors Significance Correction

3.4.3 Unlinked SMR01 and STATS19

The SMR01 records with ICD diagnostic codes for RTC injuries (ICD-9: E810-E819; ICD-10: V01 - V82, V87, V89.2) that did not link to a STATS19 record were also made available (about 9,000 casualties). The unlinked SMR01 data excluded all non-traffic RTC injuries (as STATS19 does not record non-traffic RTCs, defined by ICD-10 as "any vehicle accident that occurs entirely in any place other than a public highway" [178]). Non-traffic SMR01 records made up about 28% of the total unlinked SMR01 RTC casualties at the start of the process. The unlinked SMR01 records were not internally linked i.e. they did not hold the follow-up admissions. This data set was only utilised in the analysis of accuracy of police recording.

All STATS19 casualty records that did not link to a hospital admission were also made available (70,000 casualties). All casualty data sources by survivors and fatalities are detailed in Table 5 below.

	Unlinked		Linked		Unlinked	
				SMR01		
		STA	TS19			
	Slight	KSI	Slight	KSI		
	casualties	casualties	casualties	casualties		Total
Survivors	63,059	6,391	3,473	6,325	8701	87,949
Fatal in both GROS and STATS19				884		884
Fatal in GROS only			39	7		46
Fatal in STATS19 only		78		9		87
Total	63,059	6,469	3,512	7,225	8,701	88,966
Additional GROS fatal records (fatalities occurring later than 30 days post RTC)						
Fatal records 1 - 6 months			29	44		73
Fatal records 6 m - full time period			48	147		195

 Table 5 Number of casualties in each data set

3.4.4 Statistical power

As highlighted before, this study included the whole recorded RTC population of Strathclyde region i.e. not just a sample. This proved to have sufficient statistical power to detect before-and-after differences at camera sites as well as statistically significant associations of other epidemiological analyses (see result sections 4.1and 4.2).

3.5 <u>Analyses</u>

Analyses in this study were mainly epidemiological and, as stated in the section of aims and objectives, the intention was to work in three parallel and connected directions, namely: general epidemiology (incorporating the important methodological issue of data validity), the effect of safety cameras and the economic impact of RTCs (applied to the general epidemiology and safety camera assessment). This section considers these three strands of research separately, as each part of the study included some specific methods (although many methods overlapped or were the same). Additionally, the cost methods are here divided in two as the general costs in Strathclyde region is based on all casualty costs, while the analysis of safety camera sites calculates the most expensive casualty of an RTC.

3.5.1 Epidemiology of road traffic crashes, injuries and accuracy of police recording

The methods discussed here relates to the first objective:

To establish the overall epidemiology and evaluate accuracy of reporting road traffic casualties in Strathclyde region

And research question:

What is the epidemiology of RTC casualties in Strathclyde region and how reliable are the routine data?

3.5.1.1 Records used

Linking hospital and police RTC records produced three groups of records: linked hospital/police records, unlinked hospital records and unlinked police records. These three data sets were all utilised when assessing time trends and general epidemiology. The analysis tested whether there was a difference in linked/unlinked records and if patterns of recording changed over time.

3.5.1.2 Study period

The study period was 1997-2005 inclusive. As the definitions for the road user categories in the STATS19 data changed slightly in 2005, the time period was restricted to 1997-2004 in the analysis of road users.

3.5.1.3 Severity categories in STATS19 and SMR01

One of the main variables of interest in the analysis was the measurement of severity as utilised by the police to describe the seriousness of the injury suffered by a casualty. In recording an injury the police officer selects a severity grading from three levels: fatal, seriously injured, and slightly injured [184]. Details of these can be found in section 3.4.1.2, but to reiterate: "fatal" means a casualty dying within 30 days of a RTC, "serious injury" means a casualty requiring hospital treatment (but not always admission) and "slight injury" means includes sprains and whiplash injuries.

Fatalities were also determined by using the linked data –in particular the follow-up admissions. In this analysis, this involved using a cut-off point for fatalities at 1.5 years post RTC instead of the police fatality definition (casualty dying within 30 days of RTC). For

information on how follow-up admissions were deemed related to the RTC, see section 3.4.2.2.

The police fatality definition was used when assessing linked and unlinked data, while the longer time period was used in assessing injury severity by RTC determinants (age, gender and SIMD etc).

3.5.1.4 Linked and unlinked categories

For the purpose of this analysis, the data were grouped into three categories: casualties with only STATS19 records, casualties with only SMR01 records, and casualties with both SMR01 and STATS19 records (linked data). Subgroups of linked and unlinked police coded severe and slight casualties were also assessed. Comparing the number of identified road casualties in SMR01, which were also recorded ("linked") in STATS19 records with those that were not ("unlinked") indicates the extent of underreporting of hospitalised road casualties in the police data.

3.5.1.5 Assessing time trends

Casualty reductions over time were assessed by comparing the first three years with the middle and last three years. Three yearly groups were used to iron out any random fluctuations in the data, especially where fatalities were included. Population based three year groups were calculated and the percentage changes between the first and last group are shown. Time trends were also assessed with a trend gradient and p values derived from linear regression (using all years separately). Tables showing all years (counts and percentages) were also constructed.

3.5.1.6 Tests of association with epidemiological characteristics

Pearson's chi-squared statistic was used to test for association of which factors influenced whether or not a RTC hospital admission was recorded by the police (linked and unlinked SMR01). The following variables were tested: age (0-14, 15-24, 25-34, 35-44, 45-54, 55-64 and 65+), gender of casualty, road user (car user, pedal cyclist, pedestrian, moped/motorcycle rider and other vehicle), length of stay in hospital (day case, 1, 2-3, 4-7, and >7 nights stay), third party involvement, Scottish Index of Multiple Deprivation (SIMD), year, month and day of week of crash.

Using binary logistic regression, univariable odds ratios and 95% confidence intervals were calculated for each variable of interest. To assess the interrelationship between the variables, multivariable binary logistic regression analysis was performed, which produced adjusted odds ratios and confidence intervals.

3.5.1.7 Length of hospital stay analysis

As a further validation of the severity reporting in STATS19, an analysis of the type of injury in relation to police severity coding and length of stay (LoS) in hospital (including admissions from the first 6 months after the RTC) was performed to determine whether there were any time trends in LoS in relation to severity codes. Selection methods for follow-up admissions are detailed in section 3.4.2.2. This analysis is, to date, only descriptive and tests may be performed in the future. The length of stay analysis includes LoS means and casualty proportions by (nonfatal) single injury and police severity in three year groups. Injuries were grouped as follows: Injuries to the abdomen, lower back or lumbar spine; ankle and foot; elbow and forearm; head; hip and thigh; knee and lower leg; neck; shoulder and upper arm; thorax; wrist and hand.

3.5.2 Epidemiological impact of safety cameras

The methods discussed here relates to the second objective:

To determine the epidemiology and effectiveness at the safety camera sites with special reference to different road users, RTC types and severity, before and after installation for the years 1997 to 2005;

And research question:

What are the epidemiological characteristics of RTCs that are prevented through safety camera enforcement?

3.5.2.1 Records used in safety camera (STATS19) evaluation

Only STATS19 records were utilised as no SMR01 records were available at this stage (this evaluation was made before the linked data was completed, see further in the section in study weaknesses about difficulties in retrieving the data 5.3.2.2).

3.5.2.2 Study period in safety camera (STATS19) evaluation

The analysis utilises police casualty records (STATS19) for nine years (1997-2005).

The study period was also divided into before and after safety camera installation (Pre SCI and Post SCI), further details in section 3.5.2.6.

3.5.2.3 Safety cameras included in study

In this analysis all 47 speed camera sites installed between 2000 and 2004 (44 of which were fixed and three mobile, as described in 3.2.4) were included in nearly all analysis. The exception was when assessing camera effectiveness over time. In this analysis, the earliest

installed cameras (12 sites), were excluded as these sites did not have as stringent installation criteria as the latter ones. Including cameras with less potential of effectiveness would distort the results over time as they would make up a substantial proportion of the 2nd and 3rd year results.

RTCs at camera sites were identified using geographical information system software. A camera site was defined as a stretch of road (usually) 500 metres before a speed camera to 500 metres after a camera. The RTCs were selected within this range.

The types of RTCs were categorised using the information recorded in STATS19, which included whether or not a RTC was junction related, the number of vehicles involved in the RTC, whether there was pedestrian involvement, the environmental conditions during the crash, casualty severity (defined by the police as fatal, serious or slight [184]), and the number of casualties per RTC.

3.5.2.4 Assessing time trends in safety camera (STATS19) evaluation

Statistical significance was assessed by chi square tests (including the Mantel-Haenszel test for linear association) and whether or not there was overlap of the 95% confidence intervals of the rate ratios. Time trends were also assessed using linear regression.

3.5.2.5 Test of epidemiological characteristics in safety camera (STATS19) evaluation

To determine whether the epidemiological characteristics of RTCs at camera sites and in the rest of Strathclyde were comparable before cameras were installed, chi square tests were performed on some key variables describing the characteristics of the RTCs and crash circumstances. Nine variables were tested comparing the "before" (Pre SCI) characteristics of RTC with the remainder of Strathclyde: whether crashes were junction related, number of casualties per RTC, severity of RTC according to STATS19, number of vehicles involved in RTC, type of weather/road surface/light conditions/day of week when RTC occurred, pedestrian RTC vs. vehicle only RTC.

3.5.2.6 Pre and post camera installation

The data were analysed both in relation to the camera sites themselves and in comparison with the remainder of Strathclyde. Before safety camera installation (Pre SCI) and after (Post SCI) daily RTC rates were calculated. 'Pre SCI days' were all days available for analysis in the nine years before camera installation for each camera, and 'Post SCI days' were the number of available days after camera installation. The number of days Pre SCI and Post SCI varied for every camera, as installation dates differed, but the total number of days was always nine years.

RTC rates Pre and Post SCI were calculated as follows:

Daily rate = N of RTCs / N of days

<u>Rate ratios</u> = "after" RTC rate / "before" RTC rate

Standard Error (SE):

SE log (rate ratio) = $\sqrt{(1/d1+1/d2)}$

where d1 is N RTC exposed and d2 N RTC unexposed

Error factor (EF):

exp {1.96 * SE log (rate ratio)}

95% confidence intervals:

rate ratio / EF to rate ratio * EF

3.5.3 Cost of road traffic crashes in Strathclyde region

The methods discussed here relates to the first part of the third objective:

To estimate the economic burden of RTC hospital admissions in Strathclyde (and at the "safety camera" sites before and after installation)

And research question:

What hospital admission costs are incurred by different types of injuries and casualties in Strathclyde (and at safety camera sites)?

3.5.3.1 Records used in RTC cost in Strathclyde region

Only the linked dataset was included in cost part of the study. This was for two reasons. Firstly, unlinked STATS19 had no hospital admissions available to attach costs to and, secondly, the unlinked SMR01 records had no follow-up admissions available (and could therefore not provide full information on costs).

3.5.3.2 Study period for RTC cost in Strathclyde region

The study period of RTC casualties was 1997 to 2004 inclusive and 1997 to July 2006, for hospital admissions. This permitted the inclusion of a maximum of 1.5 years of hospital admissions for all casualties subsequent to the RTC. Hospital episodes relating to a casualty, with the first admission in the end of 2004, could include all admissions up until July 2006, while a first admission at the start of 2000 could include follow-up admissions until mid 2001. Thus allowing for a relatively extensive follow-up time as well as including most casualties available over eight years.

3.5.3.3 Follow-up admissions in RTC cost in Strathclyde region

Follow-up admissions (FUAs) were selected utilising a method consisting of both the likelihood of a specific follow-up diagnosis occurring, in relation to the initial injury, and how close in time the follow-up admission happened (both in terms of time since RTC and time since last included FUA) (see further section 3.4.2.2).

Two time frames for FUA inclusion were employed:

- The full period of available FUAs. This was only utilised when looking at specific RTC years, as there was more available time for FUAs relating to RTCs that occurred earlier (e.g. 1998) than later (e.g. 2004).
- FUAs 1.5 years following the RTC. This allowed us to utilise all RTCs occurring between 1997 and 2004 as the available FUA records stretched to July 2006 inclusive. This time frame was the most used in the analysis as it allowed for a relatively large time-frame and the inclusion of nearly all available RTCs.

3.5.3.4 Weighted mean costs

Costs were calculated by specialty per day of length of stay. The costs used were weighted mean costs generated from each health board area's mean total gross cost, weighted by the proportion of discharges in each health board during that year. Costs obtained from ISD's Cost Book from the year 2005/06 [185], which includes around 90% of the NHS net operating costs, giving information on the boards that provide hospital and community care directly to patients. This part of the analysis was performed by health economist Kirstin Dickson at NHS Ayrshire & Arran. All weighted mean costs by speciality are detailed in a table in appendix i.

Total of costs were the sum costs of hospital admissions (initial plus follow-up admissions with specific time frames being detailed in tables). Costs were applied for *each hospital day* by speciality and whether it was an inpatient stay or a day case. E.g. if a casualty had related admissions (see section 3.4.2.2 for selection methods of follow-up admissions) of 4 inpatient days and one day as a day case in orthopaedic surgery (£599.67 and £936.29 per day respectively) plus two days in general surgery (£510.11 per day) the total cost would be: $\pounds 599.67*4 + \pounds 936.29*1 + \pounds 510.11*2 = \pounds 4355.21$. This sum of cost from all selected LoS were utilised in analysing mean costs by top level injury and other epidemiological variables (see next section).

Mean costs were calculated including only the casualties that had an admission e.g. total cost of inpatient admissions / number of casualties with an inpatient admission, or total cost of head injury diagnosis / number of casualties with this diagnosis.

Costs were specified in pounds sterling (f).

3.5.3.5 Costs by epidemiological variables

Costs were analysed by the following variables: injury type, road user and vehicle type, Scottish Index of Multiple Deprivation (SIMD) of place of residence, age of casualty and over time.

Injuries were analysed by top level of ICD-diagnosis codes (injuries to the head, knee and lower leg etc.) A single injury could therefore conceal several sub-diagnoses within the same heading e.g. "head injury" includes "injury of cranial nerves" as well as "injury of eye and orbit" etc. Multiple injuries were excluded in the cost analysis as it would be difficult to determine which injury was the main source of the cost. Road users were defined as driver/rider, passenger (vehicle or pillion passenger) and pedestrian and vehicle type included car, motorcycle or moped, pedal cycle, goods vehicle, other vehicle and bus or minibus. Age groups were: 0-14, 15-24, 25-34, 35-44, 45-54, 55-64 and 65+.

3.5.3.6 Cost of fatalities

There were many fatalities in the data that were never admitted to hospital and these were excluded from the mean cost calculations although the fatality count is still shown in tables. This is to show the full impact of the categories analysed. Fatalities (defined here by death occurring within 1.5 years of the RTC) that did have admissions were analysed separately from surviving casualties with hospital admissions, in cost calculations.

Cost of fatalities can be included with a willingness to pay estimate, but this is beyond the scope of this study.

3.5.3.7 Distribution of cost

The full hospital cost for casualties including 1.5 years post RTC were not normally distributed (but had a very skewed distribution **Figure 7**) and attempts were made to transform the data using the the natural logarithm (LN) and square root. The distribution before and after (LN) transformation is shown in **Figure 7** and **Figure 8** and the conclusion is that the transformation were not successful (the Kolmogorov-Smirnov test showed a sign. value of less than 0.05 = significantly different from a normal distribution, see Table 6). Tests of means with 95% confidence intervals were therefore not made on this data.



Figure 7 Hospital admission costs up to 1,5 years post RTC



Figure 8 Log transformed hospital admission costs up to 1,5 years post RTC

Table 6 Tests of Normality of log transformed cost

	Kolmogorov-Smirnov(a) Statistic df Sig.						
LN_cost_1.5	.136	9112	.000				
- Lilliofora Significance Correction							

a Lilliefors Significance Correction

3.5.4 Cost of road traffic crashes at safety camera sites

The methods discussed here relates to the second part of the third objective:

at the "safety camera" sites before and after installation

To estimate the economic burden of RTC hospital admissions (in Strathclyde and)

And research question:

What hospital admission costs are incurred by different types of injuries and casualties (in Strathclyde and) at safety camera sites)?

3.5.4.1 Records used in RTC cost at safety camera sites

Similar to what was discussed in the previous section this part of the study only utilises the linked data (as unlinked STATS19 do not have any hospital admissions and unlinked SMR01 records have no follow-up admissions attached). All safety camera sites available were used (N 47, see section 3.2.4.1 for information on how sites were chosen) and crashes were selected as described in section 3.2.4.2.

3.5.4.2 Study period in RTC cost at safety camera sites

The inclusion period in this analysis was slightly different from the analysis described in section 3.5.3. Admissions and associated costs were included for the first 6 months only post RTC, in order to ensure a standard period at risk for all RTC casualties to be

hospitalised, as it was more important here to include as many RTCs after as possible i.e. RTCs from 2005 were also included as the shorter follow-up period allowed for this.

3.5.4.3 Epidemiological variables included in RTC cost at safety camera sites

Casualties were analysed by the following categories: casualty severity, type of road user and injury type. RTCs were analysed by casualty severity, junction / non-junction related and number of vehicles involved in the crash. Casualty severity was based on whether an injury caused no hospitalisation, hospitalisation of surviving casualties and fatalities. Type of road user included driver/rider, vehicle or pillion passenger and pedestrians as defined by STATS19.

3.5.4.4 Injury definitions in RTC cost at safety camera sites

In analysing costs by injury, ICD-10 diagnoses were used (e.g. injury to the head). If a casualty sustained more than one injury, we selected the injury that was found to be most costly in analysis of all casualties in Strathclyde, see section 3.5.3.5, as the injury label for the casualty. Injuries were grouped further in to the categories "head and neck", "abdomen, lower back, lumbar spine and thorax", "upper limb" and "lower limb", to create large enough groups of casualties for injury analysis.

3.5.4.5 Costs per road traffic crash at safety camera sites

The most expensive casualty was used in analysing severity by RTCs. Junction was defined as any RTC occurring within 20 metres of a junction and non-junction outside 20 metres of a junction (as defined by STATS19). Number of vehicles in RTC were defined by single, two and three or more vehicles –derived from STATS19.

3.5.4.6 Maximum and mean maximum costs

Maximum costs and costs per day were analysed. Maximum cost was the highest hospital cost for a single casualty in an RTC, whether or not the outcome was fatal. The "costs per day" were calculated using the sum of maximum RTC costs (i.e. not actual sum costs) and using the number of days before camera installation (Pre SCI) and after camera installation (Post SCI) as the denominator.

Mean maximum cost and cost per day are useful proxy indicators of injury severity, and also to determine what type of casualties and RTCs cameras have had the optimal costbenefit.

3.6 Summary of Chapter 3

- Several epidemiological methods were used including descriptive epidemiology (evaluating distributions of linked/unlinked data, length of stay and cost analysis), analytical epidemiology (examining associations using chi square tests and logistic regression models) and interventional epidemiology (before and after study).
- The economic evaluation utilised weighted mean costs generated from each health board area's mean total gross cost, weighted by the proportion of discharges in each health board during that year.
- The study setting was Strathclyde region of Scotland and comprised nine years (1997-2005). All RTCs in the region were included and these had sufficient statistical power to detect before-and-after differences at camera sites as well as statistically significant associations of other epidemiological analyses.
- Data from multiple sources were linked including police RTC records (STATS19), NHS hospitalisation records (SMR01) and Scottish index of multiple deprivation

(SIMD), which together with information on camera sites made up the greater part of data used in the study.

- Linkage was performed by ISD Scotland who used matching standards together with probabilistic methods for linkage. The linked database included all follow-up admissions and major work was performed in determining which follow-up admissions were related to the initial admission.
- The linkage resulted in nearly 11,000 police road casualty records that related to approximately 30,000 hospital and death records. Unlinked RTC hospital and police casualties (nearly 9,000 and 70,000 respectively) were also utilised in analysis.
- The variables most used in analysis were from STATS19: casualty demographics, severity of casualty, vehicle and road user type, RTC conditions and RTC map coordinates and from SMR01: injury type, road user (both defined by ICD codes), length of hospital stay and SIMD (attached through postcodes).
- The approach to the analysis was threefold: 1. Epidemiology of road traffic crashes, injuries and accuracy of police recording, 2. Epidemiological impact of safety cameras, 3a. Cost of road traffic crashes in Strathclyde region and 3b. Cost of road traffic crashes at safety camera sites.

CHAPTER 4. RESULTS

Contents of Chapter 4

4.1	EPIDEN	AIOLOGY OF ROAD TRAFFIC CRASHES, INJURIES AND ACCURACY OF POLICE	
RECO	ORDING		102
4	. <u>1.1</u> Tim	e trends in STATS19 and SMR01	
	4.1.1.1	Distribution of STATS19 data over time and severity	103
	4.1.1.2	Distribution of the linked vs. unlinked STATS19 and SMR01 records	104
	4.1.1.3	Time trends of linked and unlinked data	106
4	.1.2 Det	erminants of road traffic crashes and severity	
	4.1.2.1	Age: incidence and severity	109
	4.1.2.2	Gender: incidence and severity	111
	<u>4.1.2.3</u>	Deprivation: incidence and severity	112
	<u>4.1.2.4</u>	Road user: incidence and severity	113
	<u>4.1.2.5</u>	Third party involvement: incidence and severity	116
<u>4</u>	<u>.1.3 Risl</u>	k of hospitalised casualties not being recorded by police: univariable and	
m	nultivariable	<u>ə results</u>	117
	<u>4.1.3.1</u>	Univariable results	117
	<u>4.1.3.2</u>	Multivariable results	119
4	<u>.1.4 Typ</u>	<u>e of injury by road user</u>	
4	.1.5 Rep	porting levels by injury and length of hospital stay	
4.2	EPIDEN	NOLOGICAL IMPACT OF SAFETY CAMERAS	123
4	.2.1 Epi	demiology of road traffic crashes in Strathclyde region vs. at safety camera	
si	ites 124		
	4.2.1.1	Time trends in incidence of road traffic crashes	124
4	.2.2 Epi	demiology of road traffic crashes at safety camera sites	
	4.2.2.1	Impact of cameras on road traffic crash epidemiology	
	4.2.2.2	Camera effect over time	128
4.3	Costi	MPACT OF ROAD TRAFFIC CRASHES	129
-4	.3.1 Cos	t impact of road traffic crashes in Strathclyde region	
	4.3.1.1	Road traffic casualty cost by injury type	
	4.3.1.2	Road traffic casualty cost by road user and vehicle type: Strathclyde region	131
	4.3.1.3	Road traffic casualty cost by road context: Strathclyde region	132
	4.3.1.4	Road traffic casualty cost by Scottish Index of Multiple Deprivation: Strathclyde reg	<u>aion</u> 134
	<u>4.3.1.5</u>	Road traffic casualty cost by age: Strathclyde region	135
	<u>4.3.1.6</u>	Road traffic casualty cost over time: Strathclyde region	136
4	<u>.3.2</u> Cos	t impact of road traffic crashes at safety camera sites	137
	<u>4.3.2.1</u>	Road traffic casualty cost by road user: safety camera sites	138
	<u>4.3.2.2</u>	Road traffic casualty cost by injury type: safety camera sites	139
	<u>4.3.2.3</u>	Cost by road traffic crashes: safety camera sites	139
	4.3.2.4	Cost by type of road traffic crash at junction / non-junction: safety camera sites	140
	4.3.2.5	Cost by type of road traffic crash and number of vehicles involved: safety camera	
	<u>sties</u> 140		
4.4	SUMMA	ARY OF CHAPTER 4	141

4.1 Epidemiology of road traffic crashes, injuries and accuracy of police recording

This section presents the results from analysis of the general epidemiology of RTCs in Strathclyde region including time trends and characteristics of casualties, as well as an assessment of why some hospitalised casualties were recorded by police and others not. It begins with the time trends in STATS19 and SMR01 followed by determinants of RTC and severity, risk of hospitalised casualties not being recorded by police (univariable and multivariable analysis) and injury and length of hospital stay.

The results in this section relate to the first objective:

To establish the overall epidemiology and evaluate accuracy of reporting road traffic casualties in Strathclyde region

And research question:

What is the epidemiology of RTC casualties in Strathclyde region and how reliable are the routine data?

4.1.1 Time trends in STATS19 and SMR01

Over the nine years 1997-2005, nearly 89,000 people were injured on roads in Strathclyde region (counting all sources of data).

Contrasting SMR01 admission rates with STATS19 killed and seriously injured (KSI) rates suggests both that STATS19 had fewer casualties in this category and that the decline over time was steeper (Figure 9) The total decline in STATS19 KSI rates was 38%, compared to the SMR01 reduction in rates of 21% (comparing the first and last three years).



Figure 9 Casualties by SMR01 road traffic crashes admissions and STATS19 killed and seriously injured records in Strathclyde region 1997-2005 per 100,000 population

4.1.1.1 Distribution of STATS19 data over time and severity

Fatal casualties in Strathclyde, as recorded by STATS19, comprised just over 1% of the total in all years (Table 7). The relative proportions of serious and slight casualties changed over time and injuries coded as serious decreased from 18% to 12% of the total while injuries coded as slight increased from 81% to 87%.

There was a significant downward trend over the study period in the overall numbers of road casualties in STATS19 (trend gradient -12.7 and -18%, from 431 per 100,000 pop. in the first three years to 355 in the last three years (Table 8). There were considerable differences in casualty reduction across the three severity categories: fatalities declined by 7% (trend gradient -0.088, a non-significant reduction), serious casualties by 40% (trend gradient -5.089, a significant reduction) and slight injuries by 13% (trend gradient -7.599, a significant reduction).

		Police sever	Police severity grading of casualty				
Year of RT	C	Fatal	Serious	Slight			
1997	Count	117	1,786	8,173	10,076		
	% of row	1.2%	17.7%	81.1%	100%		
1998	Count	116	1,858	8,146	10,120		
	% of row	1.1%	18.4%	80.5%	100%		
1999	Count	109	1,734	7,607	9,450		
	% of row	1.2%	18.3%	80.5%	100%		
2000	Count	106	1,475	7,512	9,093		
	% of row	1.2%	16.2%	82.6%	100%		
2001	Count	117	1,342	7,391	8,850		
	% of row	1.3%	15.2%	83.5%	100%		
2002	Count	92	1,324	7,048	8,464		
	% of row	1.1%	15.6%	83.3%	100%		
2003	Count	116	1,271	7,046	8,433		
	% of row	1.4%	15.1%	83.6%	100%		
2004	Count	107	1,036	6,969	8,112		
	% of row	1.3%	12.8%	85.9%	100%		
2005	Count	91	897	6,679	7,667		
	% of row	1.2%	11.7%	87.1%	100%		
Total	Count	971	12,723	66,571	80,265		
	% of row	1.2%	15.9%	82.9%	100%		

Table 7 Casualty severity STATS19 data, Strathclyde, 1997-2005

Table 8 Road traffic crash casualty rates in Strathclyde per 100,000 population and trend gradient

Three year averages				% Change fro	Trend	n	
Police severity grading	1997-99	2000-02	2003-05	2000-02	2003-05	gradient	value
Fatal	5.0	4.6	4.6	-7%	-7%	-0.088	0.127
Serious	78.2	60.8	47.0	-22%	-40%	-5.089	0.000
Slight	348.0	322.1	303.9	-7%	-13%	-7.599	0.000
Total	431.2	387.5	355.5	-10%	-18%	-12.777	0.000

All years were utilised in estimating the trend gradient, which was calculated using linear regression

4.1.1.2 Distribution of the linked vs. unlinked STATS19 and SMR01 records

The proportion of linked STATS19 casualties out of total STATS19 casualties were assessed and this showed that there were little difference in the distribution i.e. in all periods between 12.5 and 14.7% of total STATS 19 records were linked (Table 9).

Year o	f RTC	Unlinked STATS19	Linked SATS19 and SMR01	Total
1997	Count	8,745	1,331	10,076
	% of row	86.8%	13.2%	100%
1998	Count	8,836	1,284	10,120
	% of row	87.3%	12.7%	100%
1999	Count	8,214	1,236	9,450
	% of row	86.9%	13.1%	100%
2000	Count	7,862	1,231	9,093
	% of row	86.5%	13.5%	100%
2001	Count	7,551	1,299	8,850
	% of row	85.3%	14.7%	100%
2002	Count	7,257	1,207	8,464
	% of row	85.7%	14.3%	100%
2003	Count	7,322	1,111	8,433
	% of row	86.8%	13.2%	100%
2004	Count	7,033	1,079	8,112
	% of row	86.7%	13.3%	100%
2005	Count	6,708	959	7,667
	% of row	87.5%	12.5%	100%
Total	Count	69,528	10,737	80,265
	% of row	86.6%	13.4%	100%

Table 9 Distribution of linked and unlinked STATS 19 casualties over time (1997-2005) in Strathclyde

More than half of all casualties hospitalised following an RTC were identified in STATS19 records. The proportions over time did not change greatly (varying between 50 and 60%, Table 10). Of the SMR01 casualty records, 55% could be linked over the whole study period, suggesting an underreporting rate for hospitalised RTC casualties of 45%.

Year of RTC		Unlinked SMR01	Linked SATS19 and SMR01	Total
1997	Count	1159	1331	2490
	% of row	46.5%	53.5%	100
1998	Count	1262	1284	2546
	% of row	49.6%	50.4%	100
1999	Count	1091	1236	2327
	% of row	46.9%	53.1%	100
2000	Count	831	1231	2062
	% of row	40.3%	59.7%	100
2001	Count	838	1299	2137
	% of row	39.2%	60.8%	100
2002	Count	907	1207	2114
	% of row	42.9%	57.1%	100
2003	Count	940	1111	2051
	% of row	45.8%	54.2%	100
2004	Count	818	1079	1897
	% of row	43.1%	56.9%	100
2005	Count	855	959	1814
	% of row	47.1%	52.9%	100
Total	Count	8701	10737	19438
	% of row	44.8%	55.2%	100

Table 10 SMR01 data on road casualties unlinked and linked (to police data, STATS19), Strathclyde 1997-2005

The highest proportion of STATS19 linked records was in 2001 (14.7% of all recorded casualties) and in SMR01 the highest proportion of linked records was in 2000 (60.8%). The lowest linked proportion was in 2005 for STATS19 and 1998 for SMR01 (12.5% and 50.4% respectively).

4.1.1.3 Time trends of linked and unlinked data

A significant decline was observed in RTCs in all linked and unlinked SMR01 and STATS19 categories (Figure 10 and Table 11). The decline in unlinked SMR01 data, however, appears to have levelled out over the final six years of the study period. There also appears to be a higher proportion of SMR01 casualties with a STATS19 record during the middle time period (2000- 2002).



Figure 10 Numbers of SMR01 road traffic crash records over time by total, linked to STATS19 and unlinked records

	Three yea	r averages		% Change fr	Trend	n	
Data sources	1997-99	2000-02	2003-05	2000-02	2003-05	gradient	value
Unlinked STATS19	375.2	332.7	309.3	-11%	-18%	-11.12	0.000
Linked STATS19 and SMR01	56.0	54.8	46.2	-2%	-17%	-1.656	0.002
Unlinked SMR01	51.1	37.8	38.4	-26%	-25%	-1.957	0.015
Total STATS19	431.2	387.5	355.5	-10%	-18%	-12.777	0.000
Total SMR01	107.1	92.6	84.6	-13%	-21%	-3.614	0.000
Total ALL	482.3	425.3	393.9	-12%	-18%	-14.735	0.000

Table 11 Road traffic crash casualty rates in Strathclyde per 100,000 population by linked and unlinked data

Trend gradient calculated using linear regression (all years were utilised in estimating the trend gradient)

There was a relatively high proportion of the unlinked SMR01 casualties in the first threeyear period (3,512 out of 8,701 i.e. 40 %). This could reflect the general time trend in RTC casualty reduction, rather than a linkage problem in earlier years (as a similar proportion of SMR01 were linked to STATS19 over time, see Table 12).

In summary, about one third (33%) of linked SMR01 records and 18% of all SMR01 records were recorded as slight by the police and a little over half of KSI STATS19 records linked to an SMR01 record.

		Unlinked ST/	AIS19	Linked STATS19 and SMR01			
		Slight	KSI	Slight		SMR	
Year		casualties	casualties	casualties	KSI casualties	unlinked	Total
1997	Count	7,708	1,037	465	866	1,159	11,235
	% of row	68.6	9.2	4.1	7.7	10.3	100
1998	Count	7,732	1,104	414	870	1,262	11,382
	% of row	67.9	9.7	3.6	7.6	11.1	100
1999	Count	7,322	892	285	951	1,091	10,541
	% of row	69.5	8.5	2.7	9.0	10.4	100
2000	Count	7,203	659	309	922	831	9,924
	% of row	72.6	6.6	3.1	9.3	8.4	100
2001	Count	6,990	561	401	898	838	9,688
	% of row	72.2	5.8	4.1	9.3	8.6	100
2002	Count	6,624	633	424	783	907	9,371
	% of row	70.7	6.8	4.5	8.4	9.7	100
2003	Count	6,677	645	369	742	940	9,373
	% of row	71.2	6.9	3.9	7.9	10.0	100
2004	Count	6,563	470	406	673	818	8,930
	% of row	73.5	5.3	4.5	7.5	9.2	100
2005	Count	6,240	468	439	520	855	8,522
	% of row	73.2	5.5	5.2	6.1	10.0	100
Total	Count	63,059	6,469	3,512	7,225	8,701	88,966
	% of row	70.9	7.3	3.9	8.1	9.8	100

Table 12	Linked and unlinked SMR01 a	and STATS19 by year and police severity gradir	۱g
	Liplinkod STATS10	Linked STATS10 and SMD01	
Of a total 89,000 casualties 22% had an SMR01 record of which 12% linked to a STATS19 record (see Figure 11).



Figure 11 Proportions of linked and unlinked STATS19 and SMR01 records 1997-2005

The only category (of linked and unlinked casualties) that showed an increase in numbers over the study period was the STATS19 slight casualties linked to SMR01 records (+5% comparing first and last three years, Table 13), while the numbers of both linked and unlinked STATS19 KSI (killed or seriously injured - combined fatal and serious casualties) decreased substantially (-27% and -47% respectively). There were increases over the study period in the proportion of records that could be linked to SMR01 of both total KSI (47% in 97-99 and 55% in 03-05) and total slight casualties (5% in 1997-99 and 6% in 2003-05, not shown in table).

Table 13 Road traffic crash casualty rates in Strathclyde per 100,000 population by linked and unlinked data and police severity grading

	Police severity	Three year	Three year averages		% change	e from 97-99	Trend	р
Data sources	grading	1997-99	2000-02	2003-05	2000-02	2003-05	gradient	value
Unlinked STATS19	Slight casualties	331.1	305.5	286.0	-8%	-14%	-7.741	0.000
	KSI casualties	44.1	27.2	23.2	-38%	-47%	-3.379	0.001
Linked STATS19 and SMR01 Unlinked	Slight casualties	16.9	16.6	17.8	-2%	5%	0.142	0.700
	KSI casualties	39.1	38.2	28.4	-2%	-27%	-1.799	0.007
SMR		51.1	37.8	38.4	-26%	-25%	-1.957	0.015

Trend gradient calculated using linear regression (all years were utilised in estimating the trend gradient).

4.1.2 Determinants of road traffic crashes and severity

The determinants of RTC incidence and severity analysed included age, gender, deprivation, type of road user and whether or not a third party was involved in the RTC.

4.1.2.1 Age: incidence and severity

Young people, aged 15-24, made up the largest number of total casualties (20,256 i.e. 23% and 782 per 100,000 population), while older people (55-64 and 65+) had the lowest (7% / 8% and 268 / 228 per 100,000 pop respectively, Table 14).

The older age groups, however, had a high proportion of casualties that were admitted to hospital (24% and 34% of total within the oldest age group respectively) i.e. an older RTC casualty was more likely to sustain injuries that required hospital admission compared to younger casualties. Furthermore, the oldest group had a very high mortality rate - as many as 4% of these casualties had died within 1.5 years of the RTC (compared to 0.6% of the youngest and 1-1.4% in the other groups, see Table 15).

Children (0-14 year olds) were the most likely to be unrecorded by police and made up 24% of total unlinked SMR01 data and 47% of SMR01 child RTC records. At the same time a relatively large proportion of STATS19 recorded child casualties had been admitted to hospital (20%), which was the second highest proportion after the oldest age group (22%).

		Unlinked	Linked STATS19/ SMR01	Unlinked	Total
0-14 years	No. of casualties	9476	2305	2046	13827
-	Per 100 000 pop	256.0	62.3	55.3	373.5
	Row %	69%	17%	15%	100%
	Column %	14%	21%	24%	16%
	% of total STATS19	80%	20%		100%
	% of total SMR01	-	53%	47%	100%
15-24	No. of casualties	16344	2205	1707	20256
years	Per 100 000 non	631 7	85.2	66.0	782.9
	Row %	81%	11%	8%	100%
	Column %	24%	21%	20%	23%
	% of total STATS10	2470	12%	2070	100%
	% of total SMR01	0078	56%	- 1/10/	100%
25-34	No. of casualties	1/507	1565	1202	17/5/
years	Per 100 000 pop	516 /	1505 55 A	1232	617 /
	Pow %	9.10.4	0%		100%
		04 /0	970 159/	1 /0	200/
		21%	10%	13%	20%
	% of total SIAISI9	90%	10%	-	100%
35-44	% OI LOLAI SIVIRUT	-	55%	43%	100%
vears	No. of casualties	12057	1357	1158	14572
,	Per 100,000 pop	383.6	43.2	36.8	463.6
	ROW %	83%	9%	8%	100%
		17%	13%	13%	16%
	% of total STATS19	90%	10%	-	100%
45-54	% of total SMR01	-	54%	46%	100%
vears	No. of casualties	7470	1072	759	9301
,	Per 100,000 pop	276.8	39.7	28.1	344.7
	Row %	80%	12%	8%	100%
	Column %	11%	10%	9%	10%
	% of total STATS19	87%	13%	-	100%
EE CA	% of total SMR01	-	59%	41%	100%
vears	No. of casualties	4669	824	620	6113
youro	Per 100,000 pop	204.5	36.1	27.2	267.7
	Row %	76%	13%	10%	100%
	Column %	7%	8%	7%	7%
	% of total STATS19	85%	15%	-	100%
	% of total SMR01	-	57%	43%	100%
65+ years	No. of casualties	4913	1409	1119	7441
	Per 100,000 pop	150.7	43.2	34.3	228.3
	Row %	66%	19%	15%	100%
	Column %	7%	13%	13%	8%
	% of total STATS19	78%	22%	-	100%
	% of total SMR01	-	56%	44%	100%
Total	No. of casualties	69526	10737	8701	88964*
	Per 100,000 pop	339.1	52.4	42.4	434.0
	Row %	78%	12%	10%	100%
	Column %	100%	100%	100%	100%
	% of total STATS19	87%	13%	-	100%
	% of total SMR01	-	55%	45%	100%
*2 casualties wi	th missing age details				

Table 14 Linked and unlinked STATS19 and SMR01 by age group, data for 1997-2005

Age	Sight inju	iries	Hospital	admissions	Fatal (with	nin 1.5 y)
group	Count	% of row	Count	% of row	Count	% of row
0-14	8761	80.5	2057	18.9	61	0.6
15-24	14732	88.2	1775	10.6	204	1.2
25-34	13419	90.3	1279	8.6	155	1.0
35-44	10763	89.7	1099	9.2	135	1.1
45-54	6624	87.1	877	11.5	107	1.4
55-64	4130	84.7	680	13.9	65	1.3
65+	4389	77.4	1055	18.6	229	4.0
Total	62818	86.5	8822	12.2	956	1.3

Table 15 Injury severity by age excluding unlinked SMR01 and year 2005

4.1.2.2 Gender: incidence and severity

A higher proportion of STATS19 male casualties was admitted to hospital than females (25% vs. 18%, using all available data) and there were more male casualties in total (516 compared to 360 per 100,000 pop, Table 16). Including 1.5 years follow-up time (which excludes unlinked SMR01 and year 2005 records) also showed that more males than females had fatal outcomes of RTC, see Table 17 and Figure 12. The difference was most pronounced in ages 25-34 and no difference could be observed in childhood.

Males also made up a larger proportion of total unlinked SMR01 records (66%) and 45% of total male SMR01 records were unlinked.

Gender		Unlinked STATS19	Linked STATS19/ SMR01	Unlinked SMR01	Total
Male	No. of casualties	37,733	6,994	5,732	50,459
	Per 100,000 pop	385.5	71.5	58.6	515.5
	Row %	75%	14%	11%	100%
	Column %	54%	65%	66%	57%
	% of total STATS19	84%	16%	-	100%
	% of total SMR01	-	55%	45%	100%
Female	No. of casualties	31,795	3,743	2,969	38,507
	Per 100,000 pop	296.8	34.9	27.7	359.5
	Row %	83%	10%	8%	100%
	Column %	46%	35%	34%	43%
	% of total STATS19	89%	11%	-	100%
	% of total SMR01	-	56%	44%	100%
Total	No. of casualties	69,528	10,737	8,701	88,966
	Per 100,000 pop	339.2	52.4	42.4	434.0
	Row %	78%	12%	10%	100%
	Column %	100%	100%	100%	100%
	% of total STATS19	87%	13%	-	100%
	% of total SMR01	-	55%	45%	100%

Table 16 Linked and unlinked STATS19 and SMR01 by gender, data for 1997-2005

Table 17 Injury severity by gender excluding unlinked SMR01 and year 2005							
Age	Sight inju	uries	Hospital	admissions	Fatal (within 1.5 y)		
group	Count	% of row	Count	% of row	Count	% of row	
Male	34,024	84.2	5,690	14.1	680	1.7	
Female	28,796	89.4	3,132	9.7	276	0.9	
Total	62,820	86.5	8,822	12.2	956	1.3	



Figure 12 Percentage of injured that had a fatal outcome within 1.5 years, by age and gender

4.1.2.3 Deprivation: incidence and severity

There were large numbers of casualties from the most deprived areas (38% from SIMD quintile 1 and 22% from SIMD 2, while only 12% of casualties were from SIMD 5, Table 18).

Deprivation appeared to have had no effect on whether a hospitalised casualty was recorded by the police or not (55-56% of SMR01 RTC records also had a STATS19 record, Table 18).

There appeared to be no pattern in fatality risk related to SIMD, Table 19.

SIMD		Unlinked STATS19	Linked STATS19/ SMR01	Unlinked SMR01	Total
1 (most	No. of casualties	15,228	3,943	3,204	22,375
deprived)	Row %	68%	18%	14%	100%
	Column %	37%	38%	39%	38%
	% of total STATS19	79%	21%	-	100%
	% of total SMR01	-	55%	45%	100%
2	No. of casualties	9,104	2,363	1,867	13,334
	Row %	68%	18%	14%	100%
	Column %	22%	23%	22%	22%
	% of total STATS19	79%	21%	-	100%
	% of total SMR01	-	56%	44%	100%
3	No. of casualties	6,498	1,727	1,382	9,607
	Row %	68%	18%	14%	100%
	Column %	16%	17%	17%	16%
	% of total STATS19	79%	21%	-	100%
	% of total SMR01	-	56%	44%	100%
4	No. of casualties	5,135	1,229	965	7,329
	Row %	70%	17%	13%	100%
	Column %	13%	12%	12%	12%
	% of total STATS19	81%	19%	-	100%
	% of total SMR01	-	56%	44%	100%
5 (least	No. of casualties	4,894	1,102	903	6,899
deprived)	Row %	71%	16%	13%	100%
	Column %	12%	11%	11%	12%
	% of total STATS19	82%	18%	-	100%
	% of total SMR01	-	55%	45%	100%
Total	No. of casualties	40,859	10,364	8,321	59,544
	Row %	69%	17%	14%	100%
	Column %	100%	100%	100%	100%
	% of total STATS19	80%	20%	-	100%
	% of total SMR01	-	55%	45%	100%
	Missing data	28,669	373	380	29,422

Table 18 Linked and unlinked STATS19 and SMR01 by SIMD, data for 1997-2005

Table 19 Injury severity by SIMD excluding unlinked SMR01 and year 2005

Sight injuries			Hospital	admissions	ssions Fatal (within 1.5 y)			
SIMD		Count	% of row	Count	% of row	Count	% of row	
	1	13,164	78.6	3,290	19.6	301	1.8	
	2	7,919	78.7	1,921	19.1	228	2.3	
	3	5,547	77.7	1,420	19.9	173	2.4	
	4	4,439	79.9	1,016	18.3	100	1.8	
	5	4,280	81.2	882	16.7	107	2.0	
Total		35,349	78.9	8,529	19.0	909	2.0	

4.1.2.4 Road user: incidence and severity

Both SMR01 and STATS19 data have information available on road user (external cause in ICD-10 and casualty class/vehicle type in STATS19). Contrasting road user categories showed which types of casualties were more likely to be missed by police records. In Table

20, both the STATS19 and the equivalent ICD-10 definitions were included (using the STATS19 definition for unlinked STATS19 and linked STATS19/SMR01 and ICD definition for unlinked SMR01).

Pedal cyclist and motorbike/moped casualties were most likely to be hospitalised (47 and 44% of all in respective category recorded, Table 20). Motorcyclists had the highest risk of having a fatal outcome within 1.5 years of RTC (3.6% of injuries recorded by police) followed by pedestrians (2.3% Table 20).

Nearly two thirds of the total injured casualties were car occupants (62%). 38% of hospitalised car occupants (SMR01) did not have a STATS19 record and 10% of all injured car occupants in STATS19 could be linked to SMR01.

The proportions of unlinked SMR01 were higher than linked in the following categories: pedal cyclists, motorcycles/mopeds, and other vehicle.

The road user type that appeared most likely to be missed by police recording was pedal cyclists: 22% of total unlinked traffic related hospital admissions, although only comprising 6% of total casualties. And while 47% of all recorded injured cyclists were admitted to hospital (2,145 of total 4,531), 82% of these had no STATS19 record. In STATS19, there were 2,769 pedal cycle casualties, of which 14% linked to an SMR01 record.

The unlinked SMR01 records contained a substantial proportion of pedestrians (19%), which was a similar proportion to the total recorded pedestrian casualties (20%). 30% of hospitalised pedestrians had no STATS19 record and of all pedestrians, recorded injured in STATS19, 24% were linked to SMR01.

Vehicle type/r	oad user			Linked		
STATS19	ICD-10		Unlinked	STATS19/	Unlinked	Tatal
Dedestrian	Dedectrion		51A1519	SMRUT	SIVIRUT	Total
reuesinan	recesilian	No. of casualties	10,933	3,515	1,502	15,950
		Row %	69%	22%	9%	100%
		Column %	17%	36%	19%	20%
		% of total STATS19	76%	24%	-	100%
		% of total SMR01	-	70%	30%	100%
Pedal cycle	Pedal cyclist	No. of casualties	2,386	383	1,762	4,531
		Row %	53%	8%	39%	100%
		Column %	4%	4%	22%	6%
		% of total STATS19	86%	14%	-	100%
		% of total SMR01	-	18%	82%	100%
Motorcycle	Motorcycle	No. of casualties	1,882	649	781	3,312
or moped	rider	Row %	57%	20%	24%	100%
		Column %	3%	7%	10%	4%
		% of total STATS19	74%	26%	-	100%
		% of total SMR01	-	45%	55%	100%
Car (incl.	Car	No. of casualties	42,024	4,736	2,958	49,718
taxi)	occupant	Row %	85%	10%	6%	100%
		Column %	67%	48%	38%	62%
		% of total STATS19	90%	10%	-	100%
		% of total SMR01	-	62%	38%	100%
Other	Other	No. of casualties	5,595	495	843	6,933
venicie	venicie	Row %	81%	7%	12%	100%
		Column %	9%	5%	11%	9%
		% of total STATS19	92%	8%	-	100%
		% of total SMR01	-	37%	63%	100%
Total		No. of casualties	62,820	9,778	7,846	80,444
		Row %	78%	12%	10%	100%
		Column %	100%	100%	100%	100%
		% of total STATS19	87%	13%	-	100%
		% of total SMR01	-	55%	45%	100%
	Missing data ((all from year 2005)	6,708	959	855	6,708
* Bus or minil	bus, Goods vel	hicle, Other vehicle				

Table 20 Linked and unlinked STATS19 and SMR01 by road user, data for 1997-2004

** Bus occupant, Occupant of pick-up truck or van/ heavy transport, Other land transport

Table 21 Injury severity by SIMD excluding unlinked SMR01 and year 2005

	Sight injuries		Hospital	admissions	Fatal (within 1.5 y)	
Road user	Count	% of row	Count	% of row	Count	% of row
Pedal cycle	2,386	86.2	356	12.9	27	1.0
Motorbike or moped	1,882	74.4	557	22.0	92	3.6
Pedestrian	10,933	75.7	3,186	22.1	329	2.3
Car	42,024	89.9	4,275	9.1	461	1.0
Other vehicle	5,582	91.9	447	7.4	46	0.8
Totalt	62,807	86.5	8,821	12.2	955	1.3

4.1.2.5 Third party involvement: incidence and severity

Casualties involved in a collision with third party (e.g. car-pedestrian, car-car, motorbikelorry etc) made up the majority (79%) of total casualties. More non-collision casualties than collision with third party casualties were admitted to hospital. (34 and 17% respectively, Table 22). Non-collision casualties also had a higher risk of dying compared to collision casualties (1.8 and 1.2% of injuries were fatal within 1.5 years post RTC respectively, Table 23).

Casualties that were in non-collision or collision with a stationary object appeared to be more likely to be missed by police recording. 62% of these SMR01 records did not have a corresponding record in STATS19, while only 39% were missed in an RTC with third party involvement.

		Unlinked STATS19	Linked STATS19/ SMR01	Unlinked SMR01	Total
Collision with third party	No. of casualties	58,448	8,582	3,377	70,407
	Row %	83%	12%	5%	100%
	Column %	84%	80%	39%	79%
	% of total STATS19	87%	13%	-	100%
	% of total SMR01	-	72%	28%	100%
Non-collision or collision with stationary	No. of casualties	11,080	2,155	3,573	16,808
	Row %	66%	13%	21%	100%
object	Column %	16%	20%	41%	19%
	% of total STATS19	84%	16%	-	100%
	% of total SMR01	-	38%	62%	100%
Total	No. of casualties	69,528	10,737	6,950	87,215
	Row %	80%	12%	8%	100%
	Column %	100%	100%	100%	100%
	% of total STATS19	87%	13%	-	100%
	% of total SMR01	-	61%	39%	100%
	Missing data			1,751	1,751

 Table 22 Linked and unlinked STATS19 and SMR01 by collision/non-collision, data for

 1997-2005

Table 23 Injury severity by collision/non-collision, excluding unlinked SMR01 and year 2005

	Sight injuries		Hospital	admissions	Fatal (within 1.5 y)	
Type of RTC	Count	% of row	Count	% of row	Count	% of row
Collision with third party Non-collision or collision	52,897	87.1	7,112	11.7	741	1.2
with stationary object	9,923	83.8	1,710	14.4	215	1.8
Total	62,820	86.5	8,822	12.2	956	1.3

4.1.3 Risk of hospitalised casualties not being recorded by police: univariable and multivariable results

Univariable and multivariable logistic regression were used in analysing association of characteristics of hospitalised casualties to whether or not they were recorded by police.

4.1.3.1 Univariable results

The following categories of admitted RTC casualties were associated with a significantly increased risk of being unrecorded in STATS19 (see Table 24): no third party involvement, age 0-14, earlier years (in this data set), road user (other than car), length of stay in hospital, day of week and month of crash. The SIMD score and gender of casualty were not significantly associated with a risk of being unrecorded in STATS19.

			Odds	95% Confi Interval for	dence odds ratios
		Sign (p	ratio (not	Lower	Upper
Reference category	Category tested	value)	adjusted	Bound	Bound
Third party involvement	No third party involvement	<0.0001	4.33	4.03	4.64
Age group 0-14	Age group 15-24	0.0002	0.87	0.79	0.95
	Age group 25-34		0.93	0.84	1.02
	Age group 35-44		0.93	0.84	1.04
	Age group 45-54		0.77	0.69	0.87
	Age group 55-64		0.83	0.73	0.94
	Age group 65+		0.87	0.78	0.96
Year 2004	Year 1997	<0.0001	1.15	1.02	1.30
	Year 1998		1.30	1.15	1.46
	Year 1999		1.16	1.03	1.32
	Year 2000		0.89	0.78	1.01
	Year 2001		0.85	0.75	0.96
	Year 2002		0.99	0.87	1.12
	Year 2003		1.12	0.98	1.27
Car occupant	Pedal cyclist	<0.0001	7.37	6.54	8.30
	Moped or motorcyclist		1.93	1.72	2.16
	Pedestrian		0.68	0.63	0.74
	Other vehicle occupant		2.73	2.42	3.08
Length of stay >7 days	Day case	<0.0001	1.36	1.21	1.53
	1 overnight stay		0.94	0.85	1.04
	2-3 overnight stay		0.88	0.79	0.98
	4-7 overnight stay		0.86	0.76	0.97
Male	Female	0.4948	0.98	0.92	1.04
Saturday	Sunday	0.0309	1.17	1.05	1.30
	Monday		1.09	0.98	1.22
	Tuesday		1.10	0.98	1.23
	Wednesday		1.11	1.00	1.24
	Thursday		1.11	0.99	1.24
	Friday		0.99	0.89	1.10
December	January	<0.0001	1.26	1.08	1.48
	February		1.15	0.98	1.35
	March		1.26	1.08	1.47
	April		1.52	1.31	1.77
	Мау		1.59	1.37	1.84
	June		1.52	1.31	1.76
	July		1.74	1.51	2.02
	August		1.76	1.53	2.03
	September		1.50	1.29	1.74
	October		1.26	1.09	1.46
	November		1.05	0.90	1.22
SIMD 5 (most affluent)	SIMD 1 (most deprived)	0.9780	0.98	0.89	1.09
	SIMD 2		0.98	0.87	1.09
	SIMD 3		0.97	0.86	1.10
	SIMD 4		0.96	0.84	1.09

Table 24 Results from univariable logistic regression

4.1.3.2 Multivariable results

All variables were included in multivariable testing using logistic regression (see Table 25). The factors significantly associated with hospitalised casualties not being recorded by the police were: no third party involvement, age (progressively higher OR with older age), year (hospitalised casualties in the earlier years appear to be slightly less likely to be recorded by the police), vehicle occupant (pedal cyclist were over 8 times as likely to be missed by police compared to a car occupant), length of stay (day cases were less likely to be recorded by police) and gender (females were less likely to be recorded by the police).

		ie regressi	Odds ratio	95% confiden for odds ratio	ce intervals s
Reference category	Category tested	Sign. (p value)	(sing. In bold)	Lower Bound	Upper Bound
Third party					
involvement	No third party involvement	<0.0001	3.43	3.13	3.76
Age group 0-14	Age group 15-24	0.0806	1.12	0.99	1.27
	Age group 25-34	0.0002	1.30	1.13	1.49
	Age group 35-44	<0.0001	1.38	1.19	1.59
	Age group 45-54	0.0002	1.34	1.14	1.56
	Age group 55-64	<0.0001	1.51	1.28	1.78
	Age group 65+	<0.0001	1.80	1.56	2.07
Year 2004	Year 1997	0.0022	1.26	1.09	1.46
	Year 1998	<0.0001	1.46	1.26	1.69
	Year 1999	0.0426	1.17	1.01	1.36
	Year 2000	0.0024	0.78	0.67	0.92
	Year 2001	0.0032	0.79	0.68	0.92
	Year 2002	0.8803	0.99	0.85	1.15
	Year 2003	0.5456	1.05	0.90	1.22
Car occupant	Pedal cyclist	<0.0001	8.85	7.63	10.26
	Moped or motorcyclist	<0.0001	2.12	1.83	2.44
	Pedestrian	<0.0001	1.60	1.44	1.78
	Other vehicle occupant	<0.0001	2.67	2.30	3.09
Length of stay	Day case	0.0003	1.30	1.13	1.50
>7 days	1 overnight stay	0.0006	0.81	0.71	0.91
	2-3 overnight stay	0.0063	0.83	0.73	0.95
	4-7 overnight stay	0.0060	0.81	0.70	0.94
Male	Female	<0.0001	1.18	1.09	1.28

Table 25 Results from multivariable logistic regression model

4.1.4 Type of injury by road user

There was large variation in the type of injury sustained by type of road user. All injuries showed significant differences by road user (using logistic regression).

Head injuries were the most common injury of total admitted casualties (44%) and sustained by 57% of pedal cyclists and 54% of pedestrians. Both pedal cyclists and pedestrians were about twice as likely to sustain a head injury compared to a car occupant.

Knee and lower leg injuries were the second largest injury type (22%), incurred by 35% of pedestrians and 26% of drivers of motorbikes and mopeds (equally common as head injuries for the latter group). A motorbike or moped casualty was 2.5 times and pedestrians 3.7 times as likely to suffer knee and lower leg injury compared to car occupants. Motorcyclists and drivers of mopeds were also very likely to get hip and thigh injuries (over twice as likely compared to car occupants).

Thorax injuries were more commonly sustained by car occupants than other road users as were injuries to the abdomen, lower back and lumbar spine (30% and 17% of all car occupants respectively).

Drivers of two wheeled vehicles (motorbikes, mopeds and pedal cycle) were more likely to suffer injuries to the elbow and forearm.

		Sign.	Odds	95% Conf Interval fo ratios	idence or odds	
Iniury diagnosis	Category tested	(p value)	ratio	Lower Bound	Upper Bound	% with
Injuries to the	Pedal cycle	0 0000	0.75	0.55	1 02	13.1%
abdomen, lower	Motorbike or moned	0.0000	0.95	0.00	1.02	16.1%
back, lumbar spine	Pedestrian		0.00	0.63	0.81	12.5%
	Other vehicle		0.89	0.69	1.15	15.2%
	Car (ref cat)		0.00	0.00		16.7%
Injuries to the ankle	Pedal cycle	0.0047	0.55	0.28	1.09	2.3%
and foot	Motorbike or moped		1.43	1.00	2.05	5.9%
	Pedestrian		1.24	1.01	1.53	5.1%
	Other vehicle		0.62	0.35	1.10	2.6%
	Car (ref cat)					4.2%
Injuries to the elbow	Pedal cycle	0.0000	2.14	1.57	2.91	14.1%
and forearm	Motorbike or moped		2.77	2.20	3.49	17.6%
	Pedestrian		0.99	0.84	1.18	7.1%
	Other vehicle		1.47	1.07	2.00	10.1%
	Car (ref cat)					7.1%
Injuries to the head	Pedal cycle	0.0000	2.19	1.77	2.70	56.9%
	Motorbike or moped		0.55	0.46	0.66	25.0%
	Pedestrian		1.95	1.79	2.13	54.1%
	Other vehicle		1.19	0.98	1.43	41.7%
	Car (ref cat)					37.6%
Injuries to the hip	Pedal cycle	0.0000	1.37	0.94	1.99	8.6%
and thigh	Motorbike or moped		2.21	1.71	2.85	13.3%
	Pedestrian		1.57	1.34	1.84	9.8%
	Other vehicle		1.28	0.91	1.80	8.1%
	Car (ref cat)					6.5%
Injuries to the knee	Pedal cycle	0.0000	1.59	1.21	2.09	18.5%
and lower leg	Motorbike or moped		2.52	2.08	3.07	26.5%
	Pedestrian		3.75	3.35	4.18	34.9%
	Other vehicle		1.39	1.08	1.79	16.6%
	Car (ref cat)					12.5%
Injuries to the neck	Pedal cycle	0.0000	0.35	0.22	0.58	4.4%
	Motorbike or moped		0.59	0.44	0.81	7.2%
	Pedestrian		0.11	0.08	0.14	1.4%
	Other vehicle		0.90	0.66	1.21	10.5%
	Car (ref cat)					11.6%
Injuries to the	Pedal cycle	0.0010	1.24	0.90	1.71	12.0%
shoulder and upper	Motorbike or moped		1.56	1.23	1.97	14.6%
ann	Pedestrian		1.19	1.03	1.37	11.6%
	Other vehicle		1.40	1.06	1.85	13.4%
Injuring to the	Car (ref cat)					9.9%
Injuries to the	Pedal cycle	0.0000	0.26	0.19	0.37	9.9%
lioiax	Motorbike or moped		0.50	0.40	0.62	17.3%
	Pedestrian		0.17	0.15	0.20	6.7%
			0.56	0.44	0.70	18.8%
Injurioo to the wrist	Car (rer cat)					29.5%
and hand	Pedal cycle	0.0000	1.33	0.89	1.98	7.6%
	Motorbike or moped		2.18	1.67	2.85	11.9%
	Pedestrian		0.60	0.48	0.74	3.6%
	Other vehicle		0.76	0.48	1.18	4.5%
	Car (ref cat)					5.8%

Table 26 Injury by road user: results from univariable logistic regression. Reference category: car occupant

4.1.5 Reporting levels by injury and length of hospital stay

There was considerable variation on the LoS across injury and severity diagnoses. While increasing proportions of almost all diagnoses were diagnosed as slight rather than serious over the study period, the pattern of LoS was far less consistent (see Table 27).

The numbers of hospitalised abdominal/lower back/lumbar spine injuries were quite similar over the years (only 10% reduction), but the proportion of these injuries coded as slight increased (from 38 to 48%) while the cases coded as severe decreased (from 60% to 49%). However, LoS of serious and slight coded injuries of this type increased over time.

Injuries to the head decreased by 15% overall over time, and while cases coded severe declined by 38%, cases coded slight increased by 19%. Head injuries were increasingly coded as slight (from 35% to 50% of cases with this injury type) and less often coded serious over time (from 55% to 40%). It is possible that head injuries became progressively less serious, but the overall LoS does not support this (LoS was higher in the last year group compared to the 1st; mean LoS rose from 5.1 to 8.1 – not including fatalities).

Hospitalised cases due to neck injuries declined 8% overall, but cases coded as severe declined by 43% while slight injuries increased by 22%. This possible shift in coding practices was reflected in the mean LoS where the mean LoS for slight neck injuries has tripled from 2.7 to 9.2, serious cases, however, also showed an increased mean LoS (6.2 to 11.1).

Thorax injuries decreased by 24% overall although the slight coded injuries and fatalities increased, while cases coded as serious decreased by 45%, although overall LoS remained similar.

Single injury description	Years	Years Total Number		ons of severit	y code	Mean LoS			
	Inumbe		Fatal	Serious	Slight	Serious	Slight	Total	
Injuries to the abdomen, lower	1997-1999	202	2%	60%	38%	6.8	3.4	5.5	
back, lumbal spine	2003-2005	181	3%	49%	48%	14.9	5.1	10.1	
Injuries to the ankle and foot	1997-1999	48	0%	75%	25%	8.1	2.2	6.6	
	2003-2005	42	0%	52%	48%	5.1	6.9	5.9	
Injuries to the elbow and forearm	1997-1999	107	1%	51%	48%	4.3	3.3	3.8	
loicaini	2003-2005	67	0%	52%	48%	3.2	4.3	3.7	
Injuries to the head	1997-1999	824	10%	55%	35%	5.8	3.9	5.1	
	2003-2005	697	10%	40%	50%	15	2.4	8.1	
Injuries to the hip and thigh	1997-1999	89	0%	82%	18%	24.7	20.5	24	
	2003-2005	92	1%	79%	20%	22	13.1	20.3	
Injuries to the knee and lower	1997-1999	361	0.6%	77%	23%	11.1	8.8	10.6	
leg	2003-2005	264	0.4%	75%	25%	10	8.4	9.6	
Injuries to the neck	1997-1999	138	4%	49%	47%	6.2	2.7	4.5	
	2003-2005	127	8%	30%	62%	11.1	9.2	9.8	
Injuries to the shoulder and	1997-1999	87	0%	57%	43%	4.1	3.2	3.7	
	2003-2005	75	0%	49%	51%	7.2	8.3	7.8	
Injuries to the thorax	1997-1999	310	7%	65%	28%	6.1	4.8	5.7	
	2003-2005	236	13%	48%	39%	6.4	5.4	6	
Injuries to the wrist and hand	1997-1999	39	0%	56%	44%	3.2	1.5	2.5	
	2003-2005	32	0%	37%	63%	3.8	1.2	2.1	

Table 27 STATS19 severity proportions and mean length of stay of hospital admissions by (single) injury, comparing 1st and 3rd three year time periods

4.2 Epidemiological impact of safety cameras

This section presents results from the analysis of the effect of safety cameras. In order to assess whether cameras have an effect on RTC incidence and/or severity, the first part of the analysis included a comparison of characteristics of RTCs in Strathclyde region with the characteristics of RTCs at sites before safety camera installation) (Pre SCI). If the characteristics did not differ significantly, comparison of time trends was possible. The second part of analysis assesses differences in daily rates and characteristics of RTCs Pre and Post SCI (after safety camera installation).

The results in this section relate to the second objective:

To determine the epidemiology and effectiveness at the safety camera sites with special reference to different road users, RTC types and severity, before and after installation for the years 1997 to 2005

And research question:

What are the epidemiological characteristics of RTCs that are prevented through safety camera enforcement?

4.2.1 Epidemiology of road traffic crashes in Strathclyde region vs. at safety camera sites

A total of 59,608 RTC records were obtained from STATS19 and of these 1,186 (2%) were RTCs that occurred at camera sites.

The variables that were significantly different (using chi-squared test) were severity of RTC (greater at camera sites before camera installation than in the rest of Strathclyde) and number of vehicles involved in RTC (more RTCs with multiple vehicles involved at camera sites than in the rest of Strathclyde). The remainder of variables tested were not significantly associated with camera sites compared to remainder of Strathclyde (i.e. whether crashes were junction related, number of casualties per RTC, type of weather/road surface/light conditions/day of week when RTC occurred, pedestrian RTC vs. vehicle only RTC).

4.2.1.1 Time trends in incidence of road traffic crashes

When we compared the average RTC rates between the first (1997-99) and last (2003-5) three-year periods, RTC incidence in the whole of Strathclyde region declined by 15% over the study period. The greatest proportional decrease (39%) was in serious RTCs. The trend gradients declined significantly for both slight and serious RTCs, while the decline in fatal RTCs was non-significant (Table 28).

	Three year	averages		% Change	e from 1997-99	Trend gradient	p value
Severity	1997-99	2000-02	2003-05	2000-02	2003-05		
Fatal	4.6	4.3	4.3	-7%	-7%	-0.084	0.086
Serious	67.5	53.0	41.0	-22%	-39%	-4.328	0.000
Slight	244.4	230.5	222.5	-6%	-9%	-3.836	0.000
Total	316.6	287 7	267 7	-9%	-15%	-8 247	0 000

Table 28 STATS19 recorded road traffic crash rates and trend gradient in Strathclyde per 100,000 population

Trend gradient calculated using all years and linear regression

Fatal in all this analysis means police definition of dying within 30 days of RTC



Figure 13 STATS19 recorded road traffic crashes rates in Strathclyde per 100,000 population by year and severity

The linear-by-linear association test confirms that the downward trend at the camera sites was significantly different to the trend in the whole of Strathclyde.

The difference in the distribution of RTCs within each group over time (camera sites and remainder of Strathclyde) is shown in Figure 14. As no common denominator was available, the frequency distribution (% distribution of RTCs at sites vs. remainder of Strathclyde) is shown instead. Contrasting the first three-year average with the last, there was a 32% reduction in RTC incidence at camera sites compared to 16% in the remainder of Strathclyde.



Figure 14 Frequency distribution of road traffic crashes at camera sites and the remainder of Strathclyde by year

4.2.2 Epidemiology of road traffic crashes at safety camera sites

To evaluate the impact of cameras on RTCs at camera sites, daily RTC rates and rate ratios were calculated. 72% of the RTCs at camera sites occurred before camera installation and 63% of days available for analysis were "before" days. In total, RTCs at camera sites declined by 31% (Table 29).

	Number	of				<i>a</i> c	
	RTCs		Daily R	C rates	Rate ratio ('after vs. before Rate 95% CI 95% C		
	Before	After	Before	After	ratio	lower	upper
RTCs at camera sites	854	332	0.0087	0.0060	0.69	0.61	0.78

 Table 29 Rate ratios and 95% confidence intervals of road traffic crash incidence reduction after camera installation

4.2.2.1 Impact of cameras on road traffic crash epidemiology

To investigate whether cameras had a greater effect on some types of RTC than on others, we calculated daily "before" and "after" rates, rate ratios and 95% confidence intervals for the variables listed in Table 30 and Figure 15. All but one of the variables and subcategories

show a significant reduction post-installation, but to varying degrees. Cameras appeared to contribute particularly to the reduction of more severe RTCs. Roads with many non-junction RTCs, RTCs with more than two casualties, single or multiple (more than two) vehicle RTCs and RTCs involving pedestrian injuries also appear to have particularly benefited from cameras, although the numbers were too small to produce significant results.

				Rate Ratio ('after' vs.				
	Variable description	Number	ofRICs	Daily R I	C rates	'before'		
		Refore	Δftor	Before	Δftor	Rate	95% CI	95% CI
Soverity of	KSI (fatal & serious)	212	57	0.0021		0.48	0.36	0.64
RTC	Slight	212	075	0.0021	0.0010	0.40	0.00	0.07
KIO .	Sign	642	275	0.0065	0.0049	0.70	0.00	0.07
Vicinity of	Junction related	463	198	0.0047	0.0036	0.76	0.64	0.89
junction Number of vehicles involved in RTC	Not junction related	391	134	0.0040	0.0024	0.61	0.50	0.74
	One	318	111	0.0032	0.0020	0.62	0.50	0.77
	Two	444	193	0.0045	0.0035	0.77	0.65	0.91
	Three or more	92	28	0.0009	0.0005	0.54	0.35	0.82
Pedestrian vs.	Vehicle only RTCs	623	249	0.0063	0.0045	0.71	0.61	0.82
RTC	Pedestrian RTCs	231	83	0.0023	0.0015	0.64	0.50	0.82
Number of	One	657	253	0.0067	0.0045	0.68	0.59	0.79
casualties per	Тwo	127	59	0.0013	0.0011	0.82	0.62	1.12
RTC	Three or more	70	20	0.0007	0.0004	0.51	0.31	0.83
Light conditions at	Daylight	604	226	0.0061	0.0041	0.66	0.57	0.77
site	Darkness					0.75	0.60	0.94
		250	106	0.0025	0.0019			

Table 30 Rate ratios and 95% confidence intervals of road traffic crash incidence reduction after camera installation by road traffic crash type



Figure 15 Rate ratios and 95% confidence intervals of road traffic crash incidence reduction after camera installation by road traffic crash type

4.2.2.2 Camera effect over time

The data reveal no diminishing effect of cameras on RTC incidence over time (Table 31),

where the first year suggests a reduction of 34%, the second 1% and third 46%.

Table 31 Rate ratios and 95% confidence intervals of "before" and "after" road traffic crashes, grouped by year

	Number of	of RTCs	Daily RT	C rates	Rate ratio ('after' vs. 'before')			
Year after installation					Rate	95% CI	95% CI	
(No. of cameras)	Before	After	Before	After	ratio	lower	upper	
1 st year (35)	753	76	0.0093	0.0060	0.64	0.51	0.81	
2 nd year (29)	559	62	0.0086	0.0059	0.69	0.53	0.90	
3 rd year (14)	294	28	0.0103	0.0056	0.54	0.37	0.79	

4.3 Cost impact of road traffic crashes

This part of the results is divided in to two sections. Firstly, the hospital cost impact of RTC casualties in the whole of Strathclyde region was evaluated. This included costs by injury type, road user, road context, SIMD, age and over time. Secondly, the cost impact of RTCs at camera sites was assessed. This comprised analysis of cost by casualty severity, road user, injury type, junction/non-junction, number of vehicles involved and RTC maximum cost.

The results presented in this section relates to the third objective:

To estimate the economic burden of RTC hospital admissions in Strathclyde and at the "safety camera" sites before and after installation

And research question:

What hospital admission costs are incurred by different types of injuries and casualties in Strathclyde and at safety camera sites?

4.3.1 Cost impact of road traffic crashes in Strathclyde region

The total cost of RTCs occurring in Strathclyde between 1997 and 2004 was nearly $\pounds 58$ million, of which surviving casualties accounted for over $\pounds 53$ million (Table 32). This equates to a total average cost of $\pounds 7.3$ million per year. In this analysis no 2005 data was utilised (or included as missing data.) and these comprised 959 linked records. Most tables also exclude all fatalities (956), but some include fatalities which incurred hospital costs (290).

	Hospital	aumissions					
	Surviving	g casualties		Fatal casualties (within 1.5 y)			
Cost up to 1.5 years post RTC	Count	Mean cost	Total cost	Count	Mean cost	Total cost	
Inpatient admissions	8,003	6,413	51,326,445	237	17,421	4,128,814	
Day surgery admissions	1,954	1,067	2,085,339	109	1,298	141,521	
Total casualties with admissions	8,822	6,054	53,411,784	290	14,725	4,270,335	

Table 32 Mean and total costs (£) of hospital admissions by inpatient and day surgery

4.3.1.1 Road traffic casualty cost by injury type

The data on surviving casualties, with a single injury diagnosis recorded, were analysed further to determine the cost impact of different types of injuries. Casualties with a single diagnosis comprised 60% of all surviving hospitalised casualties (5,337 out of 8,822).

The highest total cost (nearly £7 million) was incurred by head injuries, by far the most frequent single diagnosis in the data (36%, N 1,865). The highest mean costs (£12,241, Table 33) were incurred by injuries to the hip or thigh, at least double the cost of any other single injury. Hip and thigh injuries also, despite only comprising about 5% of single diagnosis casualties, reached a figure of nearly half the total of head injury costs (nearly £3 million).

Injuries to the knee/lower leg had relatively high mean and total costs (£6,218 and just over £5 million respectively). The lowest mean costs were incurred by injuries to other parts of the limbs (wrist/hand, elbow/forearm, shoulder/upper arm and ankle/foot).

		% of single		
Single injury diagnosis	Count	injury	Mean cost	Total cost
Injuries to the head	1,865	36.4	3,645	6,798,251
Injuries to the knee and lower leg	840	16.4	6,218	5,223,390
Injuries to the hip and thigh	240	4.7	12,241	2,937,779
Injuries to the abdomen, lower back, lumbar spine	491	9.6	5,172	2,539,438
Injuries to the thorax	735	14.3	3,163	2,324,791
Injuries to the neck	311	6.1	4,357	1,354,896
Injuries to the shoulder and upper arm	213	4.2	3,010	641,143
Injuries to the elbow and forearm	240	4.7	2,135	512,489
Injuries to the ankle and foot	98	1.9	3,539	346,815
Injuries to the wrist and hand	95	1.9	1,933	183,648
Total	5,128	100	4,458	22,862,640
Missing data (not single injury)	4,650			

Table 33 Mean and total costs (£) of (surviving) hospital admissions by single injury

4.3.1.2 Road traffic casualty cost by road user and vehicle type: Strathclyde region

Nearly a quarter (22.1%) of police recorded injured pedestrians were admitted to hospital and/or died (2.3%, Table 34), while injured drivers or riders only had about half that incidence (10.5% and 1.3% respectively).

The total pedestrian hospital costs were also higher than those of drivers or riders (\pounds 26 million compared to \pounds 21 million) although the number of hospital admissions and fatalities among drivers and riders was about 20% more than that of pedestrians (4,213 admissions/deaths compared to 3,515).

The mean cost of hospitalised pedestrians was substantially higher than for other road users.

1000 030		004								
Road user	Sight inju	uries	Hospital	admissior	าร		.5 y)			
		% of				% of				% of
		row		Mean		row		Mean		row
	Count	count	Count	cost	Total cost	count	Count*	cost**	Total cost	count
Pedestrian	10,933	75.7	3,186	7,481	23,834,213	22.1	329	14,802	2,323,866	2.3
Driver or rider	31,501	88.2	3,760	5,217	19,614,413	10.5	453	13,509	1,161,736	1.3
Passenger	20,386	90.9	1,876	5,311	9,963,158	8.4	174	16,696	784,733	0.8
Total	62,820	86.5	8,822	6,054	53,411,784	12.2	956	14,725	4,270,335	1.3
*all fataliti										

Table 34 Mean and total costs (£) of slight injuries, hospital admissions and fatalities by road user 1997-2004

** mean cost of fatal with hospital admissions

Cars were by far the most common vehicle involved in police recorded hospitalised injury RTCs. The total costs of hospital admissions were £45.5 million (including both surviving and fatal casualties, not shown in table) with cars hitting pedestrians accounting for nearly half of this figure (£22.3 million of which £20.6 million was incurred by surviving casualties).

Pedestrian casualties incurred high mean hospital costs for all types of vehicle RTCs, but especially so when hit by a motorcycle/moped or pedal cycle (although numbers were small, Table 35).

For driver casualties, motorcycles/mopeds incurred the highest mean hospital cost (£6,687), while drivers of buses/minibuses and pedal cyclists had the lowest (£2,400 and £3,811 respectively). Passengers did not figure frequently in any of the casualty vehicle types apart from car RTCs where the mean costs incurred were very similar to drivers (£5,175 and £5,137 respectively).

	Casualty	class									
Vehicle type	Driver or	rider		Vehicle of	or pillion	passenger	Pedestriar	Pedestrian			
		Mean			Mean			Mean			
	Count	cost	Total cost	Count	cost	Total cost	Count	cost	Total cost		
Car	2,637	5,137	13,547,534	1,638	5,175	8,475,992	2,777	7,422	20,611,906		
Motorcycle or											
moped	520	6,687	3,477,204	37	4,307	159,353	38	8,804	334,536		
Pedal cycle	355	3,811	1,353,004	1	7,554	7,554	9	11,369	102,320		
Goods vehicle	178	5,128	912,863	53	4,546	240,945	134	7,644	1,024,305		
Other vehicle	38	6,500	247,002	13	8,557	111,237	40	6,887	275,498		
Bus or minibus	32	2,400	76,807	134	7,224	968,078	188	7,902	1,485,648		
Total	3,760	5,217	19,614,413	1,876	5,311	9,963,158	3,186	7,481	23,834,213		

Table 35 Mean and total costs (\pounds) of (surviving) hospital admissions by road user and vehicle type

4.3.1.3 Road traffic casualty cost by road context: Strathclyde region

Drivers and passengers in non-junction crashes had higher total and mean costs for surviving hospitalised casualties compared to junction RTCs (Table 36). Furthermore, the proportion of casualties that were fatal at non-junction sites was over double those at junction RTCs for these road users. Pedestrians, on the other hand, appeared to incur the same hospital costs and fatality rates, independent of whether the RTC occurred at a junction or not.

		Sight inj	uries	Hospital	admission	S		Fatalities	s (within 1	.5 y)	
			% of		Mean		% of		Mean		% of
	Road user	Count	row	Count	cost	Total cost	row	Count*	cost**	Total cost	row
Not near	Driver or	14 101	010	2 205	E E10	10 149 602	12.0	210	11 700	626 610	1.0
junction	nder	14,121	04.0	2,205	5,510	12,140,003	13.2	319	11,709	030,019	1.9
	Passenger	8,999	87.7	1,147	5,835	6,693,309	11.2	112	13,912	292,150	1.1
	Pedestrian	5,802	74.8	1,778	7,306	12,989,795	22.9	181	15,460	1,252,268	2.3
At or	Driver or rider	17,380	91.1	1,555	4,801	7,465,810	8.2	134	16,410	525,118	0.7
iunction	Passenger	11,387	93.5	729	4,485	3,269,850	6.0	62	18,945	492,583	0.5
,	Pedestrian	5,131	76.7	1,408	7,702	10,844,418	21.1	148	14,100	1,071,598	2.2
	Total	62,820	86.5	8,822	6,054	53,411,784	12.2	956	14,725	4,270,335	1.3

Table 36 Mean and total costs (\pounds) of slight injuries, hospital admissions and fatalities by road user and vicinity of junction

* all fatalities

** mean cost of fatal with hospital admissions

The speed of a vehicle has a large impact on casualty outcome in a RTC i.e. the higher the speed the poorer the outcome. The mean costs for surviving hospitalised drivers and passengers, however, were higher on 30 mph than 40-50 mph roads, while pedestrian casualties followed a clear pattern of higher speed equalling greater damage (mean hospital costs: $\pounds 6,800$, $\pounds 11,000$ and $\pounds 24,000$ for 30, 40-50 and 60-70 mph respectively, Table 37). The proportion of fatalities among pedestrians also reflect this relationship: at 30 mph, 2% of injured pedestrians had a fatal outcome, at 40-50 mph 7% and at 60-70 mph 16%. The large majority of pedestrian fatalities, however, occurred on 30 mph roads, generating over $\pounds 22$ million in total hospital costs.

Table 37 Mean and total costs (£) of slight injuries, hospital admissions and fatalities by
road user and posted speed limit on road

		Sight inju	ries	Hospital	Hospital admissions				Fatalities (within 1.5 y)		
			% of		Mean		% of		Mean		% of
	Road user	Count	row	Count	cost	Total cost	row	Count*	cost**	Total cost	row
30	Driver or										
mph	rider	18,547	91.8	1,556	4,505	7,009,336	7.7	106	17,050	630,850	0.5
	Passenger	12,512	94.1	737	5,568	4,103,425	5.5	47	18,616	372,320	0.4
	Pedestrian	10,531	76.6	2,978	6,844	20,380,804	21.6	247	14,545	2,036,239	1.8
40-50	Driver or										
mph	rider	2,853	90.7	263	4,049	1,064,854	8.4	31	5,319	26,597	1.0
	Passenger	1,737	92.1	133	4,091	544,144	7.1	15	6,554	26,217	0.8
	Pedestrian	185	56.4	119	10,983	1,306,989	36.3	24	12,093	96,741	7.3
60-70	Driver or										
mph	rider	10,101	81.7	1,941	5,946	11,540,223	15.7	316	11,461	504,289	2.6
	Passenger	6,137	84.6	1,006	5,284	5,315,589	13.9	112	16,791	386,196	1.5
	Pedestrian	217	59.6	89	24,117	2,146,420	24.5	58	21,209	190,885	15.9
	Total	62,820	86.5	8,822	6,054	53,411,784	12.2	956	14,725	4,270,335	1.3

*all fatalities

** mean cost of fatal with hospital admissions

4.3.1.4 Road traffic casualty cost by Scottish Index of Multiple Deprivation: Strathclyde region

Driver mean hospital costs were highest for the most affluent population (SIMD 5, Table 38 and Figure 16), but in pedestrians this pattern was reversed (more deprived = higher mean cost). For passengers the mean costs of the most affluent population were only about half of those of the other SIMD groups.

Total costs could not be compared with any certainty as we did not calculate population rates, but pedestrians from the most deprived areas (SIMD 1) by far outnumbered the other pedestrian SIMD groups. The total costs of the SIMD 1 pedestrians were at least double those of any other pedestrian SIMD group (f_1 3.4 million, including both fatal and surviving casualties). Indeed, the total cost of this group was more than the costs of all the other pedestrians combined ($f_{12.6}$ million) and more than all passengers ($f_{10.4}$ million), despite the total number of injured passengers (all SIMD and severity categories) being well over double that of pedestrians in SIMD 1 (12,698 injured passengers and 5,064 injured SIMD 1 pedestrians).

		Sight inj	uries	Hospita	Hospital admissions			Fatal (within 1.5 y)			
			% of				% of				% of
	SIMD	-	row	-	Mean		row	-	Mean		row
	quintile	Count	count	Count	cost	Total cost	count	Count	cost	Total cost	count
Driver or	1	5,631	83.8	1,001	4,899	4,903,589	14.9	87	19,262	443,016	1.3
nder	2	4,133	81.2	847	5,280	4,471,916	16.6	110	12,424	310,605	2.2
	3	3,180	79.7	723	4,881	3,528,994	18.1	89	10,515	147,203	2.2
	4	2,729	81.8	542	5,183	2,809,079	16.2	65	17,121	136,969	1.9
	5	2,716	82.5	506	6,624	3,351,559	15.4	69	8,552	119,727	2.1
Passenger	1	4,263	86.2	622	5,724	3,560,256	12.6	58	20,386	448,483	1.2
•	2	2,535	85.1	409	6,202	2,536,765	13.7	34	12,141	109,271	1.1
	3	1,636	81.6	334	5,595	1,868,831	16.7	34	21,138	126,826	1.7
	4	1,220	83.8	220	5,321	1,170,625	15.1	16	9,489	37,956	1.1
	5	1,128	85.6	171	2,604	445,348	13.0	18	15,191	60,763	1.4
Pedestrian	1	3,244	64.1	1,666	7,350	12,245,757	32.9	154	14,141	1,117,124	3.0
	2	1,244	62.5	663	8,147	5,401,257	33.3	84	18,852	791,798	4.2
	3	722	63.7	361	7,770	2,805,083	31.9	50	11,862	260,953	4.4
	4	479	63.8	254	7,459	1,894,556	33.8	18	9,550	66,851	2.4
	5	424	65.3	205	6,283	1,287,986	31.6	20	12,449	87,140	3.1
Total		35,284	86.5	8,822	6,054	53,411,784	12.2	956	14,725	4,270,335	1.3
Missing		27.536									

Table 38 Mean and total costs (£) of slight injuries, hospital admissions and fatalities by road user and SIMD

Missing



Figure 16 Mean hospital costs (£) by SIMD quintile (1=, most deprived, 5= most affluent) and road user

4.3.1.5 Road traffic casualty cost by age: Strathclyde region

A high proportion of young (0-14 year olds) and old (65+) STATS19 recorded casualties had hospital admissions (near 19% for each group, Table 39), but the youngest age group had the lowest mean cost and the oldest had the highest. As also shown in section 4.1.2.1, severity of RTC casualty was very dependent on age and the proportion of fatal outcomes increase with age from 0.6% in the youngest age group to 4% in age 65+.

The mean hospital costs of surviving casualties rose progressively with advancing age from f_{2} ,4,000 to nearly f_{2} ,11,000.

Since age-specific population rates were not calculated, total costs in the table should be interpreted with caution. The age group 65+, however, had a comparatively high total cost despite the relatively small number of admissions (hence the high mean). In total, the elderly (65+) age group incurred costs of £13.7 million, about double that of any other age category.

	Sight inju	ght injuries Hospital admissions				Fatal (within 1.5 y)				
		% of				% of				% of
Age		row		Mean		row		Mean		row
group	Count	count	Count	cost	Total cost	count	Count	cost	Total cost	count
0-14	8,761	80.5	2,057	4,043	8,316,282	18.9	61	8,103	218,775	0.6
15-24	14,732	88.2	1,775	4,225	7,500,176	10.6	204	3,384	131,957	1.2
25-34	13,419	90.3	1,279	5,983	7,651,804	8.6	155	11,831	283,941	1.0
35-44	10,763	89.7	1,099	6,311	6,935,906	9.2	135	7,014	119,233	1.1
45-54	6,624	87.1	877	6,692	5,868,531	11.5	107	20,637	474,641	1.4
55-64	4,130	84.7	680	8,660	5,888,687	13.9	65	22,907	572,673	1.3
65+	4,389	77.4	1,055	10,664	11,250,398	18.6	229	18,290	2,469,114	4.0
Total	62,818*	86.5	8,822	6,054	53,411,784	12.2	956	14,725	4,270,335	1.3

Table 39 Mean and total costs (£) of slight injuries, hospital admissions and fatalities by age

* 2 casualties with no age

4.3.1.6 Road traffic casualty cost over time: Strathclyde region

Costs over time were analysed using all the available admissions that were regarded as related to the RTC. This was for the purpose of comparing how the total costs were distributed over time.

Mean costs for the first six months of admissions increase slightly over the years from about £5,000 the first couple of years to around £6,000 for the last years and a similar pattern can be found for the 1.5 years admissions included post RTC.

The bulk of costs were associated with admissions the first six months after an RTC although the proportion of this bulk declined progressively as data on more years became available for including follow-up admissions. For RTCs occurring in 1997 only 71% of admissions occurred within six months and a further 5% within 1.5 years following the RTC. These proportions (naturally) increase and for RTCs occurring in 2004 (for which 1-2.5 years worth of admissions were available) 91% occurred the first six months.

Year of R	тс	Hospital admissions up to 6 months post RTC	Hospital admissions up to 1.5 years post RTC	Hospital admissions for the full available period post RTC	Max years of full available data
1997	Mean cost	5,084	5,467	7,147	9
	Total cost	5,932,547	6,380,072	8,340,429	
	Cumulative %	71%	76%	100%	
1998	Mean cost	4,363	4,763	5,717	8
	Total cost	4,808,022	5,248,404	6,299,905	
	Cumulative %	76%	83%	100%	
1999	Mean cost	5,304	5,846	6,958	7
	Total cost	5,818,561	6,413,069	7,633,199	
	Cumulative %	76%	84%	100%	
2000	Mean cost	5,513	5,867	6,833	6
	Total cost	6,036,783	6,424,866	7,481,681	
	Cumulative %	81%	86%	100%	
2001	Mean cost	5,626	5,915	6,630	5
	Total cost	6,549,096	6,884,524	7,716,926	
	Cumulative %	85%	89%	100%	
2002	Mean cost	5,033	5,616	6,089	4
	Total cost	5,476,265	6,110,563	6,625,109	
	Cumulative %	83%	92%	100%	
2003	Mean cost	6,478	6,878	7,132	3
	Total cost	6,386,819	6,781,679	7,032,189	
	Cumulative %	91%	96%	100%	
2004	Mean cost	5,638	6,099	6,228	2
	Total cost	5,508,304	5,959,080	6,085,009	
	Cumulative %	91%	98%	100%	

Table 40 Mean and total costs (\mathbf{f}) of, hospital admissions by time of road traffic crash

4.3.2 Cost impact of road traffic crashes at safety camera sites

Out of 1186 RTCs that were identified at the camera sites, 854 (72%) occurred before safety camera installation (Pre SCI), while 64% of days occurred Pre SCI. The daily rate of RTCs declined by 31% comparing before and after safety camera installation (Pre and Post SCI).

	Pre SCI	Post SCI
RTC	854	332
% of row	72%	28%
Days	98,717	55,725
% of row	64%	36%
Daily RTC rate	0.0087	0.0060

 Table 41 Proportions of road traffic crashes and days Pre and Post SCI

There were 1,167 and 440 injured casualties respectively in the Pre SCI and Post SCI groups of RTCs. The mean casualty rate per RTC was 1.37 and 1.33. There appeared, however, to be a greater difference in the severity of casualties in the Pre SCI and Post SCI groups (Table 42), where 1.8% of casualties in the Pre SCI group were fatalities compared to 0.5% in the Post SCI group. There was also a smaller difference in the proportion of casualties admitted to hospital (11.9 Pre SCI and 10.5% Post SCI).

The mean costs for (surviving) casualties admitted to hospital were higher in the Pre SCI than in the Post SCI group (\pounds 6,200 and \pounds 4,742, respectively, Table 42). The mean costs for fatalities were not compared as there was only one casualty in the Post SCI group that incurred a hospital cost (the other casualty died before hospital admission) and that single hospital cost was relatively high, which further skewed the total mean cost.

Costs per day (for surviving hospital casualties) Pre SCI were over double those of the Post SCI group ($f_{...,8.8}$ and $f_{...,3.9}$ respectively).

The total cost per day was 36% less in the Post SCI group compared to Pre SCI.

	Pre SCI	re SCI				Post SCI			
	Count**	Mean cost	£ per day	Col %	Count	Mean cost	£ per day	Col %	
Sight	1,007			86.3	392			89.1	
Hospital admissions*	139	6,241	8.8	11.9	46	4,742	3.9	10.5	
Fatal (within 6 months)	21	9,314	0.8	1.8	2	120,727	2.2	0.5	
Total	1,167	6,408	9.5	100.0	440	7,210	6.1	100.0	

Table 42 Total hospital admission	cost up to 6 months p	ost road traffic crash	by injury
severity			

*surviving casualties **count includes all casualties -not only the ones that incurred costs

4.3.2.1 Road traffic casualty cost by road user: safety camera sites

The overall mean casualty hospital cost declined by 24% (Table 43). The mean costs for drivers declined by 53%, while mean passenger costs actually increased by 51% (from $f_{2,3,505}$) while pedestrian mean costs remained similar.

The daily cost declined by 55% in total (drivers 64%, passengers 14% and pedestrians 56%).

					SCI	ompre
		Count	Mean cost	£ per day	In mean costs	In daily costs
Pre SCI	Driver or rider	44	6,327	2.8		
	Vehicle or pillion passenger	25	2,314	0.6		
	Pedestrian	70	7,589	5.4		
	Group Total	139	6,241	8.8		
Post	Driver or rider	19	2,983	1.0	-53%	-64%
301	Vehicle or pillion passenger	8	3,505	0.5	+51%	-14%
	Pedestrian	19	7,022	2.4	-7%	-56%
	Group Total	46	4,742	3.9	-24%	-55%

Table 43 Total hospital admission (survivors only) costs up to 6 months post road traffic crash, by road user

4.3.2.2 Road traffic casualty cost by injury type: safety camera sites

Mean costs declined for all injury groups apart from head/neck injuries (increased by 22%, Table 44). The highest means were found in lower limb injuries (£10,600 Pre SCI and £7,300 Post SCI). This group comprised of a high proportion of hip and thigh injuries, which commonly require relatively long hospital stay.

The daily costs declined in all injury categories, but especially lower limb injuries (-72%).

					% change fror	n Pre SCI:
		Count	Mean cost	£ per day	mean costs	daily costs
Pre SCI	Lower limbs	56	10,602	6.01		
	Head/neck	43	4,307	1.88		
	Upper limbs Abdomen, lower back,	5	3,238	0.16		
	lumbar spine and thorax	29	2,120	0.62		
Post SCI	Lower limbs	13	7,285	1.70	-31%	-72%
	Head/neck	19	5,253	1.79	+22%	-5%
	Upper limbs Abdomen, lower back,	2	1,913	0.07	-41%	-58%
	lumbar spine and thorax	9	1,909	0.31	-10%	-50%

 Table 44 Injury costs (surviving) hospitalised casualties Pre SCI and Post SCI

4.3.2.3 Cost by road traffic crashes: safety camera sites

Mean maximum cost per RTC was higher in the Pre SCI group, compared to Post SCI, for the hospital survivor group (\pounds 6,500 and \pounds 5,000 respectively, Table 45). The mean cost per day was over double in the Pre SCI hospital group compared to the Post SCI group (£8.6

vs. $\pounds 3.8$) –a reduction of 56%.

RTC severity	Pre SCI				Post SCI				
	Count	Mean cost	£ per day	Col %	Count	Mean cost	£ per day	Col %	
Slight injury only	706			82.7	288			86.7	
Hospitalised survivors	129	6,548	8.6	15.1	43	4,962	3.8	13.0	
Fatal with hospital stay	8	9,314	0.8	0.9	1	120,727	2.2	0.3	
Fatal only	11			1.3					
Total	854	6,709	9.3	100.0	332	7,593	6.0	100.0	

Table 45 Costs by type of road traffic crash

4.3.2.4 Cost by type of road traffic crash at junction / non-junction: safety camera sites

Mean (max) costs per non-junction RTC was higher in the Pre SCI group compared to Post SCI –this cost declined by over half. Mean cost per day, for the same types of RTC, declined by 72%. Junction RTCs, on the other hand, had a slight increase (+15%, Table 46) while the mean cost per day declined by about one third.

					% change from Pre SCI		
		Count	Mean	£ per day	Mean costs	Daily costs	
Pre SCI	Not near junction	63	7,744	4.94			
	At or near junction*	66	5,406	3.61			
Post	Not near junction	21	3,624	1.37	-53%	-72%	
501	At or near junction*	22	6,240	2.46	+15%	-32%	

Table 46 Costs by junction / non-junction type of road traffic crash, hospitalised survivors only

*within 20 metres of junction

4.3.2.5 Cost by type of road traffic crash and number of vehicles involved: safety camera sties

The mean cost of single vehicle RTCs declined little, but the daily cost of these declined by 53% (Table 47). The costs of two vehicle RTCs, on the other hand, declined substantially in both categories (-57% for mean cost and -65% for daily costs). The costs for multiple vehicles (three or more) actually increased slightly in the mean costs, but declined in daily cost (NB the numbers in this category were fairly small).

					% change from Pre SCI	
		Count	Mean	£ per day	mean costs	daily costs
Pre SCI	Single vehicle	84	6,625	5.64		
	Two vehicles	35	7,355	2.61		
	Three or more vehicles	10	3,078	0.31		
Post SCI	Single vehicle	23	6,475	2.67	-2%	-53%
	Two vehicles	16	3,149	0.90	-57%	-65%
	Three or more vehicles	4	3,516	0.25	+14%	-19%

Table 47 Cost by number of vehicles involved in road traffic crash, hospitalised survivors only

4.4 Summary of Chapter 4

- The frequency of RTC casualties declined significantly over the study period (approximately 18-21%).
- Males were more likely than females to have a fatal outcome of an RTC injury. Young people constituted the largest casualty group although they had a low fatality rate. Old people were more likely to suffer severe or fatal injuries.
- Less protected road users (including pedal cyclists, pedestrians and motorbike/moped occupants) had more severe and fatal injuries. Pedestrians and pedal cyclists were most likely to suffer head injuries, while car occupants more frequently incurred injuries to the thorax and abdomen/lower back/lumbar spine.
- Police RTC records of serious injuries appears underreported i.e. 45% of hospitalised casualties were not recorded by police. The following RTC casualty characteristics were significantly associated with underreporting: no third party involvement, age (older age), year (casualties from earlier years), vehicle occupant (especially pedal cyclist), length of stay (day cases) and gender (females).
- Seriously injured STATS19 casualties also declined in frequency more than the SMR01 would indicate (38% and 21% respectively). At the same time linked

SMR01 casualties that were coded slight by the police increased with 5%, while linked SMR01 casualties coded serious declined 27%.

- RTCs at safety camera sites comprised 2% of all RTCs in Strathclyde region and the RTC epidemiology at camera sites before camera installation did not differ significantly from the remainder of Strathclyde region (apart from having more severe injuries and multiple vehicle RTCs).
- Compared to the rest of Strathclyde, there was a significantly greater downward linear time trend at the camera sites.
- Camera effect on RTCs over time appears stable and cameras appear to be effective in reducing RTCs with serious or fatal outcome, as well as multiple-vehicle and non-junction RTCs.
- The total inpatient costs (excluding unlinked SMR01 records) were conservatively estimated to be £53 million over the study period which equates a yearly average of £7.3 million.
- Injuries to the head and to the lower extremities incurred the highest costs (28% and 34% of total costs respectively). Occupants of motorbikes or mopeds incurred the highest mean costs. Pedestrian, constituting 36% of casualties incurred 44% of total costs.
- Casualties from deprived areas, and pedestrians in particular, incurred higher mean hospital costs than other road user groups. Drivers from the most affluent SIMD areas incurred the highest mean cost for drivers.
- Fatality rates and mean costs increased with age.

- 71% of hospital admission costs were incurred within the first six months following an RTC and a further 5% in the following year (for RTCs occurring in 1997).
- The mean costs for (surviving) casualties admitted to hospital declined by 24% Post SCI (£6,200 vs. £4,742) and the mean daily cost declined by 55% i.e. costs per day Pre SCI were £8.8 and Post SCI £3.9.
- Daily costs for pedestrian injuries declined by 56% comparing Pre and Post SCI and, in terms of injuries, the daily costs appear to have declined most for lower limbs (-72%).
- Comparing Pre and Post SCI, daily costs for non-junction RTCs and RTCs involving two vehicles declined by 72% and 65% respectively and mean costs declined 53% and 57% respectively.
CHAPTER 5. DISCUSSION

Contents of Chapter 5

5.1 M/	AIN FINDINGS	
5.1.1	Epidemiology of road traffic injuries in Strathclyde region	
5.1.2	Accuracy of police reported road traffic casualties	146
5.1.3	Epidemiological impact of safety cameras	
5.1.4	Cost of road traffic crashes in Strathclyde region	
5.1.5	Cost of road traffic crashes at safety camera sites	148
5.2 IN	ERPRETATION OF RESULTS	
5.2.1	Epidemiology of road traffic casualties in Strathclyde region	
5.2.2	Accuracy of police reported road traffic casualties	
5.2.3	Epidemiological impact of safety cameras	
524	Cost of road traffic crashes in Strathclyde region	154
5.2.5	Cost of road traffic crashes at safety camera sites	
5.3 ST	RENGTHS AND WEAKNESSES OF STUDY	
531	Strengths	158
5.3.1	1 Validity of data sources and linkage	158
5.3.1.	2 Size of sample and follow-up admissions	
5.3.1.	3 Regression towards the mean	
5.3.2	Weaknesses	
5.3.2.	1 Before and after study	
5.3.2.	2 Retrieving the data	
5.3.2.	3 Validity of data sources and linkage	
5.3.2.	4 Errors related to analyses and results	
5.4 IM	PLICATIONS	
5.5 UN	IANSWERED QUESTIONS	166
5.6 Su	IGGESTIONS FOR FURTHER RESEARCH	167
5.6.1	Suggestions for continuing research on the linked data	168
5.6.1.	1 Study setting	
5.6.1.	2 Data sources and linkage	
5.6.1.	3 Analyses	
5.6.2	Suggestions for other research	169
5.6.2.	1 Epidemiology of road traffic crashes in Strathclyde	
5.6.2.	2 Accuracy of police reported casualties	
5.6.2.	3 Impact of safety cameras	
5.6.2.	4 Cost of road traffic crashes	
5.7 SL	IMMARY OF CHAPTER 5	
5.7.1	Epidemiology of road traffic casualties in Strathciyde	
5.7.2	Accuracy of police reported road traffic casualties	
5.7.3	Epidemiological impact of safety cameras	
5.7.4	Cost impact of road traffic crashes in Strathclyde	174
5.7.5	Cost impact of road traffic crashes at safety camera sites	176

5.1 Main findings

The presentation of this section follows the order of the rest of this document beginning with the epidemiology of RTCs in Strathclyde region, which is here followed by a separate section on accuracy of police recording. The results from analysis of cost implication are divided in to two parts (whole of Strathclyde and at safety camera sites) impact.

5.1.1 Epidemiology of road traffic injuries in Strathclyde region

Over the years 1997 to 2005 estimates suggests that nearly 89,000 people were injured on Strathclyde roads, which constitutes about 4% of Strathclyde's total population of 2.3 million. RTC injuries have, however, declined approximately 18-21% over the study period.

Young people constituted the largest casualty group (23%) but they had a low fatality rate (0.6% of casualties died within 1.5 years). By contrast,, people aged 65 or over had a low proportion of casualties (8%), but were more likely to suffer severe or fatal injuries (4% of casualties died within 1.5 years).

A higher proportion of police recorded male casualties were admitted to hospital than females and there were more male casualties in total. Males were also more likely than females to have a fatal outcome of an RTC injury.

In terms of deprivation, 38% of casualties were from the fifth most deprived areas, while only 12% were from the most affluent areas. No pattern in terms of deprivation relating to fatality risk was found.

Less protected road users, including pedal cyclists, pedestrians and motorbike/moped occupants were more likely to sustain severe injuries compared to car occupants. Motorbike and moped occupants were also most likely to have a fatal outcome.

145

Head injuries, which were the most common injury of total admitted casualties (44%), were most frequently suffered by pedestrians and pedal cyclists, while car occupants more frequently incurred injuries to the thorax and abdomen/lower back/lumbar spine.

5.1.2 Accuracy of police reported road traffic casualties

Over the study period police road casualty data appeared to underreport; 45% of hospitalised casualties were not recorded by police. The following RTC casualty characteristics were significantly associated with underreporting: no third party involvement, age (progressively higher risk with older age), year (casualties from earlier years appeared to be more under-recorded), type of road user (especially pedal cyclist), length of stay (day cases were less likely to be recorded by police) and gender (females were less likely to be recorded by the police).

Looking at STATS19 and SMR01 separately suggests that casualties coded as seriously injured in STATS19 declined in frequency more than the SMR01 RTC casualties (38% and 21% respectively). The analysis of the linked STATS19-SMR01 data showed that, while linked SMR01 casualties coded *serious* declined 27%, linked SMR01 casualties that were coded *slight* by the police increased by 5% over time. Head injuries were increasingly coded as slight (from 35% to 50% of head injury cases) and less often coded as serious over time. At the same time, mean hospital length of stay increased for this group indicating that head injuries were unlikely to have become less severe. A similar pattern was found in neck, thorax and abdominal/lower back/lumbar spine injuries.

5.1.3 Epidemiological impact of safety cameras

A total of 59,608 RTC records were obtained for Strathclyde region (STATS19) and of these 1,186 (2%) occurred at safety camera sites. The RTC epidemiology at camera sites

before camera installation did not differ significantly from the remainder of Strathclyde region (apart from having more severe injuries and multiple vehicle RTCs).

There was a significantly greater downward linear time trend at the camera sites compared to the rest of Strathclyde (32% vs. 16% respectively).

Cameras appeared to be effective in reducing RTCs with serious or fatal outcome, as well as multiple-vehicle and non-junction RTCs and the camera effect on RTCs over time appeared stable.

5.1.4 Cost of road traffic crashes in Strathclyde region

The total inpatient costs (excluding unlinked SMR01 records) were conservatively estimated to be \pounds 7.3 million yearly (\pounds 58 million in total over 8 years). The highest costs were incurred by injuries to the head and to the lower extremities (28% and 34% of total costs respectively).). The highest mean costs (\pounds 12,241), which were at least double the cost of any other single injury, were incurred by injuries to the hip or thigh.

Although constituting only 36% of casualties, pedestrian RTCs incurred 44% of total costs, while the highest mean costs were incurred by occupants of motorbikes or mopeds. Costs incurred by pedestrians from the most deprived areas were more than the costs of all the other pedestrians combined. The highest mean cost for drivers was for the most affluent SIMD group while lowest for the passengers in the same group.

Speed had a great impact on pedestrian casualties with mean hospital costs of surviving casualties of £6,800 if hit on a 30 mph road, £11,000 on 40-50 mph road and £24,000 on 60-70 mph road. This was also reflected in fatalities, where 2%, 7% and 16% of pedestrian casualties were killed on 30, at 40-50 mph and at 60-70 mph respectively. The large

majority of pedestrian fatalities, however, occurred on 30 mph roads, generating over \pounds 22 million in total hospital costs (38%).

Mean costs and fatality rates increase by age; the mean hospital costs of surviving casualties rose progressively with advancing age from £4,000 to nearly £11,000 and fatality rates increased from 0.6% to 4%.

For RTCs occurring in 1997 (the group with the longest follow-up period), 71% of hospital admission costs were incurred within the first six months following an RTC and a further 5% in the following year.

5.1.5 Cost of road traffic crashes at safety camera sites

Of all casualties before safety camera installation (Pre SCI), 11.9% were hospitalised and 1.8% were fatal, while 10.5% of casualties after safety camera installation (Post SCI) were hospitalised and 0.5% fatal.

The mean costs for (surviving) casualties admitted to hospital declined by 24% after safety camera installation (£6,200 Pre SCI and £4,742 in the Post SCI group) and the daily cost declined by 55% i.e. costs per day (for surviving hospital casualties) Pre SCI were £8.8 and £3.9 Post SCI.

Daily cost reductions were most marked for pedestrian injuries (56%), lower limb injuries (72%), casualties injured in non-junction RTCs (72%) and two vehicle RTCs (65%).

5.2 Interpretation of results

This section comprises the main findings in light of other research and an interpretation of what these may mean. The section follows the same structure as the section "main results" 5.1 and endeavours to refer to all findings presented there.

5.2.1 Epidemiology of road traffic casualties in Strathclyde region

A reduction in RTC frequency is currently expected in high-income countries [15]. This study is in line with those predictions and confirms that there has been a significant general decline in RTC casualties over the nine year study period. Comparing the first three years with the last indicates a reduction of 18% in all STATS19, 38% in killed and seriously injured (KSI) STATS19 and 21% in SMR01 records. The possible reasons for the difference in KSI and SMR01 are discussed further later in this section.

Similar to other studies [27,33,83] the results showed that young people made up the largest proportion of casualties but have a low fatality rate. Older casualties, on the other hand, were more likely to sustain more severe injuries that required hospitalisation or resulted in fatalities.

More males than females had a fatal outcome of an RTC injury which both confirms [3] and contradicts published results [83,84]. One study found that especially females aged 20-35 would be more at risk of a fatal outcome compared to males while findings presented here show the opposite, but this could be due to study design e.g. they were estimating risks including controlling for seatbelt use etc while this study suggests the proportions of fatalities of injured casualties (i.e. the risk of a fatal outcome when e.g. not wearing seatbelt may be higher for females, but perhaps more females wear seatbelts hence resulting in more females than males surviving).

Similar to previous findings [87], severe injuries appeared to be more likely for less protected road users i.e. pedal cyclists, pedestrians and occupants of motorbikes and mopeds. Occupants of motorbikes and mopeds were also at highest risk of dying, probably reflecting a combination of high speed and being relatively unprotected.

Type of road user was also correlated with injury type, where head injuries were common among pedestrians and pedal cyclists and knee and lower leg injuries sustained foremost by pedestrians and occupants of motorbikes/mopeds. Car occupants, on the other hand were more likely to suffer injuries to the thorax and abdomen/lower back/lumbar spine. This is in line with other research [34,88,90,91] and is likely to reflect different circumstances in terms of speed and protection.

Collisions produced, on average, less severe injuries than non-collisions (both in terms of hospitalised and fatal casualties), but these results may vary by, for example, road user. Deprivation appeared to have had no effect on severity at this level of descriptive analysis, but is discussed further in relation to road user in Results section 4.3.1.4 and Discussion section 5.2.4.

5.2.2 Accuracy of police reported road traffic casualties

Underreporting of road traffic casualties in police records has been well documented (see further section 1.4) and relying only on police data of road traffic crashes may therefore be misleading [106-110]. A specific cause for concern is the reporting of casualty severity by police forces and it is thought that injuries labelled as serious are particularly underreported. For that reason, part of the general epidemiological analysis of the study was undertaken to investigate the extent of possible underreporting of casualties due to road traffic crashes (RTCs) in Strathclyde region. The results suggest a consistent underreporting in police road casualty data of around 45%. This is a similar figure to that reported in previous studies linking emergency department or hospital data with STATS19 data [103,118,131].

There are numerous possible reasons why a proportion of total casualties are unrecorded in STATS19 by the police. For example, when only the driver is injured and the only damage caused is to their vehicle, there is no legal requirement to report an RTC. Underreporting, however, seems to occur particularly apply in RTCs involving motorcycles or pedal cycles when no other vehicles are involved. This could be because road users may regard police reports as being necessary only when insurance claims arise. This hypothesis is consistent both with the findings in this study and those of other studies [104,105,109,115,116]. Additionally, the results showed that age, gender, year of RTC and length of stay in hospital were significantly associated with underreporting. Road users driving under the influence of alcohol or drugs, driving stolen vehicles or driving with no insurance, driving licence or MOT (roadworthiness certificate) were also less likely to bring their RTC to the attention of the police, in particular if no other vehicle was involved. Casualties who leave the RTC site without giving their details to police or other people in the incident are not recorded. In theory police forces should update the severities of casualties on the reports but in practise do not always do so. There appears to be no system for healthcare professionals to update police officers about changes in severity e.g. the police may not be aware that a casualty has subsequently been admitted to a ward and therefore should be labelled as "serious". Lastly, there may be a recording problem with patients admitted to hospital that fraudulently claim to have been involved in an RTC or exaggerate the injury symptoms, possibly to obtain compensation or to obscure the real reason for their injuries.

As matching links of casualties from neighbouring health boards also were included (4%, i.e. 444 out of 10,005 hospital admissions – excluding immediate deaths), there would also be a proportion of unlinked SMR01 records that had a matching STATS19 record in a

neighbouring police force too. Consequently, if an equal number of SMR01 records in Strathclyde matched a STATS19 elsewhere, the actual proportion of RTC SMR01 records that were not recorded by the police would be 43% rather than 45%.

In terms of casualty severity, these results are in agreement with those of Gill et al [114] implying there has been an increase over time in the underreporting by the police of serious casualties. Seriously injured STATS19 casualties declined in frequency more than the SMR01 would indicate. There are several potential explanations for this (of which some also were highlighted by the Department for Transport [186]).

Firstly, there may have occurred a true, disproportionate fall in the number of serious casualties caused by RTCs, but as the reduction was not reflected in the number of fatal injuries or total injuries in SMR01 this seems improbable. Alternatively, public reporting of serious RTCs (as in public reporting to the police) or the proportion of less severe casualties being treated in emergency departments may have declined. There is, however, little evidence in favour of either explanation. Hospital admission policies, such as a rise over time in hospital admissions of less severe road casualties, may have occurred as there was some evidence of an increase in A&E admissions in Glasgow in earlier years [187]. Hospital reporting of casualties may have changed over time but the same reporting standards have been used in Scotland for decades and the completeness of SMR data are audited regularly in terms of coding and accuracy with consistent results over time [188].

Perhaps the most likely explanation is that some types of casualty, formerly recorded by the police as severe, were in the later study years recorded as slight (i.e. a shift in police judgement of casualty severity). In the light of the length of stay analysis these casualties may, to a large extent, be those with a head or neck injury. The hospitalised casualties with injuries to hand/wrist or ankle/foot also seem more likely to have been coded slight in later years.

5.2.3 Epidemiological impact of safety cameras

As indicated in section 1.7.3, previous evaluations of speed cameras have found that they contribute to reducing the incidence of RTCs [150-156], fatalities and injuries[158-160], although so far the evidence is not conclusive [161]. There appear to be no studies that have reported the epidemiological impact or the type of crashes that are influenced by cameras. Such information could help in the future deployment of cameras where they are likely to exert the greatest effect on RTC rates.

This component of the study was designed to compare the pattern and types of RTCs before and after the installation of the cameras in the Strathclyde region and to quantify their effectiveness in reducing the incidence and severity of associated casualties.

Analysis included a large-scale epidemiological evaluation of the impact of safety cameras on the incidence, severity and type of RTCs in a high-risk region of the United Kingdom for fatal and serious road casualties [16]. The findings, in keeping with those of others [154,155,189], suggest that safety cameras have contributed a substantial reduction in the incidence of RTCs and especially of serious injury RTCs [158,159]. The results of the present study show a significantly greater downward linear time trend at the camera sites compared to the rest of Strathclyde. The findings have, in addition, enhanced understanding of which types of crashes the cameras have impacted most upon, and where cameras might be most effectively located in the future.

Before camera installation, the RTC epidemiology at camera sites did not differ significantly from the remainder of Strathclyde region. The main difference was that camera sites had, overall, more RTCs in which someone was killed or seriously injured. This confirms that the selection method of sites was related to RTC severity. There also appeared to be more multiple vehicle RTCs (three or more vehicles) at camera sites Pre SCI than the remainder of Strathclyde. The findings indicate that cameras were effective in reducing the incidence of fatal and serious RTCs at camera sites. This was especially true with RTCs involving multiple casualties and vehicles. Cameras also appeared to have a greater effect on non-junction RTCs than on those occurring at junctions. Somewhat counterintuitively, the effectiveness of cameras in reducing RTC rates did not appear to diminish over time.

Although the installation criteria based on speed and severity of RTCs seem appropriate and effective, the addition of other indicators (e.g. rates of non-junction and multiple vehicle RTCs on sites) might aid the future selection of camera sites and enhance the impact of cameras further.

5.2.4 Cost of road traffic crashes in Strathclyde region

As discussed in section 1.5, little is known about the long-term health and economic consequences of RTCs, and the financial costs are probably greatly underestimated [190].

The purpose of this part of the study was to provide an economic analysis of the costs attributable to hospitable admissions due to road traffic crashes in the Strathclyde region of Scotland.

The total cost of inpatient and day surgery treatment following RTC in Strathclyde police area was conservatively estimated at \pounds 7.3 million per year (excluding the unlinked SMR01 records).

Extremely high costs were incurred by injuries to the lower extremities (hip/thigh and knee/lower leg) and head injuries, compared to other injuries (34% and 28% of total costs respectively), which is consistent with estimates from Australia (33% and 27% respectively) [136] and the USA [137].

Pedestrians are a vulnerable group of road users and, although only constituting 36% of hospitalised RTC casualties (and 20% of total injured casualties), they incurred 44% of total hospital costs (for survivors). By contrast, a study from New Zeeland showed that pedestrians, comprising 10% of hospitalised RTC casualties, incurred 18% of total costs [138]. In the study, fatality rates for pedestrians were about double those of other injured road users (2.3% compared to 1.3% of drivers/riders and 0.8% of passengers).

Unsurprisingly, drivers of motorcycles/mopeds had the highest mean cost, probably reflecting the speed and vulnerability of these road users.

Both posted speed limit and junction context had an effect on RTC outcome, but this varied by type of road user. All road users, but especially pedestrians, had increasingly poorer outcomes with higher posted speed limits (with higher rates of fatal results). Predictably, non-junction RTCs were worse for drivers and passengers than junction RTCs, but appeared to make no difference for pedestrian casualties.

Casualties from deprived areas, and pedestrians in particular, incurred higher mean hospital costs reflecting more severe injuries. The mean cost for drivers was highest for the most affluent SIMD group while lowest for the passengers in the same group. It is possible that there is a varying degree of utilisation of passenger seatbelt and child car seats in different SIMD groups. In future analyses, exploring the interactions between age, gender (e.g. young male driver) and deprivation variables together with numbers of casualties in a vehicle (e.g. whole family or single driver injury) might yield further insights into this finding.

In terms of age, the youngest and oldest groups had the highest admission rates, possibly reflecting their predominance as pedestrians. The vulnerability to injury by age is revealed by both the mean hospital costs and fatality rates, the highest of which was found in the oldest age group. Only 71% of hospital admission costs were incurred within the first six months following a RTC (for RTCs occurring in 1997) and a further 5% in the following year. The remaining costs arise from admissions for the period between 1.5 and nine years post RTC. It may therefore be reasonable to increase the estimate of the full hospital costs accordingly.

The Department for Transport [134] has estimated a cost of serious injury of about \pounds 155,000, including medical/ambulance costs of \pounds 11,000 (for 2004). The estimate for hospital costs was only \pounds 6,000 – only about 4% of the total estimated costs.

In summary, the results show the uneven burden of suffering and costs of RTCs with the most costly RTCs occur in areas with high levels of deprivation, a history of pedestrian RTCs and/or elderly and children, areas with many non-junction RTCs and 30 mph roads.

5.2.5 Cost of road traffic crashes at safety camera sites

The mean costs and costs per day appeared to decline in most categories for both casualties and RTCs. The daily hospital cost for the sites declined by over half Post SCI.

There also appears to have been a shift in severity in the proportions so that of all injured in the Post SCI group, 11% were hospitalised or fatal while this proportion was 14% in the Pre SCI group i.e. casualties were more likely to be severely or fatally injured Pre SCI than Post SCI.

There appears to have been a reduction in pedestrian hospitalised injuries (for which the daily cost declined by 56%) although the mean cost for an hospitalised pedestrian appeared similar to Pre SCI. The overall mean casualty hospital cost for casualties declined by 24% for all road users, suggesting that the admitted casualties were less severely injured (shorter length of stay and, as a result, lower cost).

The mean costs appears to have declined most for lower limbs (and upper limbs, but these were very small numbers) indicating that perhaps cameras aid in reducing the severity of crashes causing these type of injuries. The cost per day was also greatly reduced for the upper limb category.

In line with previous findings, safety cameras appeared to have a greater impact on nonjunction RTCs compared to junction RTCs. As the mean cost for hospitalised (surviving) casualties declined by over 50% it is possible that cameras encouraged slower speeds that in turn resulted in fewer severe injuries. The impact of speed in relation to injury severity is well documented [51,83,92,93] i.e. the higher the speed the more severe injuries. As the daily costs also declined (-72% for non-junction crashes), this too may reflect lower speeds and fewer RTCs.

The costs of hospitalisation in RTCs with two vehicles involved declined both in mean cost per RTC and cost per day. This could indicate that these types of crashes were both less severe and less frequent Post SCI.

The Department for Transport (DfT) [134] has estimated a cost of serious injury of about \pounds [155 thousand, including medical/ambulance costs of \pounds [11 thousand (for 2004). The mean casualty hospital costs in this study were estimated at only \pounds 6 thousand Pre SCI and \pounds 5 thousand Post SCI – about 3-4% of the DfT total estimated costs. If a rough estimate was made using DfT's approximation, the cameras may have contributed to the savings as follows: 139 seriously injured casualties (hospitalised) in the Pre SCI period would cost \pounds [21,545,000 (applying DfT costing estimate) over 98,717 days (= daily rate \pounds [218.5) and in the Post SCI period 46 seriously injured would cost \pounds [7,130,000 over 55,725 days (= daily rate \pounds [127.9). If the daily cost rate of the casualties injured during Post SCI period was the same as for the Pre SCI period, the total costs would have been 12,161,989 for that period, which in turn is a saving of \pounds [5,031,989. This is considering the limited Post SCI period only and no fatal or slight injury costs.

Estimating cost-benefits of safety cameras require further investigation, particularly into the additional costs of slight injuries and surrounding costs incurred by RTC (these are thought to be substantial, see further strengths and weaknesses). However, given that the mean costs of crashes were reduced by over half for hospitalised casualties and over a third for total casualties, it is possible that cameras are cost-beneficial, appearing to reduce both the severity and frequency of RTCs. These results, together with previous findings that indicate that cameras do not lose effect over time, may be useful indicators for further econometric analyses.

5.3 Strengths and weaknesses of study

The section begins with strengths including validity of data sources and linkage discussing the two main sources of data (STATS19 and SMR01) and the rate of false positive and false negative as calculated by ISD. This is followed by a discussion of the merits of the sample size and follow-up admissions. The section ends with a discussion of weaknesses including regression to the mean.

5.3.1 Strengths

The study is overall relatively unique and considering the difficulties incurred in the first years of study the achievements are substantial. There is, however, considerable scope for continuing analysis with the data and this is discussed further in section 5.6.1.

5.3.1.1 Validity of data sources and linkage

The two main data sources utilised in this study are problematic when analysed separately. The linked database, however, provides a much more solid basis for analysis. STATS19 has been utilised as a recording system in its current format since 1979 and nearly all road traffic crash statistics are derived from this source. STATS19 is, however, known to underascertain some categories of RTCs, mainly those associated with injuries involving pedestrians, motorcyclists and pedal cyclists [115,116], a finding that was confirmed in this study. Additionally, the reliability of police severity judgement is questionable. Utilising linked hospital and police RTC data is much preferable in analysing RTCs as the SMR01 system h is one of the first and most complete national health data sets in the world [171].

ISD had previously performed several linkages and the method of linkage used had in the past been shown to have 3% false positive and 3% false negative rates [183]. The linkage used a range of variables for matching and, according to ISD, crude estimates indicate that the false negative rate was 2.6% - i.e. 2.6% (1,802) of the unlinked STATS19 records were missed links (i.e. should have been linked). On the other hand, a crude estimate of the false positive rate was 15.6% - i.e. 15.6% (1,607) of the linked STATS19 records were false positives (i.e. should not have been linked). As a result, the number of missed linked records.

5.3.1.2 Size of sample and follow-up admissions

The data included the full (recorded) RTC population of Strathclyde region and the study period was extensive, which together enhanced statistical power.

An unusual feature if the linked database was its inclusion of follow-up admissions. This especially improved the cost analysis, but also provided a greater understanding of the severity of RTC in terms of total length of hospital stay and how hospital admissions, related to the RTC, continued over many years.

5.3.1.3 Regression towards the mean

Among the advantages of the relatively long study period of is that regression towards the mean (RTM) and random fluctuations should have been The time periods before camera installation ranged from 2 to 7 years, depending on when the cameras were installed. Thus any random large peaks in RTCs should have been smoothed out. RTM could, however, still play a role, but probably not sufficient to invalidate the findings that cameras are effective in reducing RTC.

5.3.2 Weaknesses

5.3.2.1 Before and after study

The foremost weakness of this study is that it used a before-and after evaluation design rather than the more robust methodology of a randomised controlled trial (RCT). An RCT is, however, difficult to perform in the field of public policy and was not an option at the time this study was initiated, particularly as the cameras were mostly already installed. Before and after studies may suffer from several problems such as data artefacts (such as changing classification of severity), statistical error (due in part to small sample sizes), bias (e.g. the police expect cameras to be effective) or confounding variables (such as safer vehicles or better designed roads).

5.3.2.2 Retrieving the data

A key weakness of this study was that the procedure to attain the linked data was rather prolonged and the author faced several difficulties in achieving this. The consequence was that some analyses had to commence using STATS19 only. This especially affected the evaluation of safety cameras. There were limited data available on RTCs and its consequences. No routine data from emergency departments currently exist, nor records from primary care or outpatient hospital care. Additionally, data from social security, employers or insurances would be desirable in order to provide an insight in to sick leave and economic consequences.

5.3.2.3 Validity of data sources and linkage

Although the method utilised for linkage previously has shown to have 3% false positive and 3% false negative rates [183], this linkage may, due to the limited identifying information available, , not have achieved such accuracy. Thorough clerical checking of a large sample of best matching pairs at a wide range of probability weights was, however, completed by ISD.

There was also a potential bias in that matching links of casualties from neighbouring health boards was performed. Hence, there might have been a proportion of unlinked SMR01 records that had a matching STATS19 record in a neighbouring police force (but not linked in the study). Consequently, the estimated underrecording was likely to be 43% rather than 45% (see section 5.2.2).

There may have been errors relating to the decision process of including follow-up admissions as there was no way of knowing whether or not an admission was related to the initial RTC. The inclusion period, however, was limited to 1.5 years post RTC for cost and LoS analysis and care was taken when including admissions beyond six months post RTC i.e. only admissions close in time or related statistically to the injury at admission were included (see further section 3.4.2.2).

5.3.2.4 Errors related to analyses and results

In analysing costs and safety cameras, the data were not population based. Comparing the mean costs, however, provides reasonable estimates of the severity and cost and for the camera analyses daily rates were utilised instead.

All cost estimates in the study are likely to be extremely conservative for several reasons. Firstly, the costs presented were only incurred by acute hospital treatment; no outpatient, GP, ambulance or other health service resources were included. Large costs are also likely to have been incurred by loss of work (lost output), property damage, insurance and human costs, none of which were accounted for. Secondly, most of the analysis was limited to the first 1.5 years admissions post-RTC, for ease of comparison (probably including less than 76% of hospital costs, see results section "costs over time"). Lastly, unlinked SMR01 RTCs were not included as data on follow-up admissions were unavailable. As they comprised 45% of total traffic related hospital admissions, the costs presented here may therefore be substantially underestimated.

The safety camera cost analysis was limited to include only the first six months' admissions post-RTC, which means some acute costs were not included. The number of hospitalised casualties and RTCs in this part of analysis is not very high and some random effects could possibly have occurred e.g. regression towards the mean. However, long term means (daily rates of injury costs over several years) were utilised so the impact of RTM should have been minimised [154], as discussed earlier in this section.

5.4 Implications

The structure follows a similar pattern to previous section beginning with epidemiological findings (point 1) followed by accuracy of data (point 2-3), safety camera impact (4-5) and cost implications (6-7). The section ends with a suggestion as to how all analyses could be

improved: routinely linking police and hospital records by adding geographical information to SMR01 (point 8).

- 1. Estimates show that about 89,000 people were injured on Strathelyde's roads over a nine year period and of these about 22% (over 19,000) were fatal casualties or casualties that required hospital admission. The results show that, in line with previous findings, young people make up the largest proportion of total casualties while older casualties were more likely to die from a road traffic crash. Furthermore, less protected road users were more likely to suffer severe injuries (especially occupants of mopeds and motorcyclists) and head injuries were more common among pedestrians and pedal cyclists. Although a decline of about 18-21% occurred over the study period the results should inform policy makers that road traffic crashes continue to play a major role in disability and premature death and should remain a high priority. Insights from the study could further help direct resources towards the most vulnerable in traffic: older and younger people, deprived areas and pedestrians.
- 2. There has been a considerable decline in seriously injured casualties according to police reports, but hospital records did not show the same reduction. The results from this study show that the decline in serious injuries probably was caused by an increasing tendency, over time, for police officers to report injuries as slight rather than serious. These findings have implications for road safety policy in Scotland, and perhaps the UK as a whole, and should inform the future collection, interpretation and utilisation of road casualty statistics. In particular, government should re-assess the current achievements against the national targets for 2010 (see section 1.6 for details) which are based solely on police STATS19 data [191]. Additionally, the national safety camera programme also relies heavily on the completeness and accuracy of police data especially fatal and seriously coded

RTCs - in order to site cameras at casualty "blackspots". An overhaul of police RTC recording practices is needed.

- 3. Nearly half of all road traffic casualties that are admitted to hospital are not recorded by police. This means that there are thousands of injuries potentially missing from the databases of the police, local authorities and trunk-route operators who rely on STATS19 information to target dangerous sites for remedial measures such as new road layouts or traffic calming. There are some RTC casualty categories that are more likely than others to be missed by police including situations where no other people were involved in the incident, pedal cyclists and motorbike occupants or older people and females. Missing these road users could have implications for the targeting of road safety education by public agencies and road safety groups. It is very important to take these findings into account in various situations, but especially when updating national statistics and assessing achievement of national targets (as also discussed in the previous section). Finally, the Government's estimates of the total cost to society of not preventing RTCs will need to be radically revised [16,191].
- 4. Safety cameras have been installed over several years to remedy RTC prone stretches of road. There have been many studies showing that speed cameras probably contribute in reducing the incidence and severity of RTCs. The results from this study support earlier findings with the addition that the effect is sustained over time. Consequently, safety cameras should continue to play a major part in RTC prevention policies and expansion should be considered.
- 5. Safety cameras are one of several initiatives to reduce road traffic crashes and are placed on stretches of roads with a history of many RTCs. In order to achieve the most RTC reduction from the cameras, site selection criteria are based on average speeds and incidence and severity of RTCs. This study supports these selection

criteria in that the cameras appeared to be effective both in reducing frequency and severity of RTCs. The results also suggest that cameras may be effective in reducing crashes that are not in the vicinity of a junction and crashes that involves more than two vehicles. Using this information when selecting camera sites may therefore enhance the impact of cameras further. It could be taken into account at the next revision of safety camera regulations or could be considered, more informally, as additional selection criteria should there be too many sites qualifying for a safety camera.

- 6. The most costly RTCs occurred in areas with a history of pedestrian RTCs and/or elderly and child casualties, high levels of deprivation, areas with many non-junction RTCs and 30 mph roads. Pedestrian casualties hit on 30mph roads made up about 40% of total hospital costs (over £22 million); over half of this (£12 million) was incurred by (pedestrian) casualties from the most deprived areas; two thirds of these costs (£8 million) were sustained by the youngest and oldest casualty age-groups. Consequently, considering cost impact and size of different road casualty groups it would be advisable to evaluate current policies and practices in pedestrian traffic injury prevention, particularly in deprived areas with large elderly and child populations. Using a fuller data set (as suggested in the next paragraph) in detecting traffic injury black-spots, could aid in identifying where, geographically, current gaps in road safety exist.
- 7. The results indicate that cameras are cost-effective and that (up to the end of this study) may have saved over £5 million using a calculation based on Department of Transports estimates. The Safety Camera Partnership should utilise these findings further weighing camera costs against the calculated savings in a cost-benefit evaluation.

To further improve the quality of information relating to road casualties, the first 8. step would be to assess current police recording practices. Thereafter, the feasibility of routinely linking road casualty data derived from police and hospitalisation databases should be assessed. This could progress by adding geographical information to RTC hospital data (which should be feasible during a casualty's hospital stay). Knowing where all RTCs occurred would both enhance the quality of the linkage between police and hospital data and identify where hospitalised RTC casualties (that were not reported to or by the police) took place. Finally, the database could be updated over time with follow-up hospital admissions linked to the initial admission (by, for example, using the suggested method of inclusion). This would allow for costs to be routinely attached in a similar manner as for this study. The ideal would, obviously, be to also include RTC and emergency information in such a database, but to date s no routinely collected emergency department data are available. Consequently, by only adding geographical information to the hospital RTC casualty records there is potential to achieve a more or less complete RTC database and to utilise it in a variety of settings - from informing road safety policies to locating and evaluating interventions.

5.5 Unanswered questions

Questions relating to the RTC epidemiology:

 What are the consequences of slight and serious RTC injuries, including analysis of long-term effects (including analysis of primary care data)?

Questions related to the accuracy of police recording:

- 1. Why would the police judgement of casualty severity change over time? This question requires an answer in order to find a remedy to the problem.
- 2. What are the implications, of police underreporting severe casualties, on the national targets for casualty reduction?

Questions concerning prevention of speed related RTCs:

- 1. What are the differences between speed cameras and other traffic calming measures in preventing casualties?
- 2. What are the characteristics of a speed related crash?
- 3. Are the primary causation factors reliable and can they be utilised in further research?

Questions relating to cost:

- 1. Where are the sites/areas with the most costly RTCs and where are sites with the highest total costs (adding RTCs in vicinity)?
- 2. What are the cost benefits of speed cameras, including a full cost analysis of RTC consequences?

5.6 Suggestions for further research

This section is divided in to two parts, namely further research that could extend and/or enhance this study (including addressing weaknesses identified) and secondly, other more general research that could aid in filling the gaps as identified by the unanswered questions.

5.6.1 Suggestions for continuing research on the linked data

5.6.1.1 Study setting

Although the study setting of Strathclyde region includes about one third of Scotland's population it would be sensible to extend a study of this type to the whole of Scotland and this could probably be achieved without too much effort (especially drawing on the experience and expertise from this study). This would provide a source for a full Scottish investigation and supply a data set of use for many types of research in the road traffic crash field.

5.6.1.2 Data sources and linkage

As discussed above, extending the study would aid in an even fuller evaluation. ISD holds all Scotland's SMR01 data and linkage would not have to be restricted by council areas. The Strathclyde Safety Camera Partnership has an ongoing collaboration with the remainder of Scotland, hence retrieving information on other speed cameras would be possible. Retrieving STATS19 including postcodes from every police force in Scotland could prove more difficult. Overall, newer data could be included, which means that a higher level of accuracy of linkage can be expected as more postcodes are available (in STATS19).

It would be desirable to achieve a fuller analysis of thecost implications, even if just by attaching follow-up admissions to the unlinked SMR01 records. DfT costing methodology could be used further including the total estimates and fatal "willingness to pay" cost estimates.

Identifying control sites for the safety cameras would be helpful for a more detailed evaluation of camera effects. This could be achieved through using sites that were part of the selection process but did not qualify for a camera). The "control" sites could be matched in tome time with installation of the "real" safety camera sites (for before after analysis). These sites would provide a satisfactory control group, although not a substitute for a randomised design.

5.6.1.3 Analyses

Analytical approaches could follow more closely a weighting strategy method as developed by Hess & Polak [154] where RTC numbers at camera sites could be "detrended" and "deseasonalised".

Continuing analysis of what effect cameras have on different types of RTCs, injuries and costs, could create a basis for evaluating future widespread impact of safety cameras.

Geographical analysis of RTCs could be performed taking account of cost or other parameters of interest e.g. child casualties. Additionally, it would perhaps be useful to map where, in terms of deprivation, RTCs occur.

5.6.2 Suggestions for other research

There is a range of research that could complement the current study. Here are a few examples that would cover some of the current gaps in road traffic injury research.

5.6.2.1 Epidemiology of road traffic crashes in Strathclyde

Road traffic crashes will continue to be a major burden for the foreseeable future and even though we have come a long way in preventing RTC injuries in the developed world with a steady decline over time, RTCs will continue to increase rapidly in the developing world [1,15]. The RTC injuries investigated is usually only the more serious cases, but these are only a small proportion of total injuries and it would be of great use to add understanding of less serious injuries, as these also have a large impact on human suffering and cost to society.

It is of great importance to continue to support and pursue research into why and to whom RTC happens in order to develop and test appropriate countermeasures.

5.6.2.2 Accuracy of police reported casualties

In the light of the findings from this study there is an urgent need to investigate police judgement of injury severity over time, perhaps using a combination of quantitative and qualitative methods. Additionally, as it appears that the police are not informed of 45% of hospitalised RTC casualties. Research is indicated on to why this is and how it might be remedied. For example, determining the geographical location of the hospitalised casualties that are "missing" in STATS19 would greatly aid in both locating safety cameras and enhance the evaluation of these.

5.6.2.3 Impact of safety cameras

"Primary causation factors" for RTCs in STATS19 have recently been updated and an evaluation of these is required, with special attention to the reliability and usability of the speed related factors. If these turns out to be reliable they could be utilised in future studies of safety cameras.

There is little evidence about the efficacy of speed cameras compared to other types of RTC prevention but this would be useful in deciding on the deployment of preventative measures in the future.

5.6.2.4 Cost of road traffic crashes

The cost of road traffic crashes needs to be expanded to include slight injuries as well as serious because, even though hospitalised casualties have longer disability, the majority of the disability burden is carried by non-hospitalised casualties [74]. It would therefore be desirable to include primary care, hospital outpatient care and social services in national data collection, in a similar manner to the SMR01 and STATS19 recording. In a further step, surveys could be undertaken on sub-samples (e.g. RTC casualties from safety camera sites) in order to assess hitherto unmeasured additional costs to the individual.

5.7 Summary of Chapter 5

This chapter summarises the main findings, interpretation of results, strengths and weaknesses, implications, unanswered questions and suggestions for further research by research strand i.e. epidemiology of road traffic casualties in Strathclyde, accuracy of police reported road traffic crashes, epidemiological impact of safety cameras, cost impact of RTCs in Strathclyde and cost impact of RTCs at safety camera sites.

5.7.1 Epidemiology of road traffic casualties in Strathclyde

- <u>Main results</u>: Young people made up the largest proportion of total casualties, older casualties were more likely to die from an RTC, less protected road users suffered more severe injuries, head injuries were more common among pedestrians and pedal cyclists, while car occupants more often suffered injuries to the thorax and abdomen/lower back/lumbar spine.
- <u>Interpretation</u>: Overall, findings were in line with previous research. Older casualties were more likely to have a worse outcome of RTC because of frailty

while younger casualties were protected by their health. Less protected road users were more likely to suffer severe injuries.

- <u>Strengths and weaknesses</u>: The study had a long study period and a large sample, but the linkage may not have been completely robust hence some associations may have been missed.
- <u>Implications</u>: The results should inform policy makers that road traffic crashes continue to play a major role in disability and premature death.
- <u>Unanswered questions</u>: What are the RTC consequences of slight and serious injuries, including analysis of long-term effects (including analysis of primary care data)?
- <u>Suggestion for further research</u>: Adding understanding of less serious injuries would be desirable. Further extending the current study to include all of Scotland and up to date records would be feasible and useful.

5.7.2 Accuracy of police reported road traffic casualties

- <u>Main results</u>: About 45% of hospitalised RTC casualties were not reported in police records. Underreporting was mainly associated with no third party involvement and type of road user (especially pedal cyclists). Casualties coded as seriously injured in STATS19 declined in frequency more than the SMR01 RTC casualties.
- <u>Interpretation</u>: Underreporting of RTC casualties in police records is well known, as is the decline in serious injuries reported by police, and findings from this study support this. Additionally, findings show that there has been an increase in

casualties coded slight. Hence there may have been a shift in police judgement of severity over time (for alternative explanations see section 5.2.2 for details).

- <u>Strengths and weaknesses</u>: The study period and sample size were large and provided good statistical power in detecting linear trends etc.
- <u>Implications</u>: Re-assessing the current achievements against the national targets for 2010 (based on STATS19) is necessary. Additionally, the national safety camera programme also relies heavily on the completeness and accuracy of police data in order to site cameras at casualty "blackspots."
- <u>Unanswered questions</u>: Why would the police judgement of casualty severity change over time and what are the implications of police underreporting of severe casualties on the national targets for casualty reduction?
- <u>Suggestion for further research</u>: Investigation of police changing judgement of injury severity over time and why the police are not informed of 45% of hospitalised RTC casualties.

5.7.3 Epidemiological impact of safety cameras

- <u>Main results</u>: Safety cameras appeared to have contributed to a substantial reduction in the incidence of RTCs and especially of serious injury RTCs. Cameras appeared to have impacted most upon non-junction and multiple vehicle crashes, and the reduction were sustained over time.
- <u>Interpretation</u>: Findings were mostly in line with previous research with the addition of what type of crashes cameras are most effective in preventing and that effect appeared sustained over time.

- <u>Strengths and weaknesses</u>: The main weakness of this study was that it did not include a randomised controlled trial and a before and after study design was used instead. It included a relatively long time period (which should minimise the risk of regression towards the mean) and sample size and controlled the RTCs at camera sites against the remainder of Strathclyde region.
- <u>Implications</u>: Safety cameras should continue to play a major part in RTC prevention policies and expansion should be considered.
- <u>Unanswered questions</u>: What are the differences between speed cameras and other traffic calming measures in preventing casualties and costs (including cost-benefit analysis)? What is the characteristics of a speed related crash -are the primary causation factors reliable and can they be utilised in further research?
- <u>Suggestion for further research</u>: A randomised controlled trial of speed cameras and comparing speed cameras with other types of prevention is needed. Determining the geographical location of the hospitalised casualties that are "missing" in STATS19 would aid in both locating safety cameras and enhance the evaluation of these.

5.7.4 Cost impact of road traffic crashes in Strathclyde

<u>Main results</u>: The total estimated inpatient costs were (conservatively) £7.3 million yearly. The highest costs were incurred by head injuries and injuries to the lower extremities. Hip or thigh injuries incurred the highest mean costs. Pedestrian casualties on 30mph roads incurred a cost of £22 million (40% of total costs); of these £12 million was borne by casualties from the most deprived SIMD quintile

(20% of casualties), of which $\pounds 8$ million was incurred by the youngest and oldest age-groups.

- <u>Interpretation</u>: The results indicate the uneven burden of suffering and costs of RTCs with the most costly RTCs occur in areas with high levels of deprivation, a history of pedestrian RTCs and/or elderly and children, areas with many non-junction RTCs and 30 mph roads.
- <u>Strengths and weaknesses</u>: All cost estimates are likely to be extremely conservative: only costs incurred by acute hospital treatment were included; analysis was limited to include only the first 1.5 years admissions and unlinked SMR01 RTCs were not included as data on follow-up admissions were unavailable.
- <u>Implications</u>: In view of cost impact and size of different road casualty groups it would be desirable to evaluate current policies and practices in pedestrian traffic injury prevention, particularly in deprived areas with high elderly and child populations.
- <u>Unanswered questions</u>: Where are the sites/areas with the most costly RTCs and where are sites with the highest total costs (adding RTCs in vicinity)?
- <u>Suggestion of further research</u>: The cost of road traffic crashes needs to be substantiated with slight injuries as the largest disability burden is carried by nonhospitalised casualties. Including primary care, hospital outpatient care and social services in national data collection would be desirable.

5.7.5 Cost impact of road traffic crashes at safety camera sites

- <u>Main results</u>: Safety cameras in Strathclyde may so far have contributed to a saving of over £5 million and costs per day pre safety camera installation were over double that of the post safety camera installation (£8.8 and £3.9 respectively). Daily cost of non-junction crashes declined with 72%.
- <u>Interpretation</u>: The impact of speed in relation to injury severity is well documented (higher speeds = more severe injuries). As the daily costs declined with 72% for non-junction crashes (i.e. RTCs unaffected by slow speeds required at junctions) this may reflect that safety cameras result in lower speeds and fewer RTCs.
- <u>Strengths and weaknesses</u>: This part of cost analysis was limited to incorporate only the first 6 months admissions post-RTC, hence some acute costs were unaccounted for. The low frequency of hospitalised casualties and RTCs in this analysis may cause some random effects such as regression towards the mean. However, long term means were utilised, so the impact of RTM should be minimised.
- <u>Implications</u>: Safety cameras appear cost-effective and the safety camera partnership should utilise these findings further weighing camera costs against the calculated savings in a cost-benefit evaluation.
- <u>Unanswered questions</u>: What are the cost benefits of speed cameras, including full cost analysis of RTC consequences?
- <u>Suggestion of further research</u>: Including information on primary care, hospital outpatient care and social services through e.g. completing a survey of RTC population on camera sites in order to assess additional costs to the individual.

CHAPTER 6. CONCLUSIONS AND RECOMMENDATIONS

Contents of Chapter 6

6.1	CONCLUSIONS	
6.2	RECOMMENDATIONS	
6.2.1	Recommendations for scientists	
6.2.2	Recommendations for practitioners	
6.2.3	3 Recommendations for policy makers	
6.3	SUMMARY OF CHAPTER 6	

6.1 Conclusions

- Young people were the largest group of casualties although they had a low fatality rate. Severity of injury was progressively worse with advancing age and males were more likely to suffer a fatal injury compared to females.
- Unprotected road users were more likely to sustain severe injuries and head injuries were most commonly suffered by pedal cyclists and pedestrians. Motorcycle/moped occupants were most likely to have a fatal outcome.
- 3. There has been a general decline in road traffic crashes in Strathclyde region of approximately 18% over the years (comparing the first three years with the last three). Fatalities declined by 7%. Police fatal and serious injuries declined 40% while hospital injuries declined 21%.
- 4. Although police records indicate that serious injuries declined much more than hospital admissions, a general decline in the completeness of STATS19 is unlikely to have occurred. There appears, however, to have been an increasing tendency over time for police officers to report injuries as slight rather than serious.
- 5. 45% of hospitalised casualties were not recorded by police. Likelihood of RTC casualty not being recorded by police was associated with no third party involvement, higher age, earlier year of recording, road user (especially pedal cyclists and motorbike/moped occupants), shorter length of hospital stay and females.
- 6. The evaluation of safety cameras strongly suggests that they were effective in reducing both road casualty incidence and severity and that the reduction in incidence was sustained over time. Cameras thus fulfil an important public health,

as well as law enforcement, function and should continue to play a central role in traffic calming.

- Cameras appeared effective in reducing RTCs that were not near a junction and/or multiple vehicle RTCs, as well as serious and fatal injury RTCs.
- 8. The study indicates that the most costly RTCs occured in areas with high levels of deprivation, a history of pedestrian RTCs and/or elderly and children, areas with many non-junction RTCs and 30 mph roads. Pedestrian casualties occurring on 30mph roads made up about 40% of total hospital costs for the period (over £22 million); over half of this (£12 million) was incurred by (pedestrian) casualties from the most deprived SIMD quintile; two thirds of these costs (£8 million) were in the youngest and oldest age-groups.
- 9. Safety cameras in Strathclyde may so far have contributed to a saving of over £5 million and hospital admission costs per day (for surviving hospital casualties) Pre SCI were over double that of the Post SCI (£8.8 and £3.9 respectively).
- 10. Linking police and hospital RTC records provides a more comprehensive source for road traffic analysis, both in terms of evaluating time trends and national targets or investigating areas in need of remedial treatment, than any of the sources separately.
6.2 <u>Recommendations</u>

The recommendations are subdivided by professional area of interest i.e. scientists, practitioner and policy makers.

6.2.1 Recommendations for scientists

- 1. Epidemiological evaluations, similar to this study, with the addition of information from emergency departments, primary care, hospital outpatient care and social security, are required.
- 2. Further evaluation, probably using qualitative method of study, considering underlying reasons of police severity judgement, is urgently warranted.
- Evaluation of the relatively new recording of primary causation factors (in STATS19) is required, with special attention to the reliability and usability of the speed related factors.
- A cost-benefit evaluation using the findings from this study could be performed by the Safety Camera Partnership.
- 5. A randomised controlled trial of speed cameras or, as a second choice, a quasi-RCT is needed (see section 5.6.1.2).
- 6. Studies comparing speed cameras with other preventive measures is required, especially evaluations of different types of cameras in varying contexts.

6.2.2 Recommendations for practitioners

 It would be worthwhile to examine and revise current practices for updating police STATS19 records with hospital admission information.

- 2. Safety camera installation criteria based on speed and severity of RTCs appears to be effective, but adding further criteria (e.g. rates of non-junction and multiple vehicle RTCs on sites) may enhance the impact of cameras further.
- 3. It is advisable to evaluate current policies and practices in pedestrian traffic injury prevention, particularly in deprived areas with high elderly and child populations.
- 4. Using a fuller data set (as in this study) in detecting traffic injury black-spots could aid in identifying where, geographically, current gaps in road safety exists.

6.2.3 Recommendations for policy makers

- Continuing to battle road traffic crashes is of utmost importance and, although casualty rates in Scotland are comparatively low, policy makers should not be complacent. This should remain high priority in policy making.
- 2. It is of great importance that the achievements to date, towards the national targets for road traffic crash reduction for 2010, are re-examined in the light of this study (see section 5.2.2).
- 3. Supporting further research in this field is required. This study has provided a solid platform to continue building on (especially the linked data, see section 5.6.1)
- Routinely linking hospital and police data should be considered and adding geographical information to the hospital RTC records would greatly enhance research (see section 5.4).
- 5. An evaluation of expanding routinely collected RTC casualty information to include emergency departments and primary care should be done.

6.3 Summary of chapter 6

- RTC injury incidence in Strathclyde declined over the study period.
- Young and elderly people, as well as unprotected road users, carry a disproportionately great RTC injury burden.
- Many hospitalised RTC casualties were not recorded by police and there appears to have been an increasing tendency over time for police officers to report injuries as slight rather than serious. National (UK) statistics of RTCs should be interpreted with caution in the light of these findings and routinely linking police and hospital data would enhance the quality of RTC casualty statistics.
- The study indicates that the most costly RTCs occur in areas with high levels of deprivation, a history of pedestrian RTCs, elderly and child casualties, roads with many non-junction RTCs and 30 mph speed limits.
- Safety cameras appear effective in reducing both road casualty incidence and severity and the reduction in incidence is sustained over time. Additionally, safety cameras in Strathclyde may have contributed to a saving of over £5 million. Cameras thus fulfil an important public health, as well as law enforcement, function and should continue to play a central role in traffic calming. Studies comparing speed cameras with other preventive measures are required, as are randomised controlled trials where feasible.
- This study has demonstrated the value of utilising multiple data sources in the road traffic injury field.

REFERENCES

- (1) WHO. World report on road traffic injury prevention. Peden M, Scurfield R, Sleet D, Sleet D, Mohan D, Hyder AA et al., editors. Geneva: World Health Organisation, 2004.
- (2) Peden M, McGee K, Sharma G. The injury chart book: a graphical overview of the global burden of injuries. Geneva: World Health Organisation. 2002.
- (3) Jacobs G, Aeron-Thomas A, Astrop A. Estimating global road fatalities. TRL Report 445. Crowthorne: Transport Research Laboratory. 2000.
- (4) RoadPeace. World's first road death. RoadPeace, London, http://www.roadpeace.org/articles/worldfir.html (accessed 6-12-2004).
- (5) FEVR. Road traffic victims: Present situation and FEVR objectives. European Federation of Road Traffic Victims. FEVR. <u>http://www.fevr.org/english.html#</u> (accessed 16-11-2004).
- (6) Koornstra M, Lynam D, Nilsson G, Noordzij P, Pettersson H-E, Wegman F et al. SUNflower: A comparative study of the development of road safety in Sweden, the United Kingdom, and the Netherlands. SWOV Institute for Road Safety Research, The Netherlands, 2002. <u>http://www.swov.nl/en/publicaties/index.htm</u> (accessed 12-11-2004).
- (7) Odero W, Garner P, Zwi A, Zwi A. Road traffic injuries in developing countries: a comprehensive review of epidemiological studies. Tropical Medicine and International Health 1997; 2(5):445-460.
- (8) OECD. Safety of vulnerable road users (DSTI/DOT/RTR/RS7(98)1/FINAL). Organisation for Economic Co-operation and Development, 1998. <u>http://www.oecd.org/dataoecd/24/4/2103492.pdf</u> (accessed 15-11-2004).
- (9) Mohan D. Traffic Safety and Health in Indian Cities. Journal of Transport and Infrastructure 2002; 9(1):79-94.
- (10) Winston FK, Elliott MR, Chen EG, Chen EG, Simpson EM, Durbin DR. Acute healthcare utilization by children after motor vehicle crashes. Accident Analysis and Prevention 2004; 36(4):507-511.
- (11) Haukeland JV. Welfare consequences of injuries due to traffic accidents. Accident Analysis & Prevention 1996; 28(1):63-72.
- (12) de Vries APJ, Kassam-Adams N, Cnaan A, Cnaan A, Sherman-Slate E, Gallagher PR et al. Looking Beyond the Physical Injury: Posttraumatic Stress Disorder in Children and Parents After Pediatric Traffic Injury. Pediatrics 1999; 104(6):1293-1299.
- (13) Stallard P, Velleman R, Langsford J, Langsford J, Baldwin S. Coping and psychological distress in children involved in road traffic accidents. British Journal of Clinical Psychology 2001; 40:197-208.:197-208.
- (14) Mayou R, Bryant B. Outcome in consecutive emergency department attenders following a road traffic accident. British Journal of Psychiatry 2001; 179:528-534.:528-534.
- (15) Kopits E, Cropper M. Traffic fatalities and economic growth. Policy Research Working Paper No. 3035. Washington DC, World Bank. 2003.
- (16) Scottish Executive. Road Accidents Scotland 2006. Edinburgh: Scottish Excecutive, National Statistics. 2007.
- (17) Baker SP, O'Neill B, Ginsburg MJ, Li G. The injury fact book, second edition. Oxford University Press, 1992.

- (18) Rosman DL, Ferrante AM, Marom Y, Marom Y. A linkage study of Western Australian drink driving arrests and road crash records. Accident Analysis & Prevention 2001; 33(2):211-220.
- (19) Scottish Executive. Road Accidents Scotland 1999. Scottish Excecutive, National.Statistics. 2001.
- (20) Abdel-Aty MA, Abdelwahab HT. Exploring the relationship between alcohol and the driver characteristics in motor vehicle accidents. Accident Analysis & Prevention 2000; 32(4):473-482.
- (21) Flisher AJ, Ziervogel CF, Chalton DO, Leger PH, Robertson BA. Risk-taking behaviour of Cape Peninsula high-school students. Part VI. Road-related behaviour. S Afr Med J 1993; 83(7):486-490.
- (22) Finn P, Bragg BW. Perception of the risk of an accident by young and older drivers. Accid Anal Prev 1986; 18(4):289-298.
- (23) Laapotti S, Keskinen E. Has the difference in accident patterns between male and female drivers changed between 1984 and 2000? Accident Analysis & Prevention 2004; 36(4):577-584.
- (24) Hemenway D, Solnick SJ. Fuzzy dice, dream cars, and indecent gestures: correlates of driver behavior? Accident Analysis & Prevention 1993; 25(2):161-170.
- (25) Ostrom M, Eriksson A. Single-vehicle crashes and alcohol: a retrospective study of passenger car fatalities in Northern Sweden. Accident Analysis & Prevention 1993; 25(2):171-176.
- (26) Massie DL, Campbell KL, Williams AF. Traffic Accident involvement rates by driver age and gender. Accident Analysis & Prevention 1995; 27(1):73-87.
- (27) Li G, Baker SP, Langlois JA, Kelen GD. Are female drivers safer? An application of the decomposition method. Epidemiology 1998; 9(4):379-384.
- (28) Tavris DR, Kuhn EM, Layde PM. Age and gender patterns in motor vehicle crash injuries: importance of type of crash and occupant role. Accid Anal Prev 2001; 33(2):167-172.
- (29) Scottish.Excecutive. Road Accidents Scotland 1999. Scottish Excecutive, National Statistics. 2001.
- (30) ISD. Unintentional Injuries. Scottish Health Statistics (ISD website), <u>http://www.isdscotland.org/isd/info3.jsp?pContentID=2140&p_applic=CCC&p_service=C_ontent.show&</u> (accessed 16-11-2004).
- (31) Pearson J, Jeffrey S, Stone D. Varying gender pattern of childhood injury mortality over time in Scotland. (Epub ahead of print) Arch Dis Child 2008.
- (32) Eberhard J. Older drivers' "high per-mile crash involvement": the implications for licensing authorities. Traffic Inj Prev 2008; 9(4):284-290.
- (33) Hakamies-Blomqvist L. Ageing Europe: the challenges and opportunities for transport safety. Brussels: European Transport Safety Council. 2003.
- (34) New developments in retrospective data bases car crashes and car collisions with pedestrians and motorcyclists. -The research activities of the German insurers.: 1998.
- (35) DiGuiseppi C, Li L, Roberts I. Influence of travel patterns on mortality from injury among teenagers in England and Wales, 1985-95: trend analysis. BMJ 1998; 316:904-905.
- (36) Thompson R, Choonara I, Hewitt S, Holt M. Age and sex of drivers associated with child pedestrian injuries. J Child Health Care 2003; 7(3):184-190.

- (37) Moskowitz H. Marihuana and driving. Accident Analysis & Prevention 1985; 17(4):323-345.
- (38) Ellinwood J, Heatherly DG. Benzodiazepines, the popular minor tranquilizers: Dynamics of effect on driving skills. Accident Analysis & Prevention 1985; 17(4):283-290.
- (39) Compton RP, Blomberg RD, Moskowitz H, Burns M, Peck RC, Fiorentino D. Crash Risk of Alcohol Impaired Driving. Mayhew D R, Dussault C, editors. Proceedings of the 16th knternational conference on alcohol, drugs and traffic safety, Montreal. <u>http://www.saaq.gouv.qc.ca/t2002/actes/pdf/(06a).pdf</u> (accessed 25-1-2005).
- (40) Zador PL. Alcohol-related relative risk of fatal driver injuries in relation to driver age and sex. J Stud Alcohol 1991; 52(4):302-310.
- (41) Mao Y, Zhang J, Robbins G, Clarke K, Lam M, Pickett W. Factors affecting the severity of motor vehicle traffic crashes involving young drivers in Ontario. Inj Prev 1997; 3(3):183-189.
- (42) Koornstra M, Lynam D, Nilsson G, Nilsson G, Noordzij P, Pettersson H et al. SUNflower: A comparative study of the development of road safety in Sweden, the United Kingdom, and the Netherlands. SWOV Institute for Road Safety Research, The Netherlands. 2002. <u>http://www.swov.nl/en/publicaties/index.htm</u> (accessed 12-11-2004).
- (43) Keigan M, Tunbridge RJ. The incidence of alcohol in fatally injured adult pedestrians. Jackson P, editor. TRL report 579. Transport Research Laboratory. 2003.
- (44) Holubowycz OT. Age, sex, and blood alcohol concentration of killed and injured pedestrians. Accident Analysis & Prevention 1995; 27(3):417-422.
- (45) Ryb GE, Dischinger PC, Kufera JA, Soderstrom CA. Social, behavioral and driving characteristics of injured pedestrians: a comparison with other unintentional trauma patients. Accid Anal Prev 2007; 39(2):313-318.
- (46) Öström M, Eriksson A. Pedestrian fatalities and alcohol. Accident Analysis & Prevention 2001; 33(2):173-180.
- (47) Millar C. Alcohol and the Pedestrian Road Casualty. Scottish Executive Central Research Unit, Development Department Research Programme Research Findings 1998;49
- (48) Abargel A, Sharabi Y, Scope A, Scope A, Hanin A, Eldad A. Do Accidents Happen Accidentally? A Study of Trauma Registry and Periodical Examination Database. Journal of Trauma-Injury Infection & Critical Care 2001; 50(1):20-23.
- (49) Ryb GE, Dischinger P, Kufera J, Soderstrom C. Smoking is a marker of risky behaviors independent of substance abuse in injured drivers. Traffic Inj Prev 2007; 8(3):248-252.
- (50) Global Road Safety Partnership. Speed management: a road safety manual for decisionmakers and practitioners. Geneva: WHO. 2008.
- (51) Taylor M. Managing Vehicle Speeds for Safety: Why? And How? Transport Research Laboratory 2002.
- (52) Taylor M, Lynam D, Baum A. The effects of drivers' speed on the frequency of road accidents. 421. Transport Research Laboratory. 2000.
- (53) Can't go, won't go. In Procs. Speed: Whose business is it?. PACTS, 1999.
- (54) Finch DJ, Kompfner P, Lockwood CR, Maycock G. Speed, speed limits and accidents.58. Transport Research Laboratory. 1994.
- (55) Campbell M, Stradling S. Strathclyde drivers, summary report. Napier University. 2003.
- (56) Essex Partnership. Safety Camera Research, Offenders Home Addresses, Summary Report. The Essex Partnership. 2001.

- (57) Engstrom K, Diderichsen F, Laflamme L. Socioeconomic differences in injury risks in childhood and adolescence: a nation-wide study of intentional and unintentional injuries in Sweden. Inj Prev 2002; 8(2):137-142.
- (58) Roberts I. Cause specific social class mortality differentials for child injury and poisoning in England and Wales. J Epidemiol Community Health 1997; 51(3):334-335.
- (59) Hasselberg M, Laflamme L, Ringbäck WG. Socioeconomic differences in road traffic injuries during childhood and youth: a closer look at different kinds of road user. Journal of Epidemiology & Community Health 2001; 55(12):858-862.
- (60) Hasselberg M, Laflamme L. Road traffic injuries among young car drivers by country of origin and socioeconomic position. Int J Public Health 2008; 53(1):40-45.
- (61) Abdalla IM, Raeside R, Barker D, Barker D, McGuigan DRD. An investigation into the relationships between area social characteristics and road accident casualties. Accident Analysis & Prevention 1997; 29(5):583-593.
- (62) Hasselberg M, Vaez M, Laflamme L. Socioeconomic aspects of the circumstances and consequences of car crashes among young adults. Soc Sci Med 2005; 60(2):287-295.
- (63) Ryb G, Dischinger P, Smith G, Soderstrom C. Adult Correlates of Early Behavioral Maladjustment: A Study of Injured Drivers. Ann Adv Automotive Med 2008; 52:39-48.
- (64) Dumais A, Lesage AD, Boyer R, Lalovic A, Chawky N, Menard-Buteau C et al. Psychiatric risk factors for motor vehicle fatalities in young men. Can J Psychiatry 2005; 50(13):838-844.
- (65) Gulliver P, Begg D. Personality factors as predictors of persistent risky driving behavior and crash involvement among young adults. Inj Prev 2007; 13(6):376-381.
- (66) Lajunen T, Sumer N, Ozkan T. Do symptoms of psychopathology predict a professional driver's involvement in traffic accidents? J Traffic Med 2001; 29(1-2):32-35.
- (67) Gorrie CA, Rodriguez M, Sachdev P, Duflou J, Waite PM. Increased neurofibrillary tangles in the brains of older pedestrians killed in traffic accidents. Dement Geriatr Cogn Disord 2006; 22(1):20-26.
- (68) Gorrie CA, Brown J, Waite PM. Crash characteristics of older pedestrian fatalities: dementia pathology may be related to 'at risk' traffic situations. Accid Anal Prev 2008; 40(3):912-919.
- (69) Schanke AK, Rike PO, Molmen A, Osten PE. Driving behaviour after brain injury: a follow-up of accident rate and driving patterns 6-9 years post-injury. J Rehabil Med 2008; 40(9):733-736.
- (70) Håland Y, Lövsund P, Nygren Å, Nygren Å. Life-threatening and disabling injuries in carto-car side impacts—implications for development of protective systems. Accident Analysis and Prevention 1993; 25(2):199-205.
- (71) Norin H, Krafft M, Korner J, Korner J, Nygren Å, Tingvall C. Injury severity assessment for car occupants in frontal impacts, using disability scaling. Journal of Clinical Epidemiology 1997; 50(1):95-103.
- (72) Bylund P-O, Björnstig U. Sick Leave and Disability Pension Among Passenger Car Occupants Injured in Urban Traffic. Spine 1998; 23(9):1023-1028.
- (73) Malm S, Krafft M, Kullgren A, Ydenius A, Tingvall C. Risk of Permanent Medical Impairment (RPMI) in Road Traffic Accidents. Ann Adv Automotive Med 2008; 52:93-100.
- (74) Waller JA, Skelly JM, Davis JH, Herreid TW. Characteristics, hospital charges, and effects of road transportation injuries in Vermont. Accident Analysis & Prevention 1994; 26(5):635-645.

- (75) Brookes M, MacMillan R, Cully S, Anderson E, Murray S, Mendelow AD et al. Head injuries in accident and emergency departments. How different are children from adults? J Epidemiol Community Health 1990; 44(2):147-151.
- (76) Wittenberg MD, Sloan JP, Barlow IF. Head injuries in Leeds: changes in epidemiology and survival over 12 years. Emerg Med J 2004; 21(4):429-432.
- (77) Stuss DT, Ely P, Hugenholtz H, Richard MT, LaRochelle S, Poirier CA et al. Subtle neuropsychological deficits in patients with good recovery after closed head injury. Neurosurgery 1985; 17(1):41-47.
- (78) WHO. The ICD-10 classification of mental and behavioural disorders. Diagnostic criteria for research. Peden M, Scurfield R, Sleet D, Sleet D, Mohan D, Hyder AA et al., editors. Geneva: World Health Organisation. 1993.
- (79) Youngjohn J. Post-concussion syndrome after road and other accidents. Brain damage, somatoform disorder, or litigation induced? In: Mitchell M, editor. The aftermath of road accidents, Psychological, social and legal consequences of an everyday trauma. London and New York: Routledge, 2001: 99-109.
- (80) Berglund A, Alfredsson L, Jensen I, Jensen I, Cassidy JD, Nygren Å. The association between exposure to a rear-end collision and future health complaints. Journal of Clinical Epidemiology 2001; 54(8):851-856.
- (81) Mayou R, Bryant B. Outcome 3 years after a road traffic accident. Psychological Medicine 2002; 32:671-675.:671-675.
- (82) Singleton M, Qin H, Luan J. Factors associated with higher levels of injury severity in occupants of motor vehicles that were severely damaged in traffic crashes in Kentucky, 2000-2001. Traffic Inj Prev 2004; 5(2):144-150.
- (83) Bédard M, Guyatt GH, Stones MJ, Stones MJ, Hirdes JP. The independent contribution of driver, crash, and vehicle characteristics to driver fatalities. Accident Analysis & Prevention 2002; 34(6):717-727.
- (84) Evans L. Female compared with male fatality risk from similar physical impacts. J Trauma 2001; 50(2):281-288.
- (85) VTI. Kvinnlig krockdocka kan ge förbättrat skydd i trafiken. <u>http://www.vti.se/templates/Page 5026.aspx</u> (accessed 10-10-2008).
- (86) Waller PF, Stewart JR, Hansen AR, Stutts JC, Popkin CL, Rodgman EA. The potentiating effects of alcohol on driver injury. JAMA 1986; 256(11):1461-1466.
- (87) Mayou R, Bryant B. Consequences of road traffic accidents for different types of road user. Injury 2003; 34(3):197-202.
- (88) Markogiannakis H, Sanidas E, Messaris E, Koutentakis D, Alpantaki K, Kafetzakis A et al. Motor vehicle trauma: analysis of injury profiles by road-user category. Emerg Med J 2006; 23(1):27-31.
- (89) Koornstra M, Broughton J, Esberger R, Glansdorp C, Köppel W, Taylor F et al. Transport safety performance in the EU - a statistical overview. European Transport Safety Council, <u>http://www.etsc.be/documents/statoverv.pdf</u> (accessed 18-11-2004).
- (90) Lane PL, McClafferty KJ, Nowak ES. Pedestrians in real world collisions. J Trauma 1994; 36(2):231-236.
- (91) Hill DA, Delaney LM, Duflou J, Duflou J. A Population-Based Study of Outcome after Injury to Car Occupants and to Pedestrians. Journal of Trauma 1996; 40(3):351-355.
- (92) Hobbs CA, Mills PJ. Injury probability for car occupants in frontal and side impacts. 1124. Transport Research Laboratory. 1984.

- (93) Ashton SJ. Pedestrian injuries: the influence of vehicle design. In: Foot HC, et al., editors. Road Safety Research and Practice. Praeger, 1981.
- (94) RoSPA. Speed Cameras. Report, RoSPA. 2001.
- (95) Shafi S, Gentilello L. A nationwide speed limit < or = 65 miles per hour will save thousands of lives. Am J Surg 2007; 193(6):719-722.
- (96) Mock CN, Grossman DC, Kaufman RP, Mack CD, Rivara FP. The relationship between body weight and risk of death and serious injury in motor vehicle crashes. Accid Anal Prev 2002; 34(2):221-228.
- (97) Mayou R, Bryant B, Duthie R, Duthie R. Psychatric consequences of road traffic accidents. British Medical Journal 1993; 307(6905):647-651.
- (98) Horne DJdL. Treatment of pain, fear and loss following a road accident. A case study. In: Mitchell M, editor. The aftermath of road accidents, Psychological, social and legal consequences of an everyday trauma. London and New York: Routledge, 2005: 188-198.
- (99) Mayou R, Bryant B, Duthie R. Psychiatric consequences of road traffic accidents. BMJ 1993; 307(6905):647-651.
- (100) Murray J, Ehlers A, Mayou RA, Mayou RA. Dissociation and post-traumatic stress disorder: two prospective studies of road traffic accident survivors. British Journal of Psychiatry 2002; 180:363-368.
- (101) Kuch K, Swinson RP, Kirby M. Post-traumatic stress disorder after car accidents. Can J Psychiatry 1985; 30(6):426-427.
- (102) Blanchard EB, Hickling EJ, Taylor AE, Loos W. Psychiatric morbidity associated with motor vehicle accidents. J Nerv Ment Dis 1995; 183(8):495-504.
- (103) Ward H, Lyons R, Thoreau R. Under-reporting of road casualties Phase 1. Road safety research report no. 69. Department for Transport, London. 2006.
- (104) Hvoslef H. Under-Reporting of road traffic accidents recorded by the police, at the international level. OECD-IRTAD and Norwegian Public Roads Administration. 1994.
- (105) Elvik R, Mysen AB. Incomplete Accident Reporting: Meta-Analysis of Studies Made in 13 Countries. Transportation Research Record 1999; 1665:133-140.
- (106) Amoros E, Martin JL, Laumon B. Under-reporting of road crash casualties in France. Accid Anal Prev 2006; 38(4):627-635.
- (107) Alsop J, Langley J. Under-reporting of motor vehicle traffic crash victims in New Zealand. Accid Anal Prev 2001; 33(3):353-359.
- (108) Cryer PC, Westrup S, Cook AC, Cook AC, Ashwell V, Bridger P et al. Investigation of bias after data linkage of hospital admissions data to police road traffic crash reports. Injury Prevention 2001; 7(3):234-241.
- (109) Lujic S, Finch C, Boufous S, Hayen A, Dunsmuir W. How comparable are road traffic crash cases in hospital admissions data and police records? An examination of data linkage rates. Aust N Z J Public Health 2008; 32(1):28-33.
- (110) Lyons RA, Ward H, Brunt H, Macey S, Thoreau R, Bodger OG et al. Using multiple datasets to understand trends in serious road traffic casualties. Accid Anal Prev 2008; 40(4):1406-1410.
- (111) Petridou ET, Yannis G, Terzidis A, Dessypris N, Germeni E, Evgenikos P et al. Linking emergency medical department and road traffic police casualty data: a tool in assessing the burden of injuries in less resourced countries. Traffic Inj Prev 2009; 10(1):37-43.

- (112) Farmer CM. Reliability of police-reported information for determining crash and injury severity. Traffic Inj Prev 2003; 4(1):38-44.
- (113) Agran PF, Castillo DN, Winn DG. Limitations of data compiled from police reports on pediatric pedestrian and bicycle motor vehicle events. Accid Anal Prev 1990; 22(4):361-370.
- (114) Gill M, Goldacre MJ, Yeates DG. Changes in safety on England's roads: analysis of hospital statistics. BMJ 2006; 333(7558):73.
- (115) Langley JD, Dow N, Stephenson S, Kypri K. Missing cyclists. Inj Prev 2003; 9(4):376-379.
- (116) Loo BP, Tsui KL. Factors affecting the likelihood of reporting road crashes resulting in medical treatment to the police. Inj Prev 2007; 13(3):186-189.
- (117) Rosman DL, Knuiman MW. A comparison of hospital and police road injury data. Accid Anal Prev 1994; 26(2):215-222.
- (118) Simpson HF. Comparison of Hospital and Police Casualty Data: A National Study. TRL Report No. 173. 1996. Crowthorne, Transport Research Laboratory.
- (119) Bartsch AJ, Gilbertson LG, Prakash V, Morr DR, Wiechel JF. Minor crashes and 'whiplash' in the United States. Ann Adv Automotive Med 2008; 52:117-128.
- (120) Department for Transport. Road accidents casualties: a comparison of STATS19 data with Hospital Episode Statistics. (Related Document to: Under-reporting of road casualties: Phase 1). Department for Transport. <u>http://www.dft.gov.uk/stellent/groups/dft_rdsafety/documents/page/dft_rdsafety_611756.</u> <u>pdf . 23-6-2006</u> (accessed 1-12-0006)
- (121) Department for Transport. Actions on recommendations made in two DfT commissioned reports. Department for Transport. <u>http://www.dft.gov.uk/foi/responses/2007/foisept2007/foitwocomissionedreports/pdffoitwo comreps.pdf</u> (accessed 10-10-2007).
- (122) European Transport Safety Council. EU transport accident, incident and casualty databases: -current status and future needs. Brussels: European Transport Safety Council. <u>http://www.etsc.be/oldsite/database.pdf</u> (accessed 1-12-2006)
- (123) De Mol J, Boets S. Optimisation of traffic accident statistics, Interrim report. Brussels: Belgian Public Planning Service Science Policy. 2003.
- (124) Vägverket. STRADA (Swedish Traffic Accident Data Acquisition). http://www.vv.se/templates/page3____12863.aspx (accessed 5-3-2008).
- (125) PENDANT. Pan-European Co-ordinated Accident and Injury Database (PENDANT): Deliverable D7 WP3 Analysis Protocol. Graz University of Technilogy Vehicle Safety Institute. <u>http://www.vsi.tugraz.at/pendant/</u> (accessed 15-12-2004).
- (126) Barchielli A, Balzi D. [Road traffic accidents in an urban area: linkage between municipal police data-base and routinely collected medical data to assess adverse health effects, health system activities and costs]. Epidemiol Prev 2007; 31(4):218-224.
- (127) Rosman DL. The western australian road injury database (1987-1996): ten years of linked police, hospital and death records of road crashes and injuries. Accid Anal Prev 2001; 33(1):81-88.
- (128) Ward H, Robertson S, Lester T, Pedler A. Reporting of Road Traffic Accidents in London: Matching Police STATS19 Data with Hospital Accident and Emergency Department Data. Project report PR/T/069/2002. 2002.
- (129) Broughton J, Keigan M, James FJ. Linkage of hospital trauma data and road accident data. TRL Report 518. Crowthorne: Transport Research Laboratory. 2001.

- (130) Simpson HF. National hospital study of road accident casualties. TRL Report No. 272. Crowthorne: Transport Research Laboratory. 1997.
- (131) Hopkin JM, Murray PA, Pitcher M, Galasko CSB. Police and Hospital Recording of Nonfatal Road Accident Casualties: A Study in Greater Manchester. TRL Report No. 379. Crowthorne: Transportation Research Laboratory. 1993.
- (132) Keigan M, Broughton J, Tunbridge RJ. Linkage of STATS19 and Scottish hospital inpatient data - analyses for 1980-1995. TRL report No. 420. Crowthorne: Transport Research Laboratory. 1999.
- (133) EU transport accident, incident and casualty databases: -current status and future needs. Brussels: European Transport Safety Council. <u>http://www.etsc.be/oldsite/database.pdf</u> (accessed 1-12-2006).
- (134) Department for Transport. Highways economics note. 1. Department for Transport. 2004.
- (135) Gurses D, Sarioglu-Buke A, Baskan M, Kilic I. Cost factors in pediatric trauma. Can J Surg 2003; 46(6):441-445.
- (136) Hendrie D, Rosman DL, Harris AH. Hospital inpatient costs resulting from road crashes in Western Australia. Aust J Public Health 1994; 18(4):380-388.
- (137) Miller TR, Pindus NM, Douglass JB. Medically related motor vehicle injury costs by body region and severity. J Trauma 1993; 34(2):270-275.
- (138) Langley JD, Phillips D, Marshall SW. Inpatient costs of injury due to motor vehicle traffic crashes in New Zealand. Accid Anal Prev 1993; 25(5):585-592.
- (139) Cummins G. The history of road safety. Drive and stay alive website, <u>http://www.driveandstayalive.com/info%20section/history/history.htm</u> (accessed 29-6-2005).
- (140) Rutherford WH, Greefield T, Hayes HRM, Nelson JK, Department of Health and Social Security.Office of the Chief Scientist. Medical effects of seat belt legislation in the United Kingdom: research report number 13. HMSO. 1985.
- (141) Halcrow F. Accidents at Signal Controlled Junctions in Glasgow. The Scottish Office Central Research Unit. Development Department Research Programme Research Findings No. 23. 1996.
- (142) Diamantopoulou K, Cameron M, Shtifelman M, Shtifelman M. Evaluation of Moving Mode Radar for Speed Enforcement in Victoria, 1995-1997. Report no. 141. Monash University Accident Research Centre. 1998.
- (143) Diamantopoulou K, Cameron M. An Evaluation of the Effectiveness of Overt and Covert Speed Enforcement Achieved Through Mobile Radar Operations. Report no. 187. Monash University Accident Research.Centre. 2002.
- (144) Newstead S, Narayan S. Evaluation of the crash effects of the changes in speed zones in Victoria implemented during 1992-93 (excluding 100 to 110 km/h): update including 1990-1997 crash data. Monash.University.Accident.Research.Centre. Report no. 136. 1998.
- (145) DETR. New Directions in Speed Management A Review of Policy. DETR. 2000.
- (146) Mackie A. Urban speed management methods. TREL report No. 363. Crowthorne: Transport Research Laboratory. 1998.
- (147) DETR. New Directions in Speed Management A Review of Policy. DETR. 2005.
- (148) van Driel CJG, Davidse RJ, van Maarseveen MFAM, van Maarseveen MFAM. The effects of an edgeline on speed and lateral position: a meta-analysis. Accident Analysis and Prevention 2004; 36(4):671-682.

- (149) Commandeur J, Mathijssen R, Elvik R, Janssen W, Kallberg V-P. Scenarios for the implementation of daytime running lights in the European Union (R-2003-29). SWOV Institute for Road Safety Research. The Netherlands. <u>http://www.swov.nl/en/publicaties/index.htm</u> (accessed 12-11-2004).
- (150) Newstead S, Cameron M, Gantzer S, Gantzer S, Vulcan P. Modeling of Some Major Factors Influencing Road Trauma Trends in Victoria 1989-93. Report no. 74. Monash University Accident Research Centre. 1995.
- (151) Newstead S, Cameron M, Narayan S, Narayan S. Further modeling of some major factors influencing road trauma trends in Victoria: 1990-96. Report no. 129. Monash University Accident Research Centre. 1998.
- (152) Kronberg H, Nilsson G. Automatisk hastighetsövervakning. Hastighetsförlopp före och efter montering av kameraskåp på E4 mellan Iggesund och Hudiksvall. Swedish Road and Transport Research Institute. VTI meddelande 906. 2000.
- (153) Keall MD, Povey LJ, Frith WJ, Frith WJ. The relative effectiveness of a hidden versus a visible speed camera programme. Accident Analysis and Prevention 2001; 33(2):277-284.
- (154) Hess S, Polak JW. An Analysis of the Effects of Speed Limit Enforcement Cameras on Accident Rates. Transportation Research Record 2003; 180:5-33.
- (155) Hess S. An Analysis of the Effects of Speed Limit Enforcement Cameras with Differentiation by Road Type and Catchment Area. Transportation Research Record 2004; 1830:25-34.
- (156) Christie SM, Lyons RA, Dunstan FD, Jones SJ. Are mobile speed cameras effective? A controlled before and after study. Inj Prev 2003; 9(4):302-306.
- (157) Perez K, Mari-Dell'Olmo M, Tobias A, Borrell C. Reducing road traffic injuries: effectiveness of speed cameras in an urban setting. Am J Public Health 2007; 97(9):1632-1637.
- (158) Highways Agency. West London Speed Camera Demonstration Project. London Accident Analysis Unit. 1997.
- (159) Chen G, Wilson J, Meckle W, Cooper P. Evaluation of photo radar program in British Columbia. Accid Anal Prev 2000; 32(4):517-526.
- (160) Gains A, Heydecker B, Shrewsbury J, Robertson S. The National Safety Camera Programme Three-year Evaluation Report. University College London and PA Consulting Group. 2004.
- (161) Pilkington P, Kinra S. Effectiveness of speed cameras in preventing road traffic collisions and related casualties: systematic review. BMJ 2005; 330(7487):331-334.
- (162) Victoria Police. Traffic Camera Office Information Update. Victoria Police , Traffic Camera Office , Melbourne , Australia 1997
- (163) Re-Investigation of the Effectiveness of the Victorian TAC's Road Safety Campaigns.: 2000.
- (164) Cameron MH, Newstead S, V. Response by Monash University Accident Research Centre to "Re-investigation of the effectiveness of the Victorian Transport Accident Commission's road safety campaigns" (White, Walker, Glonek and Burns, November 2000). Report no. 177. Monash University Accident Research Centre. 2000.
- (165) Elvik R. Effects of accidents of automatic speed enforcement in Norway. Transportation Research Record 1997; 1595:14-19.
- (166) Hooke A, Knox J, Portas D, Portas D. Cost Benefit Analysis of Traffic Light and Speed Cameras. Police Research Series Paper , Police Research Group, Home Office 1996; 20.

- (167) DETR. Tomorrows roads -safer for everyone. 2000. DETR.
- (168) P A consulting group. ost recovery system for speed and red light camera enforcement. Handbook for national rollout -Scotland. 2002. Scottish Executive, Justice Department.
- (169) Campbell M, Stradling S. The General Detterent Effect of Speed Camera Housing. Napier.University. 2001.
- (170) National Statistics. Road Accidents Statistics (STATS19 Returns). National Statistics . 8-8-2003. National Statistics. 12-4-2005.
- (171) Clinical indicators support team. The raw material for the Scottish Clinical Indicators: SMR data. NHS Scotland, SHOW website, Clinical indicators support team. 2005. accessed 13-2-2009.
- (172) ISD Scotland. SMR Data Manual. 2000. ISD Scotland.
- (173) McLoone P. Carstairs Scores for Scottish Postcode, Sectors from the 1991 Census. Public Health Research Unit, University of Glasgow 2000;
- (174) ISD. Deprivation, ISD recommendations for analysing health data. Scottish Health Statistics (ISD website), http://www.isdscotland.org/isd/info3.jsp?pContentID=3211&p_applic=CCC&p_s ervice=Content.show& . accessed 18-4-2005.
- (175) Scottish Executive. Scottish Index of Multiple Deprivation 2004: Summary technical report. 2004. Edinburgh, Scottish Executive.
- (176) Scottish Neighbourhood Statistics. Area data guide, index of deprivation. Scottish Neighbourhood Statistics: http://www.sns.gov.uk/guide/guide-reports.asp . 2005. accessed 18-4-2005.
- (177) PA consulting group. Cost recovery system for speed and red light camera enforcement. Handbook for national rollout -Scotland. Scottish.Executive.Justice.Department. 2002.
- (178) WHO & DIMDI (German Institute of Medical Documentation and Information). International Statistical Classification of Diseases and Related Health Problems, 10th Revision. http://www.who.int/classifications/apps/icd/icd10online/ . 2007. WHO. Accessed 20-8-2007.
- (179) Fellegi IP, Sunter AB. A Theory for Record Linkage. Journal of the American Statistical Association 1969;(64):1183-1210.
- (180) Kendrick SW, Clarke JA. The Scottish Medical Record Linkage System. Health Bulletin (Edinburgh) 1993;(51):72-79.
- (181) Razzak JA, Luby SP. Estimating deaths and injuries due to road traffic accidents in Karachi, Pakistan, through the capture-recapture method. International Journal of Epidemiology 1998; 27:866-871.:866-871.
- (182) Morrison A. Traffic related injuries among young people in Scotland -an epidemiological perspective. 2002. (thesis)
- (183) Walsh D, Smalls M, Boyd J. Electronic Health Summaries Building on the Foundation of Scottish Record Linkage System. Patel Vea, editor. 1212-1216. 2001. Amsterdam, IOS Press.
- (184) Department for Transport. Instructions for the Completion of Road Accident Reports. 2004. Department for Transport.
- (185) ISD Scotland. Scottish Health Service Costs (Costs Book) 2005/2006. http://www.isdscotland.org/isd/5201.html . 2-5-2007. ISD Scotland.

- (186) Department for Transport. Road accident statistics in Great Britain: Levels of reporting. http://www.dft.gov.uk/pgr/statistics/datatablespublications/accidents/roadaccident tstatisticsingrea1835. 2006. Department for Transport. Accessed 3-7-2007.
- (187) Blatchford O. Rise has been real in Glasgow (response to article Trends in emergency admissions). BMJ 1999; 30(319):1201.
- (188) ISD Scotland. Data Quality Assurance. http://www.isdscotland.org/isd/mdq-dataquality-services.jsp?pContentID=2700&p_applic=CCC&p_service=Content.show& ISD Scotland. Accessed 11-9-2007.
- (189) Christie SM, Lyons RA, Dunstan FD, Dunstan FD, Jones SJ. Are mobile speed cameras effective? A controlled before and after study. Injury Prevention 2003; 9(4):302-306.
- (190) EU transport accident, incident and casualty databases: -current status and future needs. http://www.etsc.be/oldsite/database.pdf. 2001. European Transport Safety Council, Brussels. Accessed 1-12-2006.
- (191) Department for Transport. Road Casualties Great Britain: 2006. Accessed 27-9-2007. Transport Statistics.

APPENDIX

Contents of appendix

I	Published paper	194
II.	Stats19 data guide	201
III.	Scottish Morbidity Records 01 and General records office Scotland mortality codes	204
IV.	ICD 10 codes	207
۷.	The Scottish record linkage system	208
VI.	Follow-up diagnosis significantly related to a 1st single injury	214
VII.	Weighted mean cost by speciality	219

I. Published paper

An evaluation of police reporting of road casualties.

Jeffrey S, Stone DH, Blamey A, Clark D, Cooper C, Dickson K, Mackenzie M, Major K.

Inj Prev. 2009 Feb;15(1):13-8.

Data Guide

The Department of the Environment, Transport and the Regions (DETR) Road Accident Data - GB Variables and Values and Export Record Layouts December 1999

Record Layouts

Accident Records Variable Character		Integer	Variable Label		
	POSIDOII	/ лірпа			
ACCYR	1 - 4	(\mathbf{I})	Accident Year (YYYY)		
ACCRE	5 - 13	(\mathbf{A})	Accident Ref. No.		
F		()			
1.2	20 - 21	(\mathbf{I})	Police Force Code		
A3	22	(I)	Accident Severity		
1.5	23 - 25	(I)	No. of Vehicles		
1.6	26 - 28	(I)	No. of Casualties		
ACCDA	29 - 30	(I)	Accident Day		
Y			ž		
ACCMT	31 - 32	(I)	Accident Month		
Н					
A7	33	(I)	Day of Week		
A8H	34 - 35	(I)	Hour of Accident		
A8M	36 - 37	(I)	Minute of Accident		
1.10	38 - 40	(I)	Local Authority		
A10	41 - 45	(I)	Location - Easting		
A11	46 - 50	(I)	Location - Northing		
1.12	51	(I)	1st Road Class		
1.13	52 - 55	(I)	1st Road Number		
1.14	56	(I)	Road Type		
1.15	57 - 59	(I)	Speed Limit		
1.16	60 - 61	(I)	Junction Detail		
1.17	62	(I)	Junction Control		
1.18	63	(I)	2nd Road Class		
1.19	64 - 67	(I)	2nd Road Number		
1.20A	68	(I)	Pedestrian Crossing - Human Control		
1.20B	69	(I)	Pedestrian Crossing- Physical		
			Facilities		
1.21	70	(I)	Light Conditions		
1.22	71	(I)	Weather Conditions		
1.23	72	(I)	Road Surface Conditions		
1.24	73	(I)	Special Conditions at Site		
1.25	74	(I)	Carriageway Hazards		
1.26	75	(I)	Place Accident Reported		

Vehicle Records

Variable	Character Position	Integer /Alpha	Variable Label
ACCYR ACCRE F	1 - 4 5 - 13	(I) (A)	Accident Year (YYYY) Accident Ref. No.
2.4	14 - 16	(I)	Vehicle Ref. No.
2.5	20 - 22	(I)	Vehicle Type
2.6	23	(I)	Towing and Articulation
2.7	24 - 25	(I)	Vehicle Manoeuvre
V7	26	(I)	Compass Point - From
V8	27	(I)	Compass Point - To
2.9A	28	(I)	Vehicle Location - Road
2.9B	29 - 30	(I)	Vehicle Location - Restricted Lane/Away from
			Main Carriageway
2.10	31	(I)	Junction Location At Impact
2.11	32	(I)	Skidding/Overturning
2.12	33 - 34	(I)	Hit Object In Carriageway
2.13	35	(I)	Vehicle Leaving Carriageway
2.14	36 - 37	(I)	Hit Object off Carriageway
V15	38	(A)	Vehicle Prefix/Suffix Letter
2.16	39	(I)	1st Point of Impact
2.17	40 - 42	(I)	Other Vehicle Hit-Ref No.
V24	43 - 44	(I)	Combined Damage
V25	45	(I)	Roof/Underside Damage
2.21	46	(I)	Sex of Driver
2.22	47 - 48	(I)	Age of Driver
2.23	49	(I)	Breath Test
2.24	50	(I)	Hit and Run

Casualty Records

Variable	Character Position	Integer /Alpha	Variable Label
ACCYR	1 - 4	(I)	Accident Year (YYYY)
ACCRE	5 - 13	(A)	Accident Ref. No.
F			
3.4	14 - 16	(I)	Vehicle Ref. No.
3.5	17 - 19	(I)	Casualty Ref. No.
3.6	20	(I)	Casualty Class
3.7	21	(I)	Sex of Casualty
3.8	22 - 23	(I)	Age of Casualty
3.9	24	(I)	Severity of Casualty
3.10	25 - 26	(I)	Pedestrian Location
3.11	27	(I)	Pedestrian Movement
3.12	28	(I)	Pedestrian Direction
3.13	29	(I)	School Pupil
C13	30	(I)	Seat Belt Usage (1979 - 1993)
3.15	31	(I)	Car Passenger
3.16	32	(I)	Bus or Coach Passenger
C16	33 - 35	(I)	Casualty Type

III.

Scottish Morbidity Records 01 and General records office Scotland mortality codes

File Sour File Loca Contact:	ce: tion:	ASCII (SDF) Unix Server: \\zippy/conf/linkag Kevin McInneny, James Boyd	e	
Position From	Position To	Field Name	Size	Description/Format/Values
1	8	Personal Identifier	8	Link number
9	16	Date of admission	8	CCYYMMDD
r 17	24	Date of discharge	8	CCYYMMDD
25	27	Record type	3	01B = SMR01 record 1997 Q2 onward
28	35	Accession number	8	Numeric
36	43	Unique record identifier	8	Numeric
44	47	Surname soundex code	4	Alpha-numeric in the format ANNN
48	51	Maiden name soundex code	4	Alpha-numeric in the format ANNN
52	55	Common unit code	4	Integer
56	61	One pass link weight (score)	6	Real number
62	62	Sort marker	1	
63	70	Date of linkage	8	CCYYMMDD
71	78	Date last amended	8	CCYYMMDD
79	81	Continuous inpatient stay	3	
82	82	Summarised admission code	1	
83	83	Summarised discharge code	1	
84	100	Filler	17	Reserved for future MRL use
101	105	Sending location	5	Alpha-numeric in the format ANNNA
106	116	Episode record key	11	
	1.00	Patient ID D	ata :-	
117	136	Surname	20	
137	156	1 st forename	20	
157	176	2 nd forename	20	
177	196	Previous surname	20	
197	204	Date of birth	8	CCYYMMDD
205	205	Sex	1	1=male 2=female
206	206	Marital status	1	i mate, 2 remate
207	216	CI/CHI number	10	
217	226	NHS number	10	
227	230	Health records system identifier	4	
231	240	Patient identifier	10	Case reference number - Alpha numeri
241	250	Alternative case reference number	10	
251	258	Postcode	8	
259	260	Ethnic group	2	
261	266	GP practice code	6	
267	274	Referring GP/GDP GMC number	8	Can be alpha numeric
1		Enisode Manageme	ant Data :	our de alpina namerie
275	204	Episode Managemo	ent Data :	-
2/3	284	Care package identifier	10	
200	289	Location	5	Alpha-numeric in the format ANNNA
290	292	Specialty	3	
293	293	Speciality local code	1	
294	295	Significant facility	2	Alpha numeric
296	298	Clinical facility – start	3	Alpha numeric
299	301	Clinical facility – end	3	
202	309	Consultant/HCP responsible for care	8	Can be alpha numeric

Page 1 of 3

LINKED SMR1/SMR4/SOCRATES/GRO DEATH CATALOG

Posit Fro	ion Position m To	Field Name	Size	Description/Format/Values
310) 310	Manage of patient	1	
311	311	Patient category	1	
312	2 316	Provider code	5	Code in the format AAANN
317	7 321	Purchaser code	5	
322	2 327	Serial number	6	
328	3 333	GP referral letter number	6	Numeric
334	4 334	Waiting list guarantee exception code	1	
335	5 342	Waiting list date	8	CCYYMMDD
343	3 350	Admission date	8	CCYYMMDD
35	351	Waiting list type	1	
- 352	2 353	Admission type	2	
354	4 355	Admission reason	2	Alpha numeric
350	5 357	Admission/transfer from	2	Alpha numeric
35	362	Admission/transfer from - location	5	Alpha-numeric in the format ANNNA
36	3 370	Ready for discharge date	8	CCYYMMDD
5 37	1 378	Discharge date	8	CCYYMMDD
37	9 380	Discharge type	2	
38	1 382	Discharge/transfer to	2	Alpha numeric
38.	3 387	Discharge/transfer to - location	5	Alpha-numeric in the format ANNNA
		General Clinical	Data:-	
38	8 393	Main condition	6	ICD10
39	4 399	Other condition 1	6	ICD10
400) 405	Other condition 2	6	ICD10
400	5 411	Other condition 3	6	ICD10
41	2 417	Other condition 4	6	ICD10
418	3 423	Other condition 5	6	ICD10
424	427	Main operation - A	4	OPCS4
428	3 431	Main operation - B	4	OPCS4
432	2 439	Date of main operation	8	CCYYMMDD
44() 447	Clinician responsible for main operation	8	Can be alpha numeric
448	3 451	Other operation 1 - A	4	OPCS4
452	455	Other operation 1 - b	4	OPCS4
450	6 463	Date of other operation 1	8	CCYYMMDD
464	471	Clinician - other operation 1	8	Can be alpha numeric
472	475	Other operation 2 - A	4	OPCS4
476	479	Other operation 2 - B	4	OPCS4
480	487	Date of other operation 2	8	CCVVMMDD
488	495	Clinician - other operation 2	8	Can be alpha numeric
496	499	Other operation 3 - A	4	OPCS4
500	503	Other operation 3 - B	4	OPCS4
504	511	Date of other operation 3	8	CCYYMMDD
512	519	Clinician - other operation 3	8	Can be alpha numeric
520	549	Filler	520	can be appla numeric
		Additional back-mann	od fields	
550	550	Inpatient/day case marker	1	Alpha numeric
551	552	Old specialty code	2	a siplica intellicence
553	553	Old type of admission code	1	
554	554	Deprivation category 5	1	
554	555	Deprivation category 7	1	
777	000	Deprivation category /	1	
556	557	Numeric health board and	2	

Page 2 of 3

LINKED SMR1/SMR4/SOCRATES/GRO DEATH CATALOG

Position From	Position To	Field Name	Size	Description/Format/Values
559	565	7 character postcode	7	Postal Sector 550 540
		Derive	d Data :	rostal Sector 559-563
566	566	HB of residence cypher	u Data	204
567	568	Local Government District	1	Character
569	578	Grid reference	2	
579	579	Catchment area	10	
580	581	Local Government Region	1	
582	584	Ward code	2	
585	586	New town code	3	
587	587	Urban/Rural code	2	
588	592	Pop locality	1	
593	598	Enumeration district	5	
599	600	Local council area	0	
601	602	Parliamentary constituency	2	
603	605	Age in years	2	
606	609	Age in months	3	
610	614	Days waiting	4	
615	619	Length of stay		
620	622	HRG	2	
623	623]	Postcode query ind	3	
624	624]	Error flag	1	0
625	625 (Query flag	1	0 = No, 1 = Yes
	100	Processing	History	0 = No, 1 = Yes, 2 = Amended
626	631 F	Batch number	mstory :-	
632	635 F	Batch sequence number	6	
636	641 1	Version number	4	
642	649 I	Date record inserted	6	
650	657 E	Date last amended	8	CCYYMMDD
658	658 D	Delete flag	8	CCYYMMDD
659	659 A	ccredited flag	1	
660	660 P	rocessing site	1	
661	665 In	put provider	1	
666	671 D	ate (of discharge)	5	

Page 3 of 3

IV. ICD 10 codes

ICD 10 injury codes and external codes utilized in the study Retrieved from WHO website: http://apps.who.int/classifications/apps/icd/icd10online/

Injury, poisoning and certain other consequences of external causes (<u>S00-T98</u>)

<u>S00-S09</u> Injuries to the head <u>S10-S19</u> Injuries to the neck

S20-S29 Injuries to the thorax

<u>S30-S39</u> Injuries to the abdomen, lower back, lumbar spine and pelvis

S40-S49 Injuries to the shoulder and upper arm

<u>S50-S59</u> Injuries to the elbow and forearm

S60-S69 Injuries to the wrist and hand

S70-S79 Injuries to the hip and thigh

S80-S89 Injuries to the knee and lower leg

S90-S99 Injuries to the ankle and foot

T00-T07 Injuries involving multiple body regions

T08-T14 Injuries to unspecified part of trunk, limb or body region

External causes of morbidity and mortality

(V01-Y98)

V01-X59 Accidents

V01-V99 Transport accidents

V01-V09 Pedestrian injured in transport accident

<u>V10-V19</u> Pedal cyclist injured in transport accident

V20-V29 Motorcycle rider injured in transport accident

V30-V39 Occupant of three-wheeled motor vehicle injured in transport accident

V40-V49 Car occupant injured in transport accident

V50-V59 Occupant of pick-up truck or van injured in transport accident

V60-V69 Occupant of heavy transport vehicle injured in transport accident

V70-V79 Bus occupant injured in transport accident

<u>V80-V89</u> Other land transport accidents

V90-V94 Water transport accidents

V95-V97 Air and space transport accidents

<u>V98-V99</u> Other and unspecified transport accidents

V. The Scottish record linkage system

Selected parts of the paper on The Scottish Record Linkage System. No author is

stated, but the full paper is online at

http://www.isdscotland.org/isd/files/The%20Scottish%20Record%20Linkage%20System

<u>.doc</u>

THE CURRENT PROJECT

It was envisioned that the creation of the national linked data sets would be carried out purely by automated algorithms with no clerical checking or intervention involved. After linkage of five years of data in the main linked data set it was found that the false positive rate in the larger groups of records was beginning to creep up beyond the 1% level felt to be acceptable for the statistical and management purposes for which the data sets are used. Limited clerical checking has been subsequently used to break up falsely linked groups. This has served to keep both the false positive and false negative rates at below three per cent. More extensive clerical checking is used for specialised purposes such as the linking of death records to the Scottish Cancer Registry to enable accurate survival analysis for example.

METHODS OF LINKING

In a world with perfect recording of identifying information and unchanging personal circumstances, all that would be necessary to link records would be the sorting of the records to be matched by personal identifiers. In the real world of data however, for each of the core items of identifying information used to link the records (surname, initial, year, month and day of birth), there may be a discrepancy rate of up to 3% in pairs of records belonging to the same person. Thus exact matching using these items could miss up to 15 % of true links.

To allow for the imperfections of the data, the system uses methods of probability matching which have been developed and refined in Canada 3, Oxford 4 and Scotland 5 itself over the last thirty years. Despite the size of the data sets, linking the records consists of carrying out the same basic operation over and over again. This operation is the comparison of two records and the decision as to whether they belong to the same individual.

THE ELEMENTS OF LINKAGE.

1. 1. Bringing pairs of records together for comparison. How do we bring the most effective subset of pairs of records together for comparison? It is usually impossible to carry out probability matching on all pairs of records involved in a linkage. Usually only a subset are compared, those which share a minimum level of identifying information. This has been traditionally achieved by sorting the files into 'blocks' or 'pockets' within which paired comparisons are carried out e.g. soundex, date of birth etc. (Gill and Baldwin, 1987).

2. 2. Calculating probability weights. How do we assess the relative likelihood that pairs of records belong to the same person? This lies at the heart of probability matching and has probably been the main focus of much of record linkage literature (Newcombe, 1988).

3. 3. Making the linkage decision. How do we convert the probability weights representing relative odds into absolute odds which will support the linkage decision? The wide variety of linkages undertaken has been particularly important in moving forward understanding in this area.

1. Blocking

In an ideal world with infinite computing power we would carry out probability matching between every pair of records in order to determine whether they belong to the same person. At present this is realistically beyond current computing capacities and would be enormously wasteful even if it were possible. It is necessary to cut down in some way the number of pair comparisons which are made in a given linkage. Instead of comparing all pairs of records we compare only those records which have some minimum level of agreement in identifying items ('blocking' the records).

In the linkages carried out at ISD we tend to compare only those pairs of records between which there is agreement on:

Soundex/NYSIIS code, first initial and sex	(Block A)	

All elements of date of birth (day, month, year) (Block B) or

Thus records will not be compared if they disagree on one or more of the first set of blocking items and also disagree on one or more of the second set of blocking items. It is of course possible that two records belonging to the same person will disagree on for example, first initial and also date of birth. Experience shows that the proportion of true links thus lost because of blocking is less than 0.5%.

2. Probability Weights

Our approach to the calculation of probability weights has been relatively conventional and can be quickly summarised. A concern has been to avoid over-elaboration and over complexity in the algorithms which calculate the weights. Beyond a certain level increasing refinement of the weight calculation routines tends to involve diminishing returns.

For the internal linking of hospital discharge (SMR1) records across Scotland we have available the patient's surname (plus sometimes maiden name), forename, sex and date of birth. We also have postcode of residence. For records within the same hospital (or sometimes the same Health Board) the hospital assigned case reference number can be used. In addition positive weights can be assigned for correspondence of the date of discharge on one record with the date of admission on another. Surnames are compressed using the Soundex/NYSIIS name compression algorithms (Newcombe, 1988) with additional scoring assigned for more detailed levels of agreement and disagreement. Wherever possible specific weights relating to degrees of agreement and disagreement are used. Soundex and related name compression algorithms overcome some of the problems associated with misspelling of names and variant spellings.

Blocking allows subsets of the records to be efficiently brought together for comparison. Finally and most importantly probability matching allows mathematically precise assessment of the implications of the levels of agreement and disagreement between records.

Probability matching

Two very simple and common sense principles underlie probability matching:

А. Every time an item of identifying information is the same on the two records, the А. probability that they apply to the same person is increased.

B. Β. Every time that an item of identifying information differs between two records, the probability that they apply to the same person is usually decreased.

Whatever kind of matching we are doing, whether linking records within a file or linking records between files, we are looking at pairs of records and trying to decide whether they belong to the same person or don't belong to the same person. We are trying to divide the pairs into two classes - which are more generally referred to as 'truly linked' or 'truly unlinked', i.e. in our case belonging to the same person or not belonging to the same person.

The common core of identifying items are as follows:

- 1. 1. Surname
- 2. 2. First initial (also full forename and second initial if available)
- 3. 3. Sex
- 4. 4. Year, month and day of birth
- 5. 5. Postcode.

In principle, any items whose level of agreement or disagreement influences the probability that two records do or do not belong to the same person can be used by the computer algorithm. However, items should be statistically independent as far as possible.

Every time we compare an item of identifying information between two records we obtain what can be called an outcome. In the first instance this is either agreement or disagreement.

For every outcome we ask the same two questions.

1. 1. How often is this outcome likely to occur if the two records really do belong to the same person (are truly linked)?

2. 2. How often is this outcome likely to occur if the two records really don't belong the same person (are truly unlinked)?

The ratio between these two probabilities or odds is what is called an odds ratio - this is a measure of how much that particular outcome has increased or decreased the chances that the two records belong to the same individual. Odds can be awkward to handle so probability matching tends to use binit weights instead. The binit weight is the odds expressed as a logarithm to base 2.

The linkage methodology is aimed at squeezing the maximum amount of discrimination from the available identifying information. Thus the distribution of probability scores differs for each kind of linkage. The threshold (or score at which the decision to link is made) is determined by clerical checking of a sample of pairs for each type of link.

The odds ratio: an example

Suppose we have two records, and we are comparing their first initials. We find that they both have first initial J'. We want to calculate an odds ratio which will tell us what effect this outcome - agreement of first initial J' - has on the chances that the records belong to the same person.

If both records belong to the same person how often will one record have the initial J?? In a perfect world with perfect data the answer would be always - the probability would be one, or in percentage terms, 100%. However, there are often going to be discrepancies in identifying information between records applying to the same person. If we estimate that the first initial is likely to disagree 3% of the time on records applying to the same person, then it will agree 97% of the time. So on the top line of our odds ratio we have a figure of 97%.

Next we look to the bottom line of the odds ratio. How often are we going to get agreement on the initial 'J' among pairs of records which do not belong to the same person? The answer quite simply depends upon how common that first initial is. If 20% of all first initials are 'J', then if we take any record with first initial 'J' and compare it with all the other records, then 20% of the time the record it is compared with will have first initial 'J'. So the bottom line of the odds ratio is 20%. The odds ratio then is 97%/20% or 4.85.

So agreement of first initial 'J' has improved our chances that the records belong to the same person by 4.85 to one.

What if the first initial disagrees? Again we compare the outcome among pairs of records which do belong to the same person against pairs of records which do not.

The top line of the odds ratio is 3% (if you take all records with initial J', then 3% of the time even among records belonging to the same person - the other record will have a different initial.) For the bottom line, we want to know how often the first initial disagrees when the records do not belong to the same person. For illustration we can take the initial as disagreeing 92.5% of the time among records not belonging to the same person. So for disagreement of first initial we have an odds ration of 3%/92.5% or 1 to 32. So disagreement of first initial has reduced the chances that the records belong to the same person by 32 to 1.

So we now have a quantitative estimate of how much an agreement on first initial 'J' has improved our chances that we are looking at records belonging to the same person. Similarly we have a quantitative estimate of how much a disagreement on first initial has reduced the chances that the records relate to the same person.

We can now give an example of how the odds ratios deriving from comparison of individual identifying items can be combined to give odds for the overall comparison of the two records.

Suppose we have two records each with the identifying information:

Male J Thompson	born 15 05 1932
Male J Thompson	born 05 05 1932

The odds associated with these comparisons are as follows:

Sex				
Agreement: odds ratio	99.5%/50%	=	1.99	+0.99
First initial				
Agreement: odds ratio	97%/20%	=	4.85	+2.28
Surname				
Agreement: odds ratio	97%/0.8%	=	121.25	+6.92
Day of birth				
Agreement: odds ratio	3%/92%=	0.0326	-4.94	
Month of birth				
Agreement: odds ratio	97%/8.3%	=	11.7	+3.55
Year of birth				
Agreement: odds ratio	97%/1.4%	=	70.0	+6.13

How much have all these comparisons of identifying information improved the chances that these two records really apply to the same person? You combine odds by multiplying them:

 $1.99 \times 4.85 \times 121.25 \times 0.0326 \times 11.7 \times 70 = 31,245 \text{ to } 1.$

Binit

So the comparisons have increased the likelihood that the two records belong to the same person by 31,245 to 1. However, that does not mean that it is a certainty. Our files have millions of records on millions of individuals. It is not inconceivable that there is more than one male J. Thompson born on the 14th or 15th of May 1932. That is why the procedure is known as probability matching - there are never any certainties. And since there are no certainties, we still have to make a decision as to whether or not the records do apply to the same person.

Binit weights

Odds like 31,245 to 1 are rather awkward to handle. Probability matching tends to use instead what are called binit weights. So far we have talked about odds ratios e.g. the odds ratio for agreement on initial 'j' is 4.85 to 1. The binit weight is this number expressed as a logarithm to base 2.

In this context, the most useful thing about logarithms in general, or binit weights in particular, is that they can be added together. Adding together the binit weights is the same as multiplying the odds ratios. So our overall improvement in the chances that the records belong to the same person of 31,245 to 1 is equivalent to a binit weight of 14.93.

The essence of record linkage is to calculate the overall binit weight for each pair of records. High binit weights mean that the records are likely to belong to the same person. Low binit weights (which reflect odds against) mean that the records are unlikely to belong to the same person.

Soundex/NYSIIS codes

Surnames are changed to a coded format in order to overcome the effects of most discrepancies in the spelling. First the NYSIIS (New York State Intelligence Information System) name compression algorithm is applied. This carries out such tasks as bringing together commonly confused letter groups like 'ch' and 'gh' or 'sh' and 'sch' as well as removing vowels. The surnames are then Soundexed 6 , which involves giving the same code to similar sounding non-initial constants. The resulting compression and soundex codes are assigned different weights for agreement depending upon their frequency in the population.

3. Decision-making

Binit weights present us with a mathematical expression of the extent to which the available identifying information increases or decreases the chances that two records belong together. These however are only relative odds. They allow us to rank order the pairs of records in order of likelihood. They are not absolute odds. Such absolute odds depend upon various factors such as the size of the data sets involved. Methods of calculating such absolute odds are available but they are usually based on rather speculative assumptions. It is much safer to base the decision on which records belong together on a match weight threshold based on empirical inspection. In other words we compare records, calculate relative odds for each pair and look at a selection of odds before deciding on the cut off point for accepting matches.

When the frequencies of pairs of records with given values of the binit weight are graphed, a bimetal pattern usually emerges (see IARC report No.32 - Automated Data Collection in Cancer Registrations). The group of pairs of records with high binit weights can be taken as matches (as belonging to the same person). The group with low binit weights can be regarded as non-matches. It is the group in between which cause problems.

The crucial step is to identify a threshold above which pairs will be taken as linking, and below which the pairs will not be accepted as linking.

This threshold is usually determined by clerical inspection of a sample of pairs of records. The threshold is usually set at the 50/50 point. In other words, at the threshold it is a fifty-fifty bet as to whether the pair of records belongs to the same person. Above the threshold it is more likely than not that they do belong to the same person. Below the threshold it is more likely than not that they do not belong to the same person.

Once a threshold in terms of the binit weight has been set, the computer can be allowed to make the decisions as to whether records belong together. In practice the development of match weights and the setting of the threshold is an iterative process with results depending on the precise characteristics of the data sets involved.

Tuning the linkages

Tuning the linkages, either in terms of adjusting the weights for particular comparisons or by adjusting the match threshold is an iterative process.

All linkages are different and the quality of the linkage is best ensured by taking careful account of the precise properties of the data sets involved and the different problems which emerge in linking two particular data sets.

The linkage threshold is established or confirmed by inspecting the pairs output file and thereby establishing the weight above which it is more likely than not that a pair of records belong to the same person and below which it is more likely than not that the pair of records do not belong to the same person. This threshold is often confirmed in terms of the graph of the frequency of the outcome weights for a particular linkage. The 50/50 threshold (the weight at which it is evens whether records do or do not belong together) often corresponds to the low point of the trough in the frequency counts.

QUALITY OF LINKAGE

The linkage system has been automated as much as possible. The probability matching algorithm alone makes the decision as to whether records belong together. Clerical monitoring shows that on a pairwise basis, both the false positive rate (the proportion of pairs which are incorrectly linked) and the false negative rate (the proportion of pairs which the system fails to link) are around three per cent.

As the data set has expanded the number of patient record sets with large numbers of records has grown. In order to construct a patient record set with 10 records, up to 45 pair comparisons will have been carried out, each comparison contributing its own possibility of a false positive link. Thus larger groups of records are more likely to be false positive. Some of the more important groups moreover tend to be the larger groups. Patient record sets containing cancer registrations tend to have more records than average have thus have a relatively high error rate. For this reason, groups of records where there is an obvious error such as two death records or a hospital admission following death have been targeted for clerical correction. Such errors will help to keep the overall false positive and false negative rates close to one percent.

By using such a focused approach to clerical checking we are intending to achieve the advantages of the quality of a fully clerically checked system without the massive investment of time and expense which such a system would involve.

ESTIMATION OF FALSE POSITIVE RATES (QUOTE FROM MR DAVID CLARK ANALYST AT ICD):

Essentially what I did was break down the cohort into the various matching standards and within this (probability weights as generated by the linkage)and sampled the matching pairs and tried to decide as best as I could from all the available information (e.g. Age, sex, dates, postcode, council area, hospital of treatment, diagnosis codes, severity, casualty class, whether there were rival potential matches or not) whether I thought the match was good, bad or undecided. This is a very subjective process. I scored 2 for good, 1 for undecided, 0 for bad. I then calculated a "probability" for each category e.g. if there were 20 cases sampled at a particular score and 10 were good, 5 were bad, 5 were unsure the probability of a good link for this group of links was 25/40 = 67.5%. If there were 200 links in total for this group then could estimate 135 good links and 65 bad links. I then summed up the total over all the categories of links to get the total estimated good and bad links. Likewise I sampled below the linkage threshold to find the number of estimated good and bad missed links.

VI. Follow-up diagnosis significantly related to a 1st single injury

P values were calculated using chi square and criteria for inclusion for further analysis were that no more than 1 cell can have an expected count of <5 and a significant p value of <0.2. A total of 41 follow-up diagnoses (FUD) were considered significantly related to one or more injuries. The specific injury that was related to a FUD were identified by analysing the proportions of these in relation to all injury categories and the FUD that have more than 30% over the expected proportion of FUD were considered statistically related (e.g. the follow-up diagnosis "cerebrovascular diseases had a p value of 0.09 (i.e. statistically related to one or more of the 1st diagnose) and casualties with injuries to the thorax, 13% of total casualties, had 21% of this diagnose i.e. 62% over the expected rate). Two diagnoses had no injury with >30% FUD, but the injury closest to >30% were identified in these cases (highlighted in *italic*).

In summary the following correlations were found (the % next to the injury is the proportion of casualties with that particular injury and the % next to FUD is the proportion of casualties with that FUD previously having that injury, see also table):

- \Rightarrow Injuries to the head (34%)
 - Accidental exposure to other and unspecified factors (53%)
 - Assault (48%)
 - Diseases of oral cavity, salivary glands and jaws (49%)
 - Mental and behavioural disorders due to psychoactive substance use (46%)
 - Persons encountering health services for examination and investigation (44%)
- \Rightarrow Injuries to the neck (6%)
 - Assault (8%)
 - Benign neoplasms (10%)
 - Diseases of oesophagus, stomach and duodenum (13%)
 - Diseases of veins, lymphatic vessels and lymph nodes, not elsewhere classified (13%)
 - Hernia (10%)
 - Hypertensive diseases (9%)
 - Ischaemic heart diseases 10%)
 - Noninflammatory disorders of female genital tract (11%)
 - Other diseases of intestines (10%)
 - Other diseases of the digestive system (14%)
 - Other diseases of urinary system (11%)
 - Other disorders of the skin and subcutaneous tissue (10%)
 - Persons encountering health services in circumstances related to reproduction (13%)
 - Symptoms and signs involving the circulatory and respiratory systems (12%)
 - Symptoms and signs involving the digestive system and abdomen (11%)
- \Rightarrow Injuries to the upper limbs (11%)
 - Arthrosis (19%)
 - Disorders of bone density and structure (22%)
 - Other disorders of the skin and subcutaneous tissue (16%)
- \Rightarrow Injuries to the thorax (13%)
 - Bacterial agents resistant to antibiotics (19%)
 - Bacterial, viral and other infectious agents (21%)
 - Benign neoplasms (24%)
 - Cerebrovascular diseases (21%)
 - Diabetes mellitus (24%)

- Diseases of oesophagus, stomach and duodenum (20%)
- Disorders of gallbladder, biliary tract and pancreas (19%)
- Disorders of lens (28%)
- Hernia (27%)
- Hypertensive diseases (23%)
- Ischaemic heart diseases (22%)
- Malignant neoplasms, stated or presumed to be primary, of specified sites, except of lymphoid, haematopoietic and related tissue (28%)
- Metabolic disorders (17%)
- Other and unspecified disorders of the circulatory system (19%)
- Other diseases of intestines (23%)
- Other diseases of the digestive system (18%)
- Other diseases of urinary system (20%)
- Other dorsopathies (21%)
- Other forms of heart disease (26%)
- Renal failure (20%)
- Symptoms and signs involving the circulatory and respiratory systems (23%)
- Symptoms and signs involving the nervous and musculoskeletal systems (17%)
- Symptoms and signs involving the urinary system (26%)
- \Rightarrow Injuries to the abdomen, lower back, lumbar spine (9%)
 - Disorders of gallbladder, biliary tract and pancreas (19%)
 - Malignant neoplasms, stated or presumed to be primary, of specified sites, except of lymphoid, haematopoietic and related tissue (13%)
 - Noninflammatory disorders of female genital tract (20%)
 - Other dorsopathies (16%)
 - Persons encountering health services in circumstances related to reproduction (18%)
 - Symptoms and signs involving the digestive system and abdomen (14%)
 - Symptoms and signs involving the urinary system (14%)
- \Rightarrow Injuries to the lower limbs (26%)
 - Bacterial, viral and other infectious agents (37%)
 - Complications of surgical and medical care, not elsewhere classified (46%)
 - Disorders of bone density and structure (60%)
 - Falls (31%)
 - Other disorders of the skin and subcutaneous tissue (35%)
 - Other joint disorders (40%)
 - Other soft tissue disorders (36%)
 - Persons encountering health services for specific procedures and health care (49%)
 - Renal failure (45%)
 - Surgical and other medical procedures as the cause of abnormal reaction of the patient, or of later complication, without mention of misadventure at the time of the procedure (44%)
 - Symptoms and signs involving the nervous and musculoskeletal systems (35%)

Some correlations may be obvious such as injuries to the lower limbs and the FUD "other joint disorders" and injuries to the thorax which is correlated to "ischaemic heart diseases", while others are more questionable. Injuries to the neck, for example appears to have a large number of related FUD including several diagnoses of the abdominal region (reproductive, digestive, intestines etc) –could this possibly be related to neck injuries resulting in paralysis? Some of the significantly related FUDs are few in numbers e.g. "arthrosis" N=59, but seeing as the p value is relatively low and that casualties with injuries to the upper limbs suffer this FUD about 70% more than the expected rate it is reasonable to include this in the FAD 'selection criteria'.
ICD-			# Cells w expected count of	Injuries to the Injuries to the upper limb head (34%) neck (6.2%) (11%)		o the bs) Injuries to the thorax (13%)		Injuries to the abdomen, lower back, lumbar spine (9%)		Injuries to the lower limbs (26%)		Total			
code	ICD-10 description	Р	5	Row %	Sum	Row %	Sum	Row %	Sum	Row %	Sum	Row %	Sum	Row %	Sum	Sum
X58- X59	Accidental exposure to other and unspecified factors	0.000	0	53.3	80	6.0	9	10.7	16	2.7	4	5.3	8	22.0	33	150
M15- M19	Arthrosis	0.130	1	20.3	12	3.4	2	18.6	11	16.9	10	11.9	7	28.8	17	59
X85- Y09	Assault	0.006	0	48.0	59	8.1	10	12.2	15	6.5	8	7.3	9	17.9	22	123
U80- U89	Bacterial agents resistant to antibiotics	0.011	0	30.1	112	7.0	26	10.5	39	19.4	72	8.9	33	24.2	90	372
B95- B97	Bacterial, viral and other infectious agents	0.031	1	20.0	14	5.7	4	10.0	7	21.4	15	5.7	4	37.1	26	70
D10- D36	Benign neoplasms	0.005	0	21.0	21	10.0	10	11.0	11	24.0	24	8.0	8	26.0	26	100
160- 169	Cerebrovascular diseases	0.090	0	33.1	44	7.5	10	8.3	11	21.1	28	6.0	8	24.1	32	133
T80- T88	Complications of surgical and medical care, not elsewhere classified	0.000	0	22.4	34	7.2	11	9.9	15	5.9	9	8.6	13	46.1	70	152
E10- E14	Diabetes mellitus	0.013	0	24.0	24	7.0	7	7.0	7	24.0	24	12.0	12	26.0	26	100
K20- K31	Diseases of oesophagus, stomach and duodenum	0.000	0	25.8	59	13.1	30	9.6	22	19.7	45	10.5	24	21.4	49	229
K00- K14	Diseases of oral cavity, salivary glands and jaws	0.000	0	48.6	70	4.2	6	8.3	12	4.2	6	4.2	6	30.6	44	144
180- 189	Diseases of veins, lymphatic vessels and lymph nodes, not elsewhere classified	0.042	0	28.2	33	12.8	15	10.3	12	14.5	17	12.0	14	22.2	26	117
M80- M85	Disorders of bone density and structure	0.000	0	8.3	8	2.1	2	21.9	21	6.3	6	1.0	1	60.4	58	96
K80- K87	Disorders of gallbladder, biliary tract and pancreas	0.027	1	23.8	15	6.3	4	14.3	9	19.0	12	19.0	12	17.5	11	63

ICD-			# Cells w expected count of	Injuries t head (34	o the ŀ%)	Injuries t neck (6.2	o the 2%)	Injuries to upper lim (11%)	o the bs	Injuries thorax (?	to the 13%)	Injuries t abdomer lower ba lumbar s (9%)	o the n, ck, pine	Injuries to lower lim (26%)	o the bs	Total
code	ICD-10 description	Р	5	Row %	Sum	Row %	Sum	Row %	Sum	Row %	Sum	Row %	Sum	Row %	Sum	Sum
H25- H28	Disorders of lens	0.000	0	22.8	21	1.1	1	7.6	7	28.3	26	7.6	7	32.6	30	92
W00- W19	Falls	0.070	0	34.6	112	3.7	12	12.3	40	11.1	36	7.4	24	30.9	100	324
K40- K46	Hernia	0.000	0	20.5	26	10.2	13	7.1	9	26.8	34	11.8	15	23.6	30	127
l10- l15	Hypertensive diseases	0.000	0	21.6	33	9.2	14	7.8	12	22.9	35	9.2	14	29.4	45	153
120- 125	Ischaemic heart diseases	0.000	0	22.1	53	10.0	24	8.3	20	22.1	53	10.8	26	26.7	64	240
C81- C96	Malignant neoplasms, stated or presumed to be primary, of specified sites, except of lymphoid, haematopoietic and related tissue	0.000	0	25.7	38	2.7	4	10.1	15	27.7	41	12.8	19	20.9	31	148
F10- F19	Mental and behavioural disorders due to psychoactive substance use	0.000	0	46.2	104	4.0	9	12.9	29	7.1	16	6.2	14	23.6	53	225
E70- E90	Metabolic disorders	0.100	0	41.3	43	6.7	7	6.7	7	17.3	18	11.5	12	16.3	17	104
N80- N98	Noninflammatory disorders of female genital tract	0.000	0	33.6	42	11.2	14	11.2	14	12.8	16	20.0	25	11.2	14	125
195- 199	Other and unspecified disorders of the circulatory system	0.000	0	30.1	157	6.3	33	9.4	49	19.3	101	7.5	39	27.4	143	522
K55- K63	Other diseases of intestines	0.000	0	24.3	60	9.7	24	9.3	23	23.1	57	10.9	27	22.7	56	247
K90- K93	Other diseases of the digestive system	0.054	1	35.6	26	13.7	10	6.8	5	17.8	13	6.8	5	19.2	14	73
N30- N39	Other diseases of urinary system	0.003	0	23.9	38	11.3	18	10.1	16	19.5	31	7.5	12	27.7	44	159
L80- L99	Other disorders of the skin and subcutaneous tissue	0.015	0	30.5	25	9.8	8	15.9	13	6.1	5	2.4	2	35.4	29	82
M50- M54	Other dorsopathies	0.109	1	25.0	17	7.4	5	10.3	7	20.6	14	16.2	11	20.6	14	68

ICD-	ICD-		# Cells w expected count of	Injuries head (34	to the 1%)	Injuries t neck (6.2	o the 2%)	Injuries to upper lim (11%)	o the bs	Injuries to the thorax (13%)		Injuries to the abdomen, lower back, lumbar spine (9%)		Injuries to the lower limbs (26%)		Total	
code	ICD-10 description	Р	5	Row %	Sum	Row %	Sum	Row %	Sum	Row %	Sum	Row %	Sum	Row %	Sum	Sum	
130- 152	Other forms of heart disease	0.000	0	22.3	39	4.6	8	7.4	13	25.7	45	9.7	17	30.3	53	175	
M20- M25	Other joint disorders	0.012	0	24.8	28	7.1	8	12.4	14	8.0	9	8.0	9	39.8	45	113	
M70- M79	Other soft tissue disorders	0.104	0	23.9	28	6.8	8	12.8	15	11.1	13	9.4	11	35.9	42	117	
Z00- Z13	Persons encountering health services for examination and investigation	0.002	0	44.0	103	5.6	13	9.8	23	14.5	34	10.7	25	15.4	36	234	
Z40- Z54	Persons encountering health services for specific procedures and health care	0.000	0	19.1	110	3.3	19	13.7	79	7.5	43	7.8	45	48.5	279	575	
Z30- Z39	Persons encountering health services in circumstances related to reproduction	0.041	1	28.6	16	12.5	7	14.3	8	10.7	6	17.9	10	16.1	9	56	
N17- N19	Renal failure	0.002	1	25.0	15	3.3	2	1.7	1	20.0	12	5.0	3	45.0	27	60	
Y83- Y84	Surgical and other medical procedures as the cause of abnormal reaction of the patient, or of later complication, without mention of misadventure at the time of the procedure	0.000	0	24.5	36	6.8	10	8.8	13	8.2	12	7.5	11	44.2	65	147	
R00- R09	Symptoms and signs involving the circulatory and respiratory systems	0.000	0	25.8	66	11.7	30	10.2	26	22.7	58	8.2	21	21.5	55	256	
R10- R19	Symptoms and signs involving the digestive system and abdomen	0.000	0	29.0	93	10.6	34	12.5	40	16.2	52	14.3	46	17.4	56	321	
R25- R29	Symptoms and signs involving the nervous and musculoskeletal systems	0.086	0	18.5	15	7.4	6	12.3	10	17.3	14	9.9	8	34.6	28	81	
R30- R39	Symptoms and signs involving the urinary system	0.000	0	28.0	33	5.9	7	5.9	7	26.3	31	13.6	16	20.3	24	118	
	All injuries			34.2	1112	6.2	203	11.0	359	13.3	432	9.3	303	26.0	847		

-

VII. Weighted mean cost by speciality

		Inpatient	Day case
General Surgery	01	510.11	624.81
Orthopaedic Surgery	02	599.67	936.29
ENT	03	894.95	783.81
Ophthalmology	04	1092.17	798.76
Urology	05	555.69	506.88
Neurosurgery	06	625.61	2208.63
Cardiothoracic Surgery	07	1013.57	No daycase data
Plastic Surgery	08	828.17	695.33
Orthodontics/Paediatric			
Dentistry	11		
Oral Surgery/Medicine	12		
General Medicine	16	328.84	426.77
Cardiology	17	830.55	1169.82
Metabolic Diseases	18		
Neurology	19	692.19	1350.11
Gastroenterology	21	455.19	561.48
Dermatology	23	231.01	142.02
Nephrology	24	546.48	799.56
Rheumatology	25	607.63	583.11
Rehabilitation Medicine	26	217.39	562.92
Respiratory Medicine	28	427.76	445.15
Communicable Diseases	31	396.43	4755.88
Diagnostic Radiology	33		
Radiotherapy	34	106.95	573.25
Homeopathy	36		
Medical Oncology	37	2723.72	647.94
Spinal Paralysis	38	338.10	513.77
Surgical Paediatrics	39	794.97	1095.63
Medical Paediatrics	40	754.08	750.56
Pain Control	41		
Gynaecology	42	704.04	711.87
Intensive Therapy Unit	48		
Accident & Emergency	49	809.08	no info
Geriatric Assessment	50	222.36	no info
Young Chronic Sick	52		
Haematology	62	499.05	349.46
Gp (Ex. Obstetrics)	73	274.95	189.72
Other Acute	98		896.42